

Waiver/Shortening of Cooling off Period by a Proposed Buyer

Issued pursuant to section 167 of the Property Occupations Act 2014

NOTE FOR COMPLETION

Under the Property Occupations Act 2014 a buyer may:

- 1. waive their cooling off rights before entering into a relevant contract; or
- 2. shorten their cooling off rights **after** they enter into a relevant contract.

This form should be completed and signed either before or after the relevant contract is entered into depending on whether the buyer wishes to exercise option 1 or 2 above.

A "relevant contract" is defined under section 160 of the Ad	et.
I/We	
(insert name/s of proposed buyer/s)	
understand that I am/ we are entitled to a five business day 160 and 166 of the Property Occupations Act 2014 in relat	y statutory cooling off period available under sections ion to a contract for the sale of the property located at:
	STATE: QLD POSTCODE: 4216
("the Property")	
I/ we wish to:	
\checkmark waive my/our statutory cooling off rights.	
shorten my/our statutory cooling off rights from 5 bus	siness days to days. (insert relevant number of days)
I/ We agree that I have/ we have voluntarily agreed to waiv	e or shorten my/our statutory cooling off rights.
SIGNATURES	
Buyer Name:	Buyer Name:
Signature:	Signature: SIGN HER
Date:	Date:

INITIALS



Managing	3 Agent	:				
AGENCY N	AME:					
N/A						
PROPERTY N/A	' MANAG	SER:				
ADDRESS:						
						_
SUBURB:					STATE:	POSTCODE:
PHONE:		MOBILE:	FAX:	EMAIL:		
POOL S	AFETY	Y FOR NON-SHARED	POOLS			
		ving questions if there is a n		←WARNIN	IG TO SELLER: Failure to	comply with the Pool
		on-shared pool on the L	•	Safety Re	equirements is an offence	with substantial penalties
		•			IG TO BUYER: If there is ron Certificate at Settlement	
□ 7 1	Yes	Clause 4.2 of this centr	act does not apply	responsib	ble at its cost to obtain a Po days after settlement. The	ool Safety Certificate
_	No	Clause 4.2 of this contr		liable to p	pay any costs of rectification of safety Requirements.	n necessary to comply
Q2. If the Certifi	answer	r to Q1 is Yes, is there a r the non-shared pool a	Compliance or Exemption t the time of contract?	' Certificate	e. The Buyer commits an c Intial penalties if the Buyer	offence and can be liable
	Yes	Clause 5.3(1)(f) applies			a pool on the Lot and Q2	is not completed then
	No	Clause 4.2 applies (exc other excluded sales)	ept for auction and some	clause 4.2	2 applies.	io not completed then
Q3. If the Certifi	answer cate be	r to Q2 is No, has a Noticen given prior to contrac	ce of No Pool Safety ct?	← Note: Thi of the Bui	is is an obligation of the Se ilding Regulation 2006.	eller under section 16
	Yes					
	No					
Pool Safe	ty Inspe	ector: N/A			Safety Inspector must be a Act 1975 and Building Reg	
Pool Safet	ty Inspe	ection Date:		← Clause 4.	.2(2) applies except where by auction and some othe	this contract is formed
STATIIT	OPV M	VADDANTIES AND C	ONTRACTUAL RIGHTS		by duction and come care	. exercise sures.
				3		
	_	notice to the Buyer of the	_			
(a) Latent NONE		nt Defects in Common Pro	operty or Body Corporate As	sets (section 223((a)(b))*	Act 1997 and the
	b) Actual or Contingent or Expected Liabilities of Body Corporate (section NONE			ion 223(2)(c)(d))*	Body Corporate and the S of a warranty may result in termination by the Buyer. whether to carry out an in Corporate records to com	n a damages claim or Seller should consider spection of the Body
(c) Circum		s in Relation to Affairs of E	Body Corporate (section 223	(3))*		
(d) Except		Warranties in clause 7.4(3	3)*			
(e) Propos	ed Body	y Corporate Resolutions (clause 8.4)*			
*Include	e in attac	chment if insufficient space				
,		opa00				

I	Excluded Fixtures:					
ı	ncluded Chattels:	Refer to	o Annexure A			
	PRICE Deposit Holder:	Coba	alt Blue Resources Pty	/ Ltd		
	eposit Holder's Tı	rust Acco	ount: Cobalt Blue Reso	ources Statutory Trust Ac	count	
	Bank:	West		indicate of the contract of th		
Е	SSB: 034-061		Account No: 424			
	Purchase Price:	\$				← Unless otherwise specified in this contract, the Purchase Price includes any GST payable on the supply of the Property to the Buyer.
D	eposit:	\$ 10%	of sale price	Initial Deposit payable o specified below. Signing of Expression of	,	ns this contract unless another time is
		\$		Balance Deposit (if any) Within 3 Business days	payable on:	t
D	efault Interest Rat	e:	%	← If no figure is inserted, the Queensland Law Society	ne Contract Rate applying y Inc will apply.	at the Contract Date published by the
F	INANCE					
	inance Amount:	\$			leted, this contract is not s	"Financier" and "Finance Date" are subject to finance and clause 3 does
F	inancier:					
F	inance Date:			_		
	SUILDING AND	OR PE	ST INSPECTION D	ATE:	← If "Inspection ← not subject to does not app	Date" is not completed, the contract is an inspection report and clause 4.1
Τ	NATTERS AFFE	es:	S PROPERTY to any Encumbrances?	√ No Yes, lis		yARNING TO SELLER: You are require
		Subject		Y 100, 10	tс и е е и Е	o disclose all Title Encumbrances which vill remain after settlement (for example, asements on your title and statutory asements for sewerage and drainage thich may not appear on a title search). ailure to disclose these may entitle the uyer to terminate the contract or to ompensation. It is NOT sufficient to state refer to title", "search will reveal", or simil
TI N	enancies: ENANT'S NAME: IL ERM AND OPTIONS:			←If the p Otherv	property is sold with vacan wise complete details from	t possession from settlement, insert 'Nil'. Residential Tenancy Agreement.
	- NIVI AIND OF HONS.					
S'	TARTING DATE OF T	ERM:	ENDING DATE OF TERM	1: RENT: \$	BOND: 	



NAME:				AE	BN:
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
NAME:				AE	BN:
ADDRESS:					-
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
NAME:	ENT (If applicable)				
ABN:			LICENCE NO:	_	
ADDRESS:					
				STATE:	POSTCODE:
SUBURB: PHONE:	MOBILE:	FAX:	EMAIL:	SIAIL	FOSTCODE
BUYER'S SOL	ICITOR			any other solicitor n	otified to the Seller
IAME:	icitor		()	ary carer denotes to	
REF:	CONTACT:				
ADDRESS:					
SUBURB:			<u> </u>	STATE:	POSTCODE:
HONE:	MOBILE:	FAX:	EMAIL:		
PROPERTY					
ot: ADDRESS:	Marine Apartment 1501 C	Couran Cove Island R	esort		
SUBURB:	South Stradbroke			STATE: QLD	POSTCODE: 421
escription: Lot:	Lot 59	on: 🗸 BUP			
Scheme:		Commu	nity Titles Scheme:		<u>.</u>
itle Reference:	50220138				
resent Use:	Residential Unit				







Contract for Residential Lots in a Community Titles Scheme

Twelfth Edition

This document has been approved by The Real Estate Institute of Queensland Limited and the Queensland Law Society Incorporated as being suitable for the sale and purchase of Residential Lots in a Community Titles Scheme in Queensland except for new residential property in which case the issue of GST liability must be dealt with by special condition.

The Sell	er and Buyer	agree to sell and bu	ly the Property u	inder this contract.		
REFER	RENCE SCH	HEDULE				
Contract	t Date:					
SELLER	R'S AGENT					
NAME: Co	balt Blue Res	sources Pty Ltd				
ABN:	50 016			LICENCE NO: 4204690		
ADDRESS:	PO Box 16					
SUBURB:	Toowong		_		STATE: QLD	POSTCODE: 4066
PHONE: 38707203	3	MOBILE: 0421878116	FAX:	EMAIL: rwilson@rwilson.co	om.au	
SELLER NAME: Coralspri					ABN:	
ADDRESS:	2 Bega Road					
SUBURB:	Kingston				STATE: QLD	POSTCODE: 4114
PHONE: 34583408	3	MOBILE:	FAX:	EMAIL: brian.lincez@rising	suncg.com.au	
NAME:					ABN:	
ADDRESS:						
SUBURB:					STATE:	POSTCODE:
PHONE:		MOBILE:	FAX:	EMAIL:		
NAME:	cCullough Law			← or a	ny other solicitor noti	fied to the Buyer
REF:	ccullough Law	CONTACT:				
ADDRESS:	2 Bayliss St	Vicki Partanen				-
SUBURB:	Toowong				STATE: QLD	POSTCODE: 4066
PHONE: 07387082		MOBILE:	FAX:	EMAIL: admin@lockettmcci		



EF003 12/19

ADDITIO	ONAL BODY CORPORATE INFO	RMATION			
Interest S	chedule Lot Entitlement of Lot:	Refer to "Body Corporate Discl	osure Statement"		
Aggregate	e Interest Schedule Lot Entitlement:	Refer to "Body Corporate Disch	osure Statement"		
Contributi	on Schedule Lot Entitlement of Lot:	Refer to "Body Corporate Discl	osure Statement"		
Aggregate	e Contribution Schedule Lot Entitlement:	Refer to "Body Corporate Discl	osure Statement'		
INSURA	NCE POLICIES				
Insurer:	See Body Corporate Disclosure Sta	atement	Policy No:		
Building:					
Public Liab	oility:				
Other:					
		<u></u>			
POOL S	AFETY FOR SHARED POOLS				
Only comp	lete the following questions if there is a sha	red pool on the Land.			
A. Is ther	e a shared pool on the Land or on adj	acent land used in association v	vith the Land?	✓ Yes	☐ No
	answer to A is Yes, is there a Complia ime of contract?	nce or Exemption Certificate for	the shared pool	✓ Yes	☐ No
C. If the a	answer to B is No, has a Notice of No	Pool Safety Certificate been giv	en prior to contract?	Yes	☐ No
Note: This	is an obligation of the Seller under section .	16 of the Building Regulation 2006.			
FLECTE	RICAL SAFETY SWITCH AND SM	OKE ALARM This section mu	ust be completed unless the L	ot is vacant	
The Selle General F	r gives notice to the Buyer that an App Purpose Socket Outlets is: ct whichever is applicable) installed in the residence not installed in the residence		← WARNING: By giving fals information in this section, a penalty. The Seller shot qualified advice about con and not rely on the Seller this section.	, the Seller may uld seek expert mpleting this sed	/ incur and ction
	r gives notice to the Buyer that a Com ct whichever is applicable)	npliant Smoke Alarm(s) is/are:	← WARNING: Failure to inst Alarm is an offence under Services Act 1990.		
\checkmark	installed in the residence				
	not installed in the residence				
NEIGHB	OURHOOD DISPUTES (DIVIDING	G FENCES AND TREES) AC	T 2011		
The Selle of the <i>Nei</i> that the L	ighbourhood Disputes (Dividing Fence	to the Buyer in accordance with section 83 Disputes (Dividing Fences and Trees) Act 2011 WARNING: Failure to comply with section 83 Neighbourhood Disputes (Dividing Fences and Trees Act) 2011 by giving a copy of an order			es and rder or
(sele	ct whichever is applicable)		application to the Buyer (to Buyer signing the cont	ract will entitle t	he Buyer
7	is not affected by any application to, Queensland Civil and Administrative to a tree on the Lot or	or an order made by, the Tribunal (QCAT) in relation	to terminate the contract	prior to Settiem	ent.
	is affected by an application to, or an relation to a tree on the Lot, a copy o to the Buyer prior to the Buyer signin	of which has been given			

GST WI	THHOLDING OBLIGATIONS	
Is the Bu	yer registered for GST and acquiring the Lot for a creditable purpose?	← WARNING: the Buyer warrants in
(sel	ect whichever is applicable)	clause 2.5(6) that this information is true and correct.
	Yes	
\checkmark	No	
the	te: An example of an acquisition for a creditable purpose would be the purchase of Land by a building contractor, who is registered for GST, for the purposes of building ouse on the Land and selling it in the ordinary course of its business.]	
	er gives notice to the Buyer in accordance with section 14-255(1)(a) of the ing Law that:	← WARNING: All sellers of residential premises or potential residential land are required to complete this notice.
(sel	ect whichever is applicable)	Section 14-250 of the Withholding Law applies to the sale of 'new residential
\checkmark	the Buyer <i>is not</i> required to make a payment under section 14-250 of the Withholding Law in relation to the supply of the Property	premises' or 'potential residential land' (subject to some exceptions) and requires an amount to be withheld from
	the Buyer <i>is</i> required to make a payment under section 14-250 of the Withholding Law in relation to the supply of the Property. Under section 14-255(1) of the Withholding Law, the Seller is required to give further details prior to settlement.	the Purchase Price and paid to the ATO. The Seller should seek legal advice if unsure about completing this section.
The	e REIQ Terms of Contract for Residential Lots in a Community Titles Sc Twelfth Edition Contain the Terms of this Contract.	heme (Pages 8-17)
SPECIA	AL CONDITIONS	
	event that the buyer or seller signs the contract and transmits it electronically via	
create a	valid and binding obligation by both parties with the force and effect as if it was an	original.
SETTLE	MENT	
Settlemer	at Date: 30 Days from the Contract date	or the next Business Day if that is not
		 a Business Day in the Place for Settlement.
Place for	Settlement: Brisbane	← If Brisbane is inserted, this is a reference to Brisbane CBD.

INITIAL 000015022438

SIGNATURES

The contract may be subject to a 5 business day statutory cooling-off period. A termination penalty of 0.25% of the purchase price applies if the Buyer terminates the contract during the statutory cooling-off period. It is recommended the Buyer obtain an independent property valuation and independent legal advice about the contract and his or her cooling-off rights, before signing.

Buyer; ×	Witness; [⋆]	HER
Buyer:	Witness:	
By placing my signature above, I warrant that I am the Buyer named in the Reference Schedule or authorised by the Buyer to sign.	(Note: No witness is required i Electronic Signature)	f the Buyer signs using an
Seller; 🔀	Witnes₅; <mark></mark> ✓	SIGN
Seller: By placing my signature above, I warrant that I am the Seller named in the Reference Schedule or authorised by the Seller to sign.	(Note: No witness is required i Electronic Signature)	f the Seller signs using an
Deposit Holder:	← Who acknowledges having received the _ to hold that amount and any Balance De Deposit Holder for the parties as provid	eposit when received as

TERMS OF CONTRACT FOR RESIDENTIAL LOTS IN A COMMUNITY TITLES SCHEME

1. DEFINITIONS

1.1 In this contract:

- terms in **bold** in the Reference Schedule and the Disclosure Statement have the meanings shown opposite them unless the context requires otherwise; and
- (2) (a) "Approved Safety Switch" means a residual current device as defined in the Electrical Safety Regulation 2013;
 - (b) "ATO" means the Australian Taxation Office;
 - (c) "ATO Clearance Certificate" means a certificate issued under section 14-220(1) of the Withholding Law which is current on the date it is given to the Buyer;
 - (d) "Balance Purchase Price" means the Purchase Price, less the Deposit paid by the Buyer, adjusted under clause 2.6;
 - (e) "Bank" means an authorised deposit-taking institution within the meaning of the Banking Act 1959 (Cth).
 - (f) "Body Corporate" means the body corporate of the Scheme;
 - (g) "Body Corporate Debt" has the meaning in the Regulation Module but excludes the Body Corporate Levies for the period which includes the Settlement Date;
 - (h) "Body Corporate Levies" means regular periodic contributions levied on the owner of the Lot (including, if applicable, levied under an exclusive use by-law) excluding any Special Contribution:
 - "Bond" means a bond under the Residential Tenancies and Rooming Accommodation Act 2008;
 - "Building" means any building that forms part of the Lot or in which the Lot is situated;
 - (k) "Building Inspector" means a person licensed to carry out completed residential building inspections under the Queensland Building and Construction Commission Regulations 2003:
 - (I) "Business Day" means a day other than:
 - (i) a Saturday or Sunday
 - (ii) a public holiday in the Place for Settlement; and
 - (iii) a day in the period 27 to 31 December (inclusive).
 - (m) "CGT Withholding Amount" means the amount determined under section 14-200(3)(a) of the Withholding Law or, if a copy is provided to the Buyer prior to settlement, a lesser amount specified in a variation notice under section 14-235;
 - (n) "Compliance or Exemption Certificate" means:
 - (i) a Pool Safety Certificate; or
 - (ii) a building certificate that may be used instead of a Pool Safety Certificate under section 246AN(2) of the Building Act 1975; or
 - (iii) an exemption from compliance on the grounds of impracticality under section 245B of the *Building Act 1975*;
 - (o) "Compliant Smoke Alarm" means a smoke alarm complying with the requirements for smoke alarms in domestic dwellings under the Fire and Emergency Services Act 1990;
 - (p) "Contract Date" or "Date of Contract" means the date inserted in the Reference Schedule;
 - (q) "Court" includes any tribunal established under statute;

- (r) "Disclosure Statement" means the statement under section 206 (Existing Lot) or section 213 (Proposed Lot) of the Body Corporate and Community Management Act 1997;
- (s) "Electronic Signature" means an electronic method of signing that identifies the person and indicates their intention to sign the contract;
- (t) "Encumbrances" includes:
 - (i) unregistered encumbrances
 - (ii) statutory encumbrances; and
 - (iii) Security Interests.
- (u) "Essential Term" includes, in the case of breach by:
 - (i) the Buyer: clauses 2.2, 2.5(1), 2.5(5),5.1 and 6.1; and
 - (ii) the Seller: clauses 2.5(5), 5.1, 5.3(1)(a) (d), 5.3(1)(e)(ii) & (iii), 5.3(1)(f), 5.5 and 6.1; but nothing in this definition precludes a Court from finding other terms to be essential;
- (v) "Exclusive Use Areas" means parts of the common property for the Scheme allocated to the Lot under an exclusive use by-law;
- (w) "Financial Institution" means a Bank, Building Society or Credit Union;
- (x) "General Purpose Socket Outlet" means an electrical socket outlet as defined in the Electrical Safety Regulation 2013;
- (y) "GST" means the goods and services tax under the GST Act;
- (z) "GST Act" means A New Tax System (Goods and Services Tax) Act 1999 (Cth) and includes other GST related legislation;
- (aa) "GST Withholding Amount" means the amount (if any) determined under section 14-250 of the Withholding Law required to be paid to the Commissioner of Taxation;
- (bb) "Improvements" means fixed structures in the Lot (such as stoves, hot water systems, fixed carpets, curtains, blinds and their fittings, clothes lines, fixed satellite dishes and television antennae, in-ground plants) but does not include the Reserved Items;
- (cc) "Keys" means keys, codes or devices in the Seller's possession or control for all locks or security systems on the Property or necessary to access the Property;
- (dd) "Land" means the scheme land for the Scheme;
- (ee) "Notice of No Pool Safety Certificate" means the Form 36 under the Building Regulation 2006 to the effect that there is no Pool Safety Certificate issued for the Land and/or the Lot;
- (ff) "Notice of nonconformity" means a Form 26 under the *Building Regulation 2006* advising how the pool does not comply with the relevant pool safety standard;
- (gg) "Outgoings" means:
 - rates or charges on the Lot by any competent authority (for example, council rates, water rates, fire service levies) but excludes land tax; and
 - (ii) Body Corporate Levies.
- (hh) "Pest Inspector" means a person licensed to undertake termite inspections on completed buildings under the Queensland Building and Construction Commission Regulations 2003;
- (ii) "Pool Safety Certificate" has the meaning in section 231C(a) of the Building Act 1975;



- "Pool Safety Requirements" means the requirements for pool safety contained in the Building Act 1975 and Building Regulation 2006;
- (kk) "Pool Safety Inspection Date" means the Pool Safety Inspection Date inserted in the Reference Schedule. If no date is inserted in the Reference Schedule, the Pool Safety Inspection Date is taken to be the earlier of the following:
 - (i) the Inspection Date for the Building and/or Pest Inspection; or
 - (ii) 2 Business Days before the Settlement Date;
- (II) "Pool Safety Inspector" means a person authorised to give a Pool Safety Certificate;
- (mm) "PPSR" means the Personal Property Securities Register established under Personal Property Securities Act 2009 (Cth);
- (nn) "Property" means:
 - (i) the Lot;
 - (ii) the right to any Exclusive Use Areas;
 - (iii) the Improvements;
 - (iv) the Included Chattels;
- (oo) "Regulation Module" means the regulation module for the Scheme;
- (pp) "Rent" means any periodic amount, including outgoings, payable under the Tenancies;
- (qq) "Reserved Items" means the Excluded Fixtures and all Chattels in the Lot and Exclusive Use Areas other than the Included Chattels;
- (rr) "Scheme" means the community titles scheme containing the Lot;
- (ss) "Security Interests" means all security interests registered on the PPSR over Included Chattels and Improvements;
- (tt) "Special Contribution" means an amount levied by the Body Corporate under the Regulation Module for a liability for which no provision or inadequate provision has been made in the budget of the Body Corporate.
- (uu) "Transfer Documents" means:
 - (i) the form of transfer under the Land Title Act 1994 required to transfer title in the Lot to the Buyer; and
 - (ii) any other document to be signed by the Seller necessary for stamping or registering the transfer;
- (vv) "Transport Infrastructure" has the meaning defined in the Transport Infrastructure Act 1994; and
- (ww) "Withholding Law" means Schedule 1 to the Taxation Administration Act 1953 (Cth).
- **1.2** Words and phrases defined in the *Body Corporate and Community Management Act 1997* have the same meaning in this contract unless the context indicates otherwise.

2. PURCHASE PRICE

2.1 **GST**

- Unless otherwise specified in this contract, the Purchase Price includes any GST payable on the supply of the Property to the Buyer.
- (2) If a party is required to make any other payment or reimbursement under this contract, that payment or reimbursement will be reduced by the amount of any input tax credits to which the other party (or the representative member for a GST group of which it is a member) is entitled.

2.2 Deposit

(1) The Buyer must pay the Deposit to the Deposit Holder at the times shown in the Reference Schedule. The Deposit Holder will hold the Deposit until a party becomes entitled to it.

- (2) The Buyer will be in default if it:
 - (a) does not pay the Deposit when required;
 - (b) pays the Deposit by a post-dated cheque; or
 - (c) pays the Deposit by cheque which is dishonoured on presentation.
- (3) The Seller may recover from the Buyer as a liquidated debt any part of the Deposit which is not paid when required.

2.3 Investment of Deposit

lf:

- the Deposit Holder is instructed by either the Seller or the Buyer; and
- (2) it is lawful to do so;

the Deposit Holder must:

- (3) invest as much of the Deposit as has been paid with any Financial Institution in an interest-bearing account in the names of the parties; and
- (4) provide the parties' tax file numbers to the Financial Institution (if they have been supplied).

2.4 Entitlement to Deposit and Interest

- (1) The party entitled to receive the Deposit is:
 - (a) if this contract settles, the Seller;
 - (b) if this contract is terminated without default by the Buyer, the Buyer;
 - (c) if this contract is terminated owing to the Buyer's default, the Seller.
- (2) The interest on the Deposit must be paid to the person who is entitled to the Deposit.
- (3) If this contract is terminated, the Buyer has no further claim once it receives the Deposit and interest unless the termination is due to the Seller's default or breach of warranty.
- (4) The Deposit is invested at the risk of the party who is ultimately entitled to it.

2.5 Payment of Balance Purchase Price

- (1) On the Settlement Date, the Buyer must pay the Balance Purchase Price by Bank cheque as the Seller or the Seller's Solicitor directs.
- (2) Despite any other provision of this contract, reference to a "Bank cheque" in clause 2.5:
 - (a) includes a cheque drawn by a Building Society or Credit Union on itself;
 - (b) does not include a cheque drawn by a Building Society or Credit Union on a Bank;
 - and the Seller is not obliged to accept a cheque referred to in clause 2.5(2)(b) on the Settlement Date.
- (3) If both the following apply:
 - (a) the sale is not an excluded transaction under section 14-215 of the Withholding Law; and
 - (b) the Seller has not given the Buyer on or before settlement for each person comprising the Seller either.
 - (i) an ATO Clearance Certificate; or
 - (ii) a variation notice under section 14-235 of the Withholding Law which remains current at the Settlement Date varying the CGT Withholding Amount to nil,

then:

(c) for clause 2.5(1), the Seller irrevocably directs the Buyer to draw a bank cheque for the CGT Withholding Amount in favour of the Commissioner of Taxation or, if the Buyer's Solicitor requests, the Buyer's Solicitor's Trust Account;

- (d) the Buyer must lodge a Foreign Resident Capital Gains Withholding Purchaser Notification Form with the ATO for each person comprising the Buyer and give copies to the Seller with the payment reference numbers (PRN) on or before settlement;
- (e) the Seller must return the bank cheque in paragraph (c) to the Buyer's Solicitor (or if there is no Buyer's Solicitor, the Buyer) at settlement; and
- (f) the Buyer must pay the CGT Withholding Amount to the ATO in accordance with section 14-200 of the Withholding Law and give the Seller evidence that it has done so within 2 Business Days of settlement occurring.
- (4) For clause 2.5(3) and section 14-215 of the Withholding Law, the market value of the CGT asset is taken to be the Purchase Price less any GST included in the Purchase Price for which the Buyer is entitled to an input tax credit unless:
 - (a) the Property includes items in addition to the Lot and Improvements; and
 - (b) no later than 2 Business Days prior to the Settlement Date, the Seller gives the Buyer a valuation of the Lot and Improvements prepared by a registered valuer,

in which case the market value of the Lot and Improvements will be as stated in the valuation.

- (5) If the Buyer is required to pay the GST Withholding Amount to the Commissioner of Taxation at settlement pursuant to section 14-250 of the Withholding Law:
 - (a) the Seller must give the Buyer a notice in accordance with section 14-255(1) of the Withholding Law;
 - (b) prior to settlement the Buyer must lodge with the ATO:
 - a GST Property Settlement Withholding Notification form ("Form 1"); and
 - (ii) a GST Property Settlement Date Confirmation form ("Form 2");
 - (c) on or before settlement, the Buyer must give the Seller copies of:
 - (i) the Form 1;
 - (ii) confirmation from the ATO that the Form 1 has been lodged specifying the Buyer's lodgement reference number and payment reference number;
 - (iii) confirmation from the ATO that the Form 2 has been lodged; and
 - (iv) a completed ATO payment slip for the Withholding Amount.
 - (d) the Seller irrevocably directs the Buyer to draw a bank cheque for the GST Withholding Amount in favour of the Commissioner of Taxation and deliver it to the Seller at settlement; and
 - (e) the Seller must pay the GST Withholding Amount to the ATO in compliance with section 14-250 of the Withholding Law promptly after settlement.
- (6) The Buyer warrants that the statements made by the Buyer in the Reference Schedule under GST Withholding Obligations are true and correct.

2.6 Adjustments to Balance Purchase Price

(1) The Seller is liable for Outgoings and is entitled to Rent up to and including the Settlement Date. The Buyer is liable for Outgoings and is entitled to Rent after the Settlement Date.

- (2) Subject to clauses 2.6(3), 2.6(5) and 2.6(17), Outgoings for periods including the Settlement Date must be adjusted:
 - (a) for those paid, on the amount paid;
 - (b) for those assessed but unpaid, on the amount payable (excluding any discount); and
 - (c) for those not assessed:
 - on the amount the relevant authority or the Body Corporate advises will be assessed (excluding any discount); or
 - (ii) if no advice on the assessment to be made is available, on the amount of the latest assessment (excluding any discount).
- (3) If there is no separate assessment of rates for the Lot at the Settlement Date and the Local Government informs the Buyer that it will not apportion rates between the Buyer and the Seller, then:
 - (a) the amount of rates to be adjusted is that proportion of the assessment equal to the ratio of the interest schedule lot entitlement of the Lot to the aggregate interest schedule lot entitlement of the Scheme; and
 - (b) if an assessment of rates includes charges imposed on a "per lot" basis, then the portion of those charges to be adjusted is the amount assessed divided by the number of lots in that assessment.
- (4) The Seller is liable for land tax assessed on the Lot for the financial year current at the Settlement Date. If land tax is unpaid at the Settlement Date and the Office of State Revenue advises that it will issue a final clearance for the Lot on payment of a specified amount, then the Buyer may deduct the specified amount from the Balance Purchase Price at settlement and must pay it promptly to the Office of State Revenue.
- (5) Any Outgoings assessable on the amount of water used must be adjusted on the charges that would be assessed on the total water usage for the assessment period, determined by assuming that the actual rate of usage shown by the meter reading made before settlement continues throughout the assessment period. The Buyer must obtain and pay for the meter reading.
- (6) If any Outgoings are assessed but unpaid at the Settlement Date, then the Buyer may deduct the amount payable from the Balance Purchase Price at settlement and pay it promptly to the relevant authority or the Body Corporate, as appropriate. If an amount is deducted under this clause, the relevant Outgoing will be treated as paid at the Settlement Date for the purposes of clause 2.6(2).
- (7) Arrears of Rent for any rental period ending on or before the Settlement Date belong to the Seller and are not adjusted at settlement.
- (8) Unpaid Rent for the rental period including both the Settlement Date and the following day ("Current Period") is not adjusted until it is paid.
- (9) Rent already paid for the Current Period or beyond must be adjusted at settlement.
- (10) If Rent payments are reassessed after the Settlement Date for periods including the Settlement Date, any additional Rent payment from a Tenant or refund due to a Tenant must be apportioned under clauses 2.6(7), 2.6(8) and 2.6(9) and 2.6(10).
- (11) Payments under clause 2.6(10) must be made within 14 days after notification by one party to the other but only after any additional payment from a Tenant has been received.

- (12) The Seller is liable for:
 - (a) any Special Contribution for which a levy notice has been issued on or before the Contract Date;
 - (b) any other Body Corporate Debt (including any penalty or recovery cost resulting from nonpayment of a Body Corporate Debt) owing in respect of the Lot at settlement.

The Buyer is liable for any Special Contribution levied after the Contract Date.

- (13) If an amount payable by the Seller under clause 2.6(12) is unpaid at the Settlement Date, the Buyer may deduct the specified amount from the Balance Purchase Price at settlement and must pay it promptly to the Body Corporate.
- (14) For the purposes of clause 2.6(12), an amount payable under an exclusive use by-law will be treated as levied on the date it is due.
- (15) The cost of Bank cheques payable at settlement:
 - (a) to the Seller or its mortgagee are the responsibility of the Buyer; and
 - (b) to parties other than the Seller or its mortgagee are the responsibility of the Seller.
- (16) The Seller is not entitled to require payment of the Balance Purchase Price by means other than Bank cheque without the consent of the Buyer.
- (17) Upon written request by the Buyer, the Seller will, prior to settlement, give the Buyer a written statement, supported by reasonable evidence, of —
 - (a) all Outgoings and all Rent for the Property to the extent they are not capable of discovery by search or enquiry at any office of public record or pursuant to the provisions of any statute; and
 - (b) any other information which the Buyer may reasonably require for the purpose of calculating or apportioning any Outgoings or Rent under this clause 2.6.

If the Seller becomes aware of a change to the information provided the Seller will as soon as practicably provide the updated information to the Buyer.

3. FINANCE

- 3.1 This contract is conditional on the Buyer obtaining approval of a loan for the Finance Amount from the Financier by the Finance Date on terms satisfactory to the Buyer. The Buyer must take all reasonable steps to obtain approval.
- 3.2 The Buyer must give notice to the Seller that:
 - approval has not been obtained by the Finance Date and the Buyer terminates this contract; or
 - the finance condition has been either satisfied or waived by the Buyer.
- 3.3 The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 3.2 by 5pm on the Finance Date. This is the Seller's only remedy for the Buyer's failure to give notice.
- 3.4 The Seller's right under clause 3.3 is subject to the Buyer's continuing right to give written notice to the Seller of satisfaction, termination or waiver pursuant to clause 3.2.

4. BUILDING AND PEST IN:SPECTION REPORTS AND POOL SAFETY

4.1 Building and Pest Inspection

(1) This contract is conditional on the Buyer obtaining a written building report from a Building Inspector and a written pest report from a Pest Inspector (which may be a single report) on the Property by the Inspection Date on terms satisfactory to the Buyer. The Buyer must take all reasonable steps to obtain the reports (subject to the right of the Buyer to elect to obtain only one of the reports).

- (2) The Buyer must give notice to the Seller that:
 - (a) a satisfactory Inspector's report under clause 4.1(1) has not been obtained by the Inspection Date and the Buyer terminates this contract. The Buyer must act reasonably; or
 - (b) clause 4.1(1) has been either satisfied or waived by the Buyer.
- (3) If the Buyer terminates this contract and the Seller asks the Buyer for a copy of the building and pest reports, the Buyer must give a copy of each report to the Seller without delay.
- (4) The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 4.1(2) by 5pm on the Inspection Date. This is the Seller's only remedy for the Buyer's failure to give notice.
- (5) The Seller's right under clause 4.1(4) is subject to the Buyer's continuing right to give written notice to the Seller of satisfaction, termination or waiver pursuant to clause 4.1(2).

4.2 Pool Safety for non-shared pool on Lot

- (1) This clause 4.2 applies if:
 - (a) there is a pool on the Lot and the answer to Q2 of the Reference Schedule is No or Q2 is not completed (for a non-shared pool on the Lot);
 and
 - (b) this contract is not a contract of a type referred to in section 160(1)(b) of the Property Occupations Act 2014.
- (2) This contract is conditional upon:
 - (a) the issue of a Pool Safety Certificate; or
 - (b) a Pool Safety Inspector issuing a Notice of Nonconformity stating the works required before a Pool Safety Certificate can be issued;
 - by the Pool Safety Inspection Date.
- (3) The Buyer is responsible for arranging an inspection by a Pool Safety Inspector at the Buyer's cost. The Seller authorises:
 - (a) the Buyer to arrange the inspection; and
 - (b) the Pool Safety Inspector to advise the Buyer of the results of the inspection and to give the Buyer a copy of any notice issued.
- (4) If a Pool Safety Certificate has not been issued by the Pool Safety Inspection Date, the Buyer may give notice to the Seller that the Buyer:
 - (a) terminates this contract; or
 - (b) waives the benefit of this clause 4.2

The Buyer must act reasonably.

- (5) The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 4.2(4) by 5pm on the Pool Safety Inspection Date.
- (6) The Seller's right under clause 4.2(5) is subject to the Buyer's continuing right to give written notice to the Seller of termination or waiver pursuant to clause 4.2(4).
- (7) The right of a party to terminate under this clause 4.2, ceases upon receipt by that party of a copy of a current Pool Safety Certificate.
- (8) If the Buyer terminates this contract under clause 4.2(4)(a), and the Seller has not obtained a copy of the Notice of Nonconformity issued by the Pool Safety Inspector, the Seller may request a copy and the Buyer must provide this to the Seller without delay.

5. SETTLEMENT

5.1 Time and Date

 Settlement must occur between 9am and 4pm AEST on the Settlement Date.

INITIALS (Note: initials not required if signed with Electronic Signature)

000015022438

(2) If the parties do not agree on where settlement is to occur, it must take place in the Place for Settlement at the office of a Solicitor or Financial Institution nominated by the Seller, or, if the Seller does not make a nomination, at the Land Registry Office in or nearest to the Place for Settlement.

5.2 Transfer Documents

- (1) The Transfer Documents must be prepared by the Buyer's Solicitor and delivered to the Seller a reasonable time before the Settlement Date.
- (2) If the Buyer pays the Seller's reasonable expenses, it may require the Seller to produce the Transfer Documents at the Office of State Revenue nearest the Place for Settlement for stamping before settlement.

5.3 Documents and Keys at Settlement

- In exchange for payment of the Balance Purchase Price, the Seller must deliver to the Buyer at settlement:
 - (a) any instrument of title for the Lot required to register the transfer to the Buyer; and
 - (b) unstamped Transfer Documents capable of immediate registration after stamping; and
 - (c) any instrument necessary to release any Encumbrance over the Property in compliance with the Seller's obligation in clause 7.2; and
 - (d) if requested by the Buyer not less than 2 clear Business Days before the Settlement Date, the Keys; and
 - (e) if there are Tenancies:
 - (i) the Seller's copy of any Tenancy agreements;
 - (ii) a notice to each tenant advising of the sale in the form required by law; and
 - (iii) any notice required by law to transfer to the Buyer the Seller's interest in any Bond; and
 - (f) if the answer to Q2 in the Reference Schedule is Yes, a copy of a current Compliance or Exemption Certificate, if not already provided to the Buyer.
- (2) If the Keys are not delivered at settlement under clause 5.3(1)(d), the Seller must deliver the Keys to the Buyer. The Seller may discharge its obligation under this provision by authorising the Seller's Agent to release the Keys to the Buyer.

5.4 Assignment of Covenants and Warranties

At settlement, the Seller assigns to the Buyer the benefit of all:

- (1) covenants by the tenants under the Tenancies;
- (2) guarantees and Bonds (subject to the requirements of the Residential Tenancies and Rooming Accommodation Act 2008) supporting the Tenancies; and
- (3) manufacturers' warranties regarding the Included Chattels;
- (4) builders' warranties on the Improvements; to the extent that they are assignable and the Buyer accepts the assignment. However, the right to recover arrears of Rent is not assigned to the Buyer and section 117 of the *Property Law Act 1974* does not apply.

5.5 Possession of Property and Title to Included Chattels

On the Settlement Date, in exchange for the Balance Purchase Price, the Seller must give the Buyer vacant possession of the Lot and Exclusive Use Areas except for the Tenancies. Title to the Included Chattels passes at settlement.

5.6 Reservations

 The Seller must remove the Reserved Items from the Property before settlement.

- (2) The Seller must repair at its expense any damage done to the Property in removing the Reserved Items. If the Seller fails to do so, the Buyer may repair that damage.
- (3) Any Reserved Items not removed before settlement will be considered abandoned and the Buyer may, without limiting its other rights, complete this contract and appropriate those Reserved Items or dispose of them in any way.
- (4) The Seller indemnifies the Buyer against any damages and expenses resulting from the Buyer's actions under clauses 5.6(2) or 5.6(3).

6. TIME

6.1 Time of the Essence

Time is of the essence of this contract, except regarding any agreement between the parties on a time of day for settlement.

6.2 Suspension of Time

- (1) This clause 6.2 applies if a party is unable to perform a settlement obligation solely as a consequence of a Delay Event but does not apply where the inability is attributable to: —
 - (a) damage to, destruction of or diminution in value of the Property or other property of the Seller or Buyer; or
 - (b) termination or variation of any agreement between a party and another person whether relating to the provision of finance, the release of an Encumbrance, the sale or purchase of another property or otherwise.
- (2) Time for the performance of the parties' Settlement Obligations is suspended and ceases to be of the essence of the contract and the parties are deemed not to be in breach of their Settlement Obligations.
- (3) An Affected Party must take reasonable steps to minimise the effect of the Delay Event on its ability to perform its Settlement Obligations.
- (4) When an Affected Party is no longer prevented from performing its Settlement Obligations due to the Delay Event, the Affected Party must give the other party a notice of that fact, promptly.
- (5) When the Suspension Period ends, whether notice under clause 6.2(4) has been given or not, either party may give the other party a Notice to Settle.
- (6) A Notice to Settle must be in writing and state:
 - (a) that the Suspension Period has ended; and
 - (b) a date, being not less than 5 nor more than 10 Business Days after the date the Notice to Settle is given, which shall become the Settlement Date;
 - (c) that time is of the essence.
- (7) When Notice to Settle is given, time is again of the essence of the contract.
- (8) In this clause 6.2:
 - (a) "Affected Party" means a party referred to in clause 6.2(1);
 - (b) "Delay Event" means:
 - a tsunami, flood, cyclone, earthquake, bushfire or other act of nature;
 - (ii) riot, civil commotion, war, invasion or a terrorist act;
 - (iii) an imminent threat of an event in paragraphs (i) or (ii); or
 - (iv) compliance with any lawful direction or order by a Government Agency;

- (c) "Government Agency" means the government of the Commonwealth of Australia or an Australian State, Territory or local government and includes their authorities, agencies, government owned corporations and authorised officers, courts and tribunals;
- (d) "Settlement Obligations" means, in the case of the Buyer, its obligations under clauses 2.5(1) and 5.1(1) and, in the case of the Seller, its obligations under clauses 5.1(1), 5.3(1)(a) – (e) and 5.5;
- (e) "Suspension Period" means the period during which the Affected Party (or if both the Buyer and Seller are Affected Parties, either of them) remains unable to perform a settlement obligation solely as a consequence of a Delay Event.

7. MATTERS AFFECTING THE PROPERTY

7.1 Title

The Lot is sold subject to:

- (1) the Body Corporate and Community Management Act 1997 and the by-Laws of the Body Corporate;
- (2) any reservations or conditions on the title or the original Deed of Grant.

7.2 Encumbrances

The Property is sold free of all Encumbrances other than the Title Encumbrances, Tenancies, statutory easements implied by part 6A of the *Land Title Act 1994* and interests registered on the common property for the Scheme.

7.3 Requisitions

The Buyer may not deliver any requisitions or enquiries on title.

7.4 Seller's Warranties

- (1) The Seller warrants that, except as disclosed in this contract, at settlement:
 - it will be the registered owner of an estate in fee simple in the Lot and will own the Improvements and Included Chattels;
 - it will be capable of completing this contract (unless the Seller dies or becomes mentally incapable after the Contract Date); and
 - (c) there will be no unsatisfied judgment, order (except for an order referred to in clause 7.6(1)(b)) or writ affecting the Property.
- (2) The Seller warrants that, except as disclosed in this contract, at the Contract Date and at settlement there are no current or threatened claims, notices or proceedings that may lead to a judgment, order or writ affecting the Property.
- (3) The Seller warrants that, except as disclosed in this contract, at the Contract Date:
 - (a) there is no unregistered lease, easement or other right capable of registration and which is required to be registered to give indefeasibility affecting the common property or Body Corporate assets;
 - (b) there is no proposal to record a new community management statement for the Scheme and it has not received a notice of a meeting of the Body Corporate to be held after the Contract Date or notice of any proposed resolution or a decision of the Body Corporate to consent to the recording of a new community management statement for the Scheme;
 - (c) all Body Corporate consents to improvements made to common property and which benefit the Lot, or the registered owner of the Lot, are in force; and

- (d) the Additional Body Corporate Information is correct (if completed).
- (4) If the Seller breaches a warranty in clause 7.4(1) or clause 7.4(2), the Buyer may terminate this contract by notice to the Seller.
- (5) If:
 - (a) the Seller breaches a warranty in clause 7.4(3);
 - (b) the Additional Body Corporate Information is not completed;

and, as a result, the Buyer is materially prejudiced, the Buyer may terminate this contract by notice to the Seller given within 14 days after the Contract Date but may not claim damages or compensation.

- (6) Clauses 7.4(4) and 7.4(5) do not restrict any statutory rights the Buyer may have which cannot be excluded by this contract.
- (7) (a) The Seller warrants that, except as disclosed in this contract or a notice given by the Seller to the Buyer under the *Environmental Protection Act 1994* ("**EPA**"), at the Contract Date:
 - there is no outstanding obligation on the Seller to give notice to the administering authority under EPA of notifiable activity being conducted on the Land; and
 - (ii) the Seller is not aware of any facts or circumstances that may lead to the Land being classified as contaminated land within the meaning of EPA.
 - (b) If the Seller breaches a warranty in clause 7.4(7), the Buyer may:
 - terminate this contract by notice in writing to the Seller given no later than 2 Business Days before the Settlement Date; or
 - (ii) complete this contract and claim compensation, but only if the Buyer claims it in writing before the Settlement Date.
- (8) The Seller does not warrant that the Present Use is lawful.

7.5 Survey and Mistake

- (1) The Buyer may survey the Lot.
- (2) If there is:
 - (a) an error in the boundaries or area of the Lot;
 - (b) an encroachment by structures onto or from the
 - (c) a mistake or omission in describing the Lot or the Seller's title to it;

which is:

- (d) immaterial; or
- (e) material, but the Buyer elects to complete this contract;

the Buyer's only remedy against the Seller is for compensation, but only if claimed by the Buyer in writing on or before settlement.

- (3) The Buyer may not delay settlement or withhold any part of the Balance Purchase Price because of any compensation claim under clause 7.5(2).
- (4) If there is a material error, encroachment or mistake, the Buyer may terminate this contract before settlement.

7.6 Requirements of Authorities

- (1) Subject to clause 7.6(5), any valid notice or order by any competent authority or Court requiring work to be done or money spent in relation to the Property ("Work or Expenditure") must be fully complied with:
 - (a) if issued before the Contract Date, by the Seller before the Settlement Date;

- (b) if issued on or after the Contract Date, by the Buyer.
- (2) If any Work or Expenditure that is the Seller's responsibility under clause 7.6(1)(a) is not done before the Settlement Date, the Buyer is entitled to claim the reasonable cost of work done by the Buyer in accordance with the notice or order referred to in clause 7.6(1) from the Seller after settlement as a debt
- (3) Any Work or Expenditure that is the Buyer's responsibility under clause 7.6(1)(b), which is required to be done before the Settlement Date, must be done by the Seller unless the Buyer directs the Seller not to and indemnifies the Seller against any liability for not carrying out the work. If the Seller does the work, or spends the money, the reasonable cost of that Work or Expenditure must be added to the Balance Purchase Price.
- (4) The Buyer may terminate this contract by notice to the Seller if there is an outstanding notice at the Contract Date under section 246AG of the Building Act 1975 that affects the Property. The Buyer may terminate this contract by notice to the Seller if there is an outstanding notice at the Contract Date under sections 247 or 248 of the Building Act 1975 or sections 167 or 168 of the Planning Act 2016 that affects the Property or Land.
- (5) Clause 7.6(1) does not apply to orders disclosed under section 83 of the Neighbourhood Disputes (Dividing Fences and Trees) Act 2011.

7.7 Property Adversely Affected

- (1) If at the Contract Date:
 - (a) the Present Use is not lawful under the relevant town planning scheme;
 - (b) the Land is affected by a proposal of any competent authority to alter the dimensions of any Transport Infrastructure or locate Transport Infrastructure on the Land;
 - (c) access or any service to the Land passes unlawfully through other land;
 - (d) any competent authority has issued a current notice to treat, or notice of intention to resume, regarding any part of the Land;
 - (e) there is an outstanding condition of a development approval attaching to the Land under section 73 of the Planning Act 2016 or section 96 of the Economic Development Queensland Act 2012 which, if complied with, would constitute a material mistake or omission in the Seller's title under clause 7.5(2)(c);
 - (f) the Property is affected by the Queensland Heritage Act 1992 or is included in the World Heritage List;
 - (g) the Property is declared acquisition land under the Queensland Reconstruction Authority Act 2011: or
 - (h) there is a charge against the Lot under section 104 of the *Foreign Acquisitions and Takeovers Act 1975* (Cth),
 - and that has not been disclosed in this contract, the Buyer may terminate this contract by notice to the Seller given on or before settlement.
- (2) If no notice is given under clause 7.7(1), the Buyer will be treated as having accepted the Property subject to all of the matters referred to in that clause.
- (3) The Seller authorises the Buyer to:
 - (a) inspect records held by any authority, including Security Interests on the PPSR, relating to the Property or the Land; and
 - (b) apply for a certificate of currency of the Body Corporate's insurance from any insurer.

7.8 Dividing Fences

Notwithstanding any provision in the *Neighbourhood Disputes (Dividing Fences and Trees) Act 2011*, the Seller need not contribute to the cost of building any dividing fence between the Lot and any adjoining land owned by it. The Buyer waives any right to claim contribution from the Seller.

8. RIGHTS AND OBLIGATIONS UNTIL SETTLEMENT

8.1 Risk

The Property is at the Buyer's risk from 5pm on the first Business Day after the Contract Date.

8.2 Access

After reasonable notice to the Seller, the Buyer and its consultants may enter the Property:

- (1) once to read any meter;
- (2) for inspections under clause 4;
- (3) once to inspect the Property before settlement; and
- (4) once to value the Property before settlement.

8.3 Seller's Obligations After Contract Date

- (1) The Seller must use the Property reasonably until settlement. The Seller must not do anything regarding the Property or Tenancies that may significantly alter them or result in later expense for the Buyer.
- (2) The Seller must promptly upon receiving any notice, proceeding or order that affects the Property or requires work on the Property, give a copy to the Buyer.
- (3) Without limiting clause 8.3(1), the Seller must not without the prior written consent of the Buyer, give any notice or seek or consent to any order that affects the Property or make any agreement affecting the Property that binds the Buyer to perform.

8.4 Body Corporate Meetings

- (1) The Seller must promptly give the Buyer a copy of:
 - (a) any notice it receives of a proposed meeting of the Body Corporate to be held after the Contract Date; and
 - (b) resolutions passed at that meeting and prior to settlement.
- (2) The Buyer may terminate this contract by notice in writing to the Seller given before settlement if it is materially prejudiced by:
 - (a) any resolution of the Body Corporate passed after the Contract Date, other than a resolution, details of which are disclosed to the Buyer in this contract: or
 - (b) where the Scheme is a subsidiary scheme, any resolution of a Body Corporate of a higher scheme.
- (3) In clause 8.4(2) a resolution includes a decision of the Body Corporate Committee to consent to recording a new community management statement.
- (4) If the Buyer is not given a copy of the resolutions before settlement, it may sue the Seller for damages.

8.5 Information Regarding the Property

Upon written request of the Buyer but in any event before settlement, the Seller must give the Buyer:

- copies of all documents relating to any unregistered interests in the Property;
- (2) full details of the Tenancies to allow the Buyer to properly manage the Property after settlement;
- (3) sufficient details (including the date of birth of each Seller who is an individual) to enable the Buyer to undertake a search of the PPSR;



8.6 Possession Before Settlement

If poss'ession is given before settlement:

- the Buyer must maintain the Property in substantially its condition at the date of possession, fair wear and tear excepted;
- (2) entry into possession is under a licence personal to the Buyer revocable at any time and does not:
 - (a) create a relationship of landlord and tenant; or
 - (b) waive the Buyer's rights under this contract;
- (3) the Buyer must insure the Property to the Seller's satisfaction; and
- (4) the Buyer indemnifies the Seller against any expense or damages incurred by the Seller as a result of the Buyer's possession of the Property.

8.7 Seller's Obligations After Contract Date

- the Seller must promptly upon receiving any notice, proceeding or order that affects the Property or requires work on the Property, give a copy to the Buyer.
- (2) After the Contract Date, the Seller must not without the prior written consent of the Buyer, give any notice, seek or consent to any order or make an agreement that affects the Property.

9. PARTIES' DEFAULT

9.1 Seller and Buyer May Affirm or Terminate

Without limiting any other right or remedy of the parties including those under this contract, or any right at law or in equity, if the Seller or Buyer, as the case may be, fails to comply with an Essential Term, or makes a fundamental breach of an intermediate term, the Seller (in the case of the Buyer's default) or the Buyer (in the case of the Seller's default) may affirm or terminate this contract.

9.2 If Seller Affirms

If the Seller affirms this contract under clause 9.1, it may sue the Buyer for:

- (1) damages;
- (2) specific performance; or
- (3) damages and specific performance.

9.3 If Buyer Affirms

If the Buyer affirms this contract under clause 9.1, it may sue the Seller for:

- (1) damages;
- (2) specific performance; or
- (3) damages and specific performance.

9.4 If Seller Terminates

If the Seller terminates this contract under clause 9.1, it may do all or any of the following:

- (1) resume possession of the Property;
- (2) forfeit the Deposit and any interest earned;
- (3) sue the Buyer for damages;
- (4) resell the Property.

9.5 If Buyer Terminates

If the Buyer terminates this contract under clause 9.1, it may do all or any of the following:

- (1) recover the Deposit and any interest earned;
- (2) sue the Seller for damages.

9.6 Seller's Resale

- (1) If the Seller terminates this contract and resells the Property, the Seller may recover from the Buyer as liquidated damages:
 - (a) any deficiency in price on a resale; and
 - (b) its expenses connected with any repossession, any failed attempt to resell, and the resale,

provided the resale settles within 2 years of termination of this contract.

(2) Any profit on a resale belongs to the Seller.

9.7 Seller's Damages

The Seller may claim damages for any loss it suffers as a result of the Buyer's default, including its legal costs on an indemnity basis and the cost of any Work or Expenditure under clause 7.6(3).

9.8 Buyer's Damages

The Buyer may claim damages for any loss it suffers as a result of the Seller's default, including its legal costs on an indemnity basis.

9.9 Interest on Late Payments

- (1) The Buyer must pay interest at the Default Rate:
 - (a) on any amount payable under this contract which is not paid when due; and
 - (b) on any judgement for money payable under this contract.
- (2) Interest continues to accrue:
 - (a) under clause 9.9(1)(a), from the date it is due until paid; and
 - (b) under clause 9.9(1)(b), from the date of judgement until paid.
- (3) Any amount payable under clause 9.9(1)(a) in respect of a period prior to settlement must be paid by the Buyer at settlement. If this contract is terminated or if any amount remains unpaid after settlement, interest continues to accrue.
- (4) Nothing in this clause affects any other rights of the Seller under this contract or at law.

10. GENERAL

10.1 Seller's Agent

The Seller's Agent is appointed as the Seller's agent to introduce a Buyer.

10.2 Foreign Buyer Approval

The Buyer warrants that either:

- the Buyer's purchase of the Property is not a notifiable action; or
- (2) the Buyer has received a no objection notification, under the *Foreign Acquisitions and Takeovers Act* 1975 (Cth).

10.3 Duty

The Buyer must pay all duty on this contract.

10.4 Notices

- (1) Notices under this contract must be in writing.
- (2) Notices under this contract or notices required to be given by law may be given and received by the party's solicitor.
- (3) Notices under this contract or notices required to be given by law may be given by:
 - (a) delivering or posting to the other party or its solicitor; or
 - (b) sending it to the facsimile number of the other party or its solicitor stated in the Reference Schedule (or another facsimile number notified by the recipient to the sender); or
 - (c) sending it to the email address of the other party or its solicitor stated in the Reference Schedule (or another email address notified by the recipient to the sender).
- (4) Subject to clause 10.4(5), a notice given after this contract is entered into in accordance with clause 10.4(3) will be treated as given:
 - (a) 5 Business Days after posting;
 - (b) if sent by facsimile, at the time indicated on a clear transmission report; and
 - (c) if sent by email, at the time it is sent.

- (5) Notices given by facsimile, by personal delivery or by 10.10 Counterparts email between 5pm on a Business Day (the "first Business Day") and 9am on the next Business Day (the "second Business Day") will be treated as given or delivered at 9am on the second Business Day.
- (6) If two or more notices are treated as given at the same time under clause 10.4(5), they will be treated as given in the order in which they were sent or delivered.
- (7) Notices or other written communications by a party's solicitor (for example, varying the Inspection Date, Finance Date or Settlement Date) will be treated as given with that party's authority.
- For the purposes of clause 10.4(3)(c) and clause 12.2 the notice or information may be contained within an email, as an attachment to an email or located in an electronic repository accessible by the recipient by clicking a link in an email.

10.5 Business Davs

- (1) If anything is required to be done on a day that is not a Business Day, it must be done instead on the next
- If the Finance Date or Inspection Date fall on a day that is not a Business Day, then it falls on the next Business Day.

10.6 Rights After Settlement

Despite settlement and registration of the transfer, any term of this contract that can take effect after settlement or registration remains in force.

10.7 Further Acts

If requested by the other party, each party must, at its own expense, do everything reasonably necessary to give effect to this contract.

10.8 Severance

If any term or part of a term of this contract is or becomes legally ineffective, invalid or unenforceable in any jurisdiction it will be severed and the effectiveness, validity or enforceability of the remainder will not be affected.

10.9 Interpretation

(1) Plurals and Genders

Reference to:

- (a) the singular includes the plural and the plural includes the singular;
- (b) one gender includes each other gender;
- (c) a person includes a Body Corporate; and
- (d) a party includes the party's executors, administrators, successors and permitted assigns.

(2) Parties

- (a) If a party consists of more than one person, this contract binds them jointly and each of them individually.
- (b) A party that is a trustee is bound both personally and in its capacity as a trustee.

(3) Statutes and Regulations

Reference to statutes includes all statutes amending, consolidating or replacing them.

(4) Inconsistencies

If there is any inconsistency between any provision added to this contract and the printed provisions, the added provision prevails.

(5) **Headings**

Headings are for convenience only and do not form part of this contract or affect its interpretation.

- (1) This contract may be executed in two or more counterparts, all of which will together be deemed to constitute one and the same contract.
- A counterpart may be electronic and signed using an Electronic Signature.

11. ELECTRONIC SETTLEMENT

11.1 Application of Clause

- (1) Clause 11 applies if the Buyer, Seller and each Financial Institution involved in the transaction agree to an Electronic Settlement using the same ELNO System and overrides any other provision of this contract to the extent of any inconsistency.
- Acceptance of an invitation to an Electronic Workspace is taken to be an agreement for clause 11.1(1)
- (3) Clause 11 (except clause 11.5(3)) ceases to apply if either party gives notice under clause 11.5 that settlement will not be an Electronic Settlement.

11.2 Completion of Electronic Workspace

- (1) The parties must:
 - (a) ensure that the Electronic Workspace is completed and all Electronic Conveyancing Documents and the Financial Settlement Schedule are Digitally Signed prior to settlement;
 - (b) do everything else required in the Electronic Workspace or otherwise to enable settlement to occur on the Settlement Date.
- (2) If the parties cannot agree on a time for settlement, the time to be nominated in the Workspace is 4pm AEST.
- (3) If any part of the Purchase Price is to be paid to discharge an Outgoing:
 - (a) the Buyer may, by notice in writing to the Seller, require that the amount is paid to the Buyer's Solicitor's trust account and the Buyer is responsible for paying the amount to the relevant authority;
 - (b) for amounts to be paid to destination accounts other than the Buyer's Solicitor's trust account, the Seller must give the Buyer a copy of the current account for the Outgoing to enable the Buyer to verify the destination account details in the Financial Settlement Schedule.
- (4) If the Deposit is required to discharge any Encumbrance or pay an Outgoing at settlement:
 - (a) the Deposit Holder must, if directed by the Seller at least 2 Business Days prior to Settlement, pay the Deposit (and any interest accrued on investment of the Deposit) less commission as clear funds to the Seller's Solicitor;
 - the Buyer and the seller authorise the Deposit Holder to make the payment in clause 11(4)(a),
 - the Seller's Solicitor will hold the money as Deposit Holder under the Contract;
 - the Seller and Buyer authorise the Seller's Solicitor to pay the money as directed by the Seller in accordance with the Financial Settlement Schedule.

11.3 Electronic Settlement

- (1) Clauses 5.1(2) and 5.2 do not apply.
- Payment of the Balance Purchase Price electronically as directed by the Seller's Solicitor in the Financial Settlement Schedule satisfies the Buyer's obligation in clause 2.5(1).

- (3) The Seller and Buyer will be taken to have complied with
 - (a) clause 2.5(3)(c), (e) and (f); and
 - (b) clause 2.5(5)(d) and (e),

(as applicable) if at settlement the Financial Settlement Schedule specifies payment of the relevant amount to the account nominated by the Commissioner of Taxation.

- (4) The Seller will be taken to have complied with clause 5.3(1)(b), (c), (d), (e) and (f) if:
 - (a) in relation to documents which are suitable for Electronic Lodgement in the Land Registry at settlement, the documents are Digitally Signed within the Electronic Workspace; and
 - (b) in relation to any other document or thing, the Seller's Solicitor:
 - confirms in writing prior to settlement that it holds all relevant documents which are not suitable for Electronic Lodgement and all Keys (if requested under clause 5.3(1)(d)) in escrow on the terms contained in the QLS E-Conveyancing Guidelines; and
 - gives a written undertaking to send the documents and Keys (if applicable) to the Buyer or Buyer's Solicitor no later than the Business Day after settlement; and
 - (iii) if requested by the Buyer, provides copies of documents in the Seller's Solicitors possession.
- (5) A party is not in default to the extent it is prevented from complying with an obligation because the other party or the other party's Financial Institution has not done something in the Electronic Workspace.
- (6) Any rights under the contract or at law to terminate the contract may not be exercised during the time the Electronic Workspace is locked for Electronic Settlement.
- (7) Electronic Settlement is taken to occur when Financial Settlement is effected, whether or not Electronic Lodgement has occurred.

11.4 Computer System Unavailable

If settlement fails and cannot occur by 4pm AEST on the Settlement Date because a computer system operated by the Land Registry, Office of State Revenue, Reserve Bank, a Financial Institution or the relevant ELNO System is inoperative, neither party is in default and the Settlement Date is deemed to be the next Business Day. Time remains of the essence.

11.5 Withdrawal from Electronic Settlement

- Either party may elect not to proceed with an Electronic Settlement by giving written notice to the other party.
- (2) A notice under clause 11.5(1) may not be given later than 5 Business Days before the Settlement Date unless an Electronic Settlement cannot be effected because:
 - (a) the transaction is not a Qualifying Conveyancing Transaction: or
 - (b) a party's solicitor is unable to complete the transaction due to death, a loss of legal capacity or appointment of a receiver or administrator (or similar) to their legal practice or suspension of their access to the ELNO System; or
 - (c) the Buyer's or Seller's Financial Institution is unable to use the relevant ELNO System to effect Electronic Settlement.

- (3) If clause 11.5(2) applies:
 - (a) the party giving the notice must provide satisfactory evidence of the reason for the withdrawal; and
 - (b) the Settlement Date will be extended to the date 5 Business Days after the Settlement Date.

11.6 Costs

Each party must pay its own fees and charges of using the relevant ELNO System for Electronic Settlement.

11.7 Definitions for clause 11

In clause 11:

"Digitally Sign" and "Digital Signature" have the meaning in the ECNL.

"ECNL" means the Electronic Conveyancing National Law (Queensland).

"Electronic Conveyancing Documents" has the meaning in the Land Title Act 1994.

"Electronic Lodgement" means lodgement of a document in the Land Registry in accordance with the ECNL.

"Electronic Settlement" means settlement facilitated by an ELNO System.

"Electronic Workspace" means a shared electronic workspace within an ELNO System that allows the Buyer and Seller to effect Electronic Lodgement and Financial Settlement

"ELNO" has the meaning in the ECNL.

"ELNO System" means a system provided by the ELNO for facilitating Financial Settlement and Electronic Lodgement.

"Financial Settlement" means the exchange of value between Financial Institutions facilitated by an ELNO System in accordance with the Financial Settlement Schedule.

"Financial Settlement Schedule" means the electronic settlement schedule within the Electronic Workspace listing the source accounts and destination accounts.

"Qualifying Conveyancing Transaction" means a transaction that is not excluded for Electronic Settlement by the rules issued by the relevant ELNO, Office of State Revenue, Land Registry, or a Financial Institution involved in the transaction.

12. ELECTRONIC CONTRACT AND DISCLOSURE

12.1 Electronic Signing

If this contract is signed by any person using an Electronic Signature, the Buyer and the Seller:

- (a) agree to enter into this contract in electronic form;
 and
- (b) consent to either or both parties signing the contract using an Electronic Signature.

12.2 Pre-contract Disclosure

The Buyer consents to the Seller's use of electronic communication to give any notice or information required by law to be given to the Buyer and which was given before the Buyer signed this contract.



Annexure A

1. MODIFICATIONS TO TERMS OF CONTRACT

- 1.1 The Terms of Contract for Houses and Residential Land which form part of this Contract (the Standard Conditions) are amended as follows:
- (a) This Contract is not subject to finance and Clause 3 is deleted in its entirety;
- (b) This Contract is not subject to a Building, Pest or Pool Inspection and clause 4 is deleted in its entirety;
- (c) Clause 5.1 (1) is amended by replacing the words 4 pm AEST with 5 pm AEST;
- (d) The following clauses are deleted: clauses 7.4 (2) and (3); clauses 7.5 (2), (3) and (4); clause 7.6 (4) and clauses 7.7 (1) and (2);
- (e) Clause 7.4 (4) is amended to delete the words or clause 7.4 (2);
- (f) A new clause 7.5 (2) is inserted as follows
- The Buyer cannot not object to, delay settlement, terminate this Contract, deduct or retain any amount from the Purchase Price or make any requisition or claim for compensation about any error in the boundaries or area of the Land, any encroachment onto or from the Land or a mistake or omission in describing the Property or the Seller's Title to it (whether material or otherwise).
- 1.2 The Standard Conditions do not apply to this Contract to the extent of any inconsistency with these special conditions and the latter will prevail.
- 2. BUYER HAS SATISFIED ITSELF
- 2.1 The Buyer acknowledges and agrees that it has inspected or has had the opportunity to inspect the Property and has satisfied itself about:
- (a) The quality, condition and state of repair of the Property;
- (b) The purposes for which the Property can be used (including without limitation any restrictions on its use or development);
- (c) The fitness of the Property for the purposes the Buyer wants it;
- (d)The existence or otherwise of necessary approvals and licences from authorities relating to the Property including any failure to comply with any approvals or licences;
- (e) The carrying on of any notifiable activity on the land or the existence or otherwise of any hazardous contaminant on the land within the meaning of the Environmental Protection Act 1994:
- (f) The existence or otherwise of any legal and physical defects;
- (g) Accepts the Property on an as is where is basis, in its present condition and subject to any latent or patent defects.
- 2.2 The Buyer acknowledges and agrees that it has not relied on any representations, warranties or information provided or statements about the Property made by or on behalf of the Seller, the Seller's Agent or their respective employees or agents, other than as set out in this Contract.
- 3. Inventory
- 1 x Microwave
- 1x Fridge
- 1x Electric Jug
- 1x Toaster
- Assorted Cutlery + Crockery+ Glasses + Pots/Pans
- 1x Desk
- 1x Chair
- 2x cane chairs
- 2x small bedside tables
- 1x lamp
- 1x blanket box/coffee table
- 1x foldout sofa bed lounge
- 1x TV
- 1x outdoor setting (table + 4 x chairs)
- 1x double (split single) bed



Waiver/Shortening of Cooling off Period by a Proposed Buyer

Issued pursuant to section 167 of the Property Occupations Act 2014

NOTE FOR COMPLETION

Under the Property Occupations Act 2014 a buyer may:

- 1. waive their cooling off rights before entering into a relevant contract; or
- 2. shorten their cooling off rights **after** they enter into a relevant contract.

This form should be completed and signed either before or after the relevant contract is entered into depending on whether the buyer wishes to exercise option 1 or 2 above.

A "relevant contract" is defined under section 160 of the	Act.
I/We	
(insert name/s of proposed buyer/s)	
understand that I am/ we are entitled to a five business d 160 and 166 of the Property Occupations Act 2014 in rela	ay statutory cooling off period available under sections ation to a contract for the sale of the property located at:
Couran Cove Island Resort SUBURB: South Stradbroke ("the Property")	STATE: QLD POSTCODE: 4216
I/ we wish to:	
waive my/our statutory cooling off rights.	
shorten my/our statutory cooling off rights from 5 be	usiness days to days. (insert relevant number of days)
I/ We agree that I have/ we have voluntarily agreed to wa	ive or shorten my/our statutory cooling off rights.
SIGNATURES	
Buyer Name:	Buyer Name:
Signature:	Signature: SIGN HERI
Date:	Date:

INITIALS





Ма	ınaging Ageı	nt:						
AG	ENCY NAME:		•					
N/A	Α							
	OPERTY MANA	AGER:						
N/A								
ADI	DRESS:							
SU	BURB:				STATE:	POSTCODE:		
PH	ONE:	MOBILE:	FAX:	EMAIL:				
PC	OOL SAFET	TY FOR NON-SHARE	O POOLS					
Col	mplete the follo	owing questions if there is a	non-shared pool in the Lot		TO SELLER: Failure to			
01	Is there a r	non-shared pool on the I	_ot?	Safety Red	quirements is an offence	with substantial penalties		
	☐ Yes ✓ No	Clause 4.2 of this con		Exemption responsibl within 90 o	← WARNING TO BUYER: If there is no Compliance or Exemption Certificate at Settlement, the Buyer becomes responsible at its cost to obtain a Pool Safety Certificate within 90 days after settlement. The Buyer can also becon liable to pay any costs of rectification necessary to comply			
Q2	. If the answ Certificate t	er to Q1 is Yes, is there for the non-shared pool	a Compliance or Exemption at the time of contract?	on with the Po Certificate to substan	ool Safety Requirements . The Buyer commits an tial penalties if the Buye	to obtain a Pool Safety offence and can be liable r fails to comply with this		
	Yes	Clause 5.3(1)(f) applie	es .	requireme		is not completed then		
	☐ No	Clause 4.2 applies (ex other excluded sales)	cept for auction and some	clause 4.2	a pool on the Lot and Q2 applies.	e is not completed then		
Q3. If the answer to Q2 is No, has a Notice of No Pool Safety Certificate been given prior to contract? Yes			← Note: This of the Build	← Note: This is an obligation of the Seller under section 16 of the Building Regulation 2006.				
	No							
Po	ol Safety Ins	pector: N/A		← The Pool S ————— Building A	Safety Inspector must be ct 1975 and Building Reg	licensed under the gulation 2006.		
Pod	ol Safety Ins	pection Date:		← Clause 4.2 on a sale b	← Clause 4.2(2) applies except where this contract is formed on a sale by auction and some other excluded sales.			
ST	ATUTORY	WARRANTIES AND	CONTRACTUAL RIGH	rs				
The	Seller gives	s notice to the Buyer of th	ne following matters:					
(a)	Latent or Pa NONE	tent Defects in Common P	roperty or Body Corporate <i>i</i>		 WARNING TO SELLI Community Managemen Contract include warrant 	ies by the Seller about the		
(b)	Actual or Co	ntingent or Expected Liabi	lities of Body Corporate (se	ction 223(2)(c)(d))*	Body Corporate and the Scheme land. Bre of a warranty may result in a damages clair termination by the Buyer. Seller should corwhether to carry out an inspection of the Buyer. Corporate records to complete this section.			
	Circumstanc NONE	es in Relation to Affairs of	Body Corporate (section 22	23(3))*				
(d)	Exceptions t	o Warranties in clause 7.4	(3)*	_				
	Proposed Bo	ody Corporate Resolutions	(clause 8.4)*					

*Include in attachment if insufficient space



Excluded Fixtures	:					
Included Chattels:	Refer to	o Annexure A				
		- , , , , , , , , , , , , , , , , , , ,				
PRICE	1		_			
Deposit Holder:	Coba	alt Blue Resources P	ty Ltd			
Deposit Holder's T	rust Acco	ount: Cobalt Blue Res	ources Statutory True	- Account		
Bank:	West	<u>'</u>	ources statutory rius	Account	_	<u>-</u>
BSB: 034-061	7763	Account No: 42	4-099			
	Ф.					/ / / / / / / / / / / / / / / / / / /
Purchase Price:	\$				•	← Unless otherwise specified in this contract,
						the Purchase Price includes any GST payable
						on the supply of the Property to the Buyer.
						, ,
Donosit	ф 1004	of colo prico	Initial Denosit navah	le on the day the Buy	er sians this cont	ract unless another time is
Deposit:	\$ 10%	of sale price	specified below.		or orgino and cond	ade amous another ame is
			Signing of Expression			_
	\$		Balance Deposit (if a Within 3 Business da	any) payable on: ays from the date of c	ontract	
Default Interest Bar	to:	%	_ If no figure is inserte	d, the Contract Rate a	oplying at the Cont	ract Date published by the
Default Interest Rat	.e		` Queensland Law So	ciety inc will apply.		
FINANCE						
Finance Amount:	\$		← U	nless all of "Finance An	nount", "Financier"	and "Finance Date" are ance and clause 3 does
				ot apply.	is not subject to iii	ance and clause 3 does
Financier:			_			_
Finance Date:						
BUILDING AND	IOR PE	ST INSPECTION D	ΔΤΕ.			
				_ If "Inst	pection Date" is no	t completed, the contract is ion report and clause 4.1
Inspection Date:				not sui does r	bject to an inspecti oot apply.	on report and clause 4.1
MATTERS AFF	ECTING	PROPERTY				
Title Encumbrance						
Is the Property sold	l subject ⁱ	to any Encumbrances'	P ✓ No ☐ Yes	, listed below:	←WARNING T	O SELLER: You are required I Title Encumbrances which
					will remain at	ter settlement (for example,
					easements fo	n your title and statutory or sewerage and drainage
					Failure to dis	ot appear on a title search). close these may entitle the
					compensation	ninate the contract or to n. It is NOT sufficient to state
					"refer to title",	"search will reveal", or simila
Tenancies:				,		
TENANT'S NAME:			←If t Ot	he property is sold with herwise complete detai	vacant possessio Is from Residentia	n from settlement, insert 'Nil'. ' Tenancy Agreement.
NIL						
TERM AND OPTIONS:						
STARTING DATE OF T	ERM:	ENDING DATE OF TER		BONI	D:	
			\$	\$		_



BUYER NAME:				ABN	:
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
NAME:				ABN	:
ADDRESS:					
 SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
BUYER'S AG	ENT (If applicable)				
ABN:			LICENCE NO:		
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
BUYER'S SO	LICITOR		←or ar	ny other solicitor not	ified to the Seller
REF:	CONTACT:		-		
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
PROPERTY					
Lot: ADDRESS	Marine Apartment 1502 C	Couran Cove Island R	esort		
SUBURB	South Stradbroke			STATE: QLD	POSTCODE: 4216
Description: Lot	Lot 60	on: 🗸 BUP			
Scheme		Commu	nity Titles Scheme:		
Title Reference:	50220138				
Present Use:	Residential Unit				
	nt: Gold Coast City				







Contract for Residential Lots in a Community Titles Scheme

Twelfth Edition

This document has been approved by The Real Estate Institute of Queensland Limited and the Queensland Law Society Incorporated as being suitable for the sale and purchase of Residential Lots in a Community Titles Scheme in Queensland except for new residential property in which case the issue of GST liability must be dealt with by special condition.

The Seller and Buyer agree to sell and buy the Property under this contract.

	o. aa 2a, o.	agree to son and se	ay the Frequency th			
REFER	RENCE SC	HEDULE				
Contrac	t Date:					
SELLER	R'S AGENT					
NAME: Co	balt Blue Res	sources Pty Ltd				
ABN: 81 600 95	50 016			LICENCE NO: 4204690		
	PO Box 16					
SUBURB:	Toowong				STATE: QLD	POSTCODE: 4066
PHONE: 38707203		MOBILE: 0421878116	FAX:	EMAIL: rwilson@rwilson.co		
SELLEF NAME: Coralspri					ABN:	
ADDRESS:	2 Bega Road					
SUBURB:	Kingston				STATE: QLD	POSTCODE: 4114
PHONE: 34583408	3	MOBILE:	FAX:	EMAIL: brian.lincez@risings	suncg.com.au	
NAME:				-	ABN:	
ADDRESS:						
SUBURB:					STATE:	POSTCODE:
PHONE:		MOBILE:	FAX:	EMAIL:		
NAME:	'S SOLICIT			←or ar	ny other solicitor noti	fied to the Buyer
REF:	cCullough Lav	yers CONTACT:				
		Vicki Partanen				
ADDRESS:	2 Bayliss St					_
SUBURB:	Toowona				STATE: QLD	POSTCODE: 4066
PHONE: 07387082		MOBILE:	FAX:	EMAIL: admin@lockettmccu		

ADDITION	IAL BODY CORPORATE INFO	RMATION			
Interest Sch	edule Lot Entitlement of Lot:	Refer to "Body Corporate Disc	closure Statement"		
Aggregate Ir	nterest Schedule Lot Entitlement:	Refer to "Body Corporate Disc	losure Statement"		
Contribution	Schedule Lot Entitlement of Lot:	Refer to "Body Corporate Disc	losure Statement"		
Aggregate C	contribution Schedule Lot Entitlement:	Refer to "Body Corporate Disc	losure Statement"		
INSURANG	CE POLICIES				
Insurer:	See Body Corporate Disclosure Sta	atement	Policy No:		
Building:					
Public Liability	<i>y</i> :				
Other:					
					_
B. If the ans at the time C. If the ans Note: This is a ELECTRIC The Seller g General Pur (select v in in The Seller g	wer to A is Yes, is there a Compliance of contract? wer to B is No, has a Notice of No is an obligation of the Seller under section is AL SAFETY SWITCH AND SM ives notice to the Buyer that an Approse Socket Outlets is: whichever is applicable) stalled in the residence of installed in the residence ives notice to the Buyer that a Complex control of the	nce or Exemption Certificate for Pool Safety Certificate been giv 16 of the Building Regulation 2006. OKE ALARM This section moroved Safety Switch for the	r the shared pool	se or misleading n, the Seller may ould seek expert mpleting this sec 's Agent to comp stall a Compliant	and ction olete Smoke
✓ in	whichever is applicable) stalled in the residence ot installed in the residence		Services Act 1990.	r the fine and Er	neigency
NEIGHBOU	JRHOOD DISPUTES (DIVIDING	FENCES AND TREES) AC	CT 2011		
of the Neight that the Land (select w	ives notice to the Buyer in accordar bourhood Disputes (Dividing Fence d: whichever is applicable) not affected by any application to, ueensland Civil and Administrative a tree on the Lot or	os and Trees) Act 2011 or an order made by, the	← WARNING: Failure to co Neighbourhood Disputes Trees Act) 2011 by givin application to the Buyer to Buyer signing the con to terminate the contract	s (Dividing Fence g a copy of an oi (where applicabl tract will entitle tl	es and rder or e) prior ne Buyer
is is	affected by an application to, or an lation to a tree on the Lot, a copy o the Buyer prior to the Buyer signin	f which has been given			

GST WI	THHOLDING OBLIGATIONS	
Is the Buy	ver registered for GST and acquiring the Lot for a creditable purpose?	←WARNING: the Buyer warrants in
(sele	ct whichever is applicable)	clause 2.5(6) that this information is true and correct.
	Yes	
\checkmark	No	
the L	e: An example of an acquisition for a creditable purpose would be the purchase of and by a building contractor, who is registered for GST, for the purposes of building use on the Land and selling it in the ordinary course of its business.]	
	r gives notice to the Buyer in accordance with section 14-255(1)(a) of the ng Law that:	← WARNING: All sellers of residential premises or potential residential land are required to complete this notice.
(sele	ct whichever is applicable)	Section 14-250 of the Withholding La applies to the sale of 'new residential
\checkmark	the Buyer <i>is not</i> required to make a payment under section 14-250 of the Withholding Law in relation to the supply of the Property	premises' or 'potential residential land (subject to some exceptions) and requires an amount to be withheld fro
	the Buyer <i>is</i> required to make a payment under section 14-250 of the Withholding Law in relation to the supply of the Property. Under section 14-255(1) of the Withholding Law, the Seller is required to give further details prior to settlement.	the Purchase Price and paid to the ATO. The Seller should seek legal advice if unsure about completing this section.
The	REIQ Terms of Contract for Residential Lots in a Community Titles Sch Twelfth Edition Contain the Terms of this Contract.	neme (Pages 8-17)
SPECIA	L CONDITIONS	
	event that the buyer or seller signs the contract and transmits it electronically via enallid and binding obligation by both parties with the force and effect as if it was an	
SETTLE	MENT	
Settlemen	Date: 30 Days from the Contract date	← or the next Business Day if that is not a Business Day in the Place for Settlement.
Place for S	Settlement: Brisbane	← If Brisbane is inserted, this is a reference to Brisbane CBD.

SIGNATURES

The contract may be subject to a 5 business day statutory cooling-off period. A termination penalty of 0.25% of the purchase price applies if the Buyer terminates the contract during the statutory cooling-off period. It is recommended the Buyer obtain an independent property valuation and independent legal advice about the contract and his or her cooling-off rights, before signing.

Buyer; ×	Witness; *	HER
Buyer:	Witness:	
By placing my signature above, I warrant that I am the Buyer named in the Reference Schedule or authorised by the Buyer to sign.	(Note: No witness is required Electronic Signature)	d if the Buyer signs using an
Seller: ×	Witnes₅; <mark>≭</mark>	SIGN HER
Seller:	Witness:	
By placing my signature above, I warrant that I am the Seller named in the Reference Schedule or authorised by the Seller to sign.	(Note: No witness is required Electronic Signature)	if the Seller signs using an
Deposit Holder:	← Who acknowledges having received to hold that amount and any Balance Deposit Holder for the parties as prov	Deposit when received as

TERMS OF CONTRACT FOR RESIDENTIAL LOTS IN A COMMUNITY TITLES SCHEME

1. DEFINITIONS

- 1.1 In this contract:
 - terms in **bold** in the Reference Schedule and the Disclosure Statement have the meanings shown opposite them unless the context requires otherwise; and
 - (2) (a) "Approved Safety Switch" means a residual current device as defined in the Electrical Safety Regulation 2013;
 - (b) "ATO" means the Australian Taxation Office;
 - (c) "ATO Clearance Certificate" means a certificate issued under section 14-220(1) of the Withholding Law which is current on the date it is given to the Buyer;
 - (d) "Balance Purchase Price" means the Purchase Price, less the Deposit paid by the Buyer, adjusted under clause 2.6;
 - (e) "Bank" means an authorised deposit-taking institution within the meaning of the Banking Act 1959 (Cth).
 - (f) "Body Corporate" means the body corporate of the Scheme;
 - (g) "Body Corporate Debt" has the meaning in the Regulation Module but excludes the Body Corporate Levies for the period which includes the Settlement Date;
 - (h) "Body Corporate Levies" means regular periodic contributions levied on the owner of the Lot (including, if applicable, levied under an exclusive use by-law) excluding any Special Contribution:
 - "Bond" means a bond under the Residential Tenancies and Rooming Accommodation Act 2008;
 - "Building" means any building that forms part of the Lot or in which the Lot is situated;
 - (k) "Building Inspector" means a person licensed to carry out completed residential building inspections under the Queensland Building and Construction Commission Regulations 2003:
 - (I) "Business Day" means a day other than:
 - (i) a Saturday or Sunday
 - (ii) a public holiday in the Place for Settlement; and
 - (iii) a day in the period 27 to 31 December (inclusive).
 - (m) "CGT Withholding Amount" means the amount determined under section 14-200(3)(a) of the Withholding Law or, if a copy is provided to the Buyer prior to settlement, a lesser amount specified in a variation notice under section 14-235;
 - (n) "Compliance or Exemption Certificate" means:
 - (i) a Pool Safety Certificate; or
 - (ii) a building certificate that may be used instead of a Pool Safety Certificate under section 246AN(2) of the Building Act 1975;
 - (iii) an exemption from compliance on the grounds of impracticality under section 245B of the *Building Act 1975*;
 - (o) "Compliant Smoke Alarm" means a smoke alarm complying with the requirements for smoke alarms in domestic dwellings under the Fire and Emergency Services Act 1990;
 - (p) "Contract Date" or "Date of Contract" means the date inserted in the Reference Schedule;
 - (q) "Court" includes any tribunal established under statute;

- (r) "Disclosure Statement" means the statement under section 206 (Existing Lot) or section 213 (Proposed Lot) of the Body Corporate and Community Management Act 1997;
- (s) "Electronic Signature" means an electronic method of signing that identifies the person and indicates their intention to sign the contract;
- (t) "Encumbrances" includes:
 - (i) unregistered encumbrances
 - (ii) statutory encumbrances; and
 - (iii) Security Interests.
- (u) "Essential Term" includes, in the case of breach by:
 - (i) the Buyer: clauses 2.2, 2.5(1), 2.5(5),5.1 and 6.1; and
 - (ii) the Seller: clauses 2.5(5), 5.1, 5.3(1)(a) (d), 5.3(1)(e)(ii) & (iii), 5.3(1)(f), 5.5 and 6.1; but nothing in this definition precludes a Court from finding other terms to be essential;
- (v) "Exclusive Use Areas" means parts of the common property for the Scheme allocated to the Lot under an exclusive use by-law;
- (w) "Financial Institution" means a Bank, Building Society or Credit Union;
- (X) "General Purpose Socket Outlet" means an electrical socket outlet as defined in the Electrical Safety Regulation 2013;
- (y) "GST" means the goods and services tax under the GST Act;
- (z) "GST Act" means A New Tax System (Goods and Services Tax) Act 1999 (Cth) and includes other GST related legislation;
- (aa) "GST Withholding Amount" means the amount (if any) determined under section 14-250 of the Withholding Law required to be paid to the Commissioner of Taxation;
- (bb) "Improvements" means fixed structures in the Lot (such as stoves, hot water systems, fixed carpets, curtains, blinds and their fittings, clothes lines, fixed satellite dishes and television antennae, in-ground plants) but does not include the Reserved Items;
- (cc) "Keys" means keys, codes or devices in the Seller's possession or control for all locks or security systems on the Property or necessary to access the Property;
- (dd) "Land" means the scheme land for the Scheme;
- (ee) "Notice of No Pool Safety Certificate" means the Form 36 under the *Building Regulation 2006* to the effect that there is no Pool Safety Certificate issued for the Land and/or the Lot;
- (ff) "Notice of nonconformity" means a Form 26 under the Building Regulation 2006 advising how the pool does not comply with the relevant pool safety standard;
- (gg) "Outgoings" means:
 - rates or charges on the Lot by any competent authority (for example, council rates, water rates, fire service levies) but excludes land tax; and
 - (ii) Body Corporate Levies.
- (hh) "Pest Inspector" means a person licensed to undertake termite inspections on completed buildings under the Queensland Building and Construction Commission Regulations 2003;
- (ii) "Pool Safety Certificate" has the meaning in section 231C(a) of the Building Act 1975;

- (jj) "Pool Safety Requirements" means the requirements for pool safety contained in the Building Act 1975 and Building Regulation 2006;
- (kk) "Pool Safety Inspection Date" means the Pool Safety Inspection Date inserted in the Reference Schedule. If no date is inserted in the Reference Schedule, the Pool Safety Inspection Date is taken to be the earlier of the following:
 - the Inspection Date for the Building and/or Pest Inspection; or
 - (ii) 2 Business Days before the Settlement Date;
- (II) "Pool Safety Inspector" means a person authorised to give a Pool Safety Certificate;
- (mm) "PPSR" means the Personal Property Securities Register established under Personal Property Securities Act 2009 (Cth);
 - (nn) "Property" means:
 - (i) the Lot:
 - (ii) the right to any Exclusive Use Areas;
 - (iii) the Improvements;
 - (iv) the Included Chattels;
- (oo) "Regulation Module" means the regulation module for the Scheme;
- (pp) "Rent" means any periodic amount, including outgoings, payable under the Tenancies;
- (qq) "Reserved Items" means the Excluded Fixtures and all Chattels in the Lot and Exclusive Use Areas other than the Included Chattels;
- (rr) "Scheme" means the community titles scheme containing the Lot;
- (ss) "Security Interests" means all security interests registered on the PPSR over Included Chattels and Improvements;
- (tt) "Special Contribution" means an amount levied by the Body Corporate under the Regulation Module for a liability for which no provision or inadequate provision has been made in the budget of the Body Corporate.
- (uu) "Transfer Documents" means:
 - the form of transfer under the Land Title Act 1994 required to transfer title in the Lot to the Buyer; and
 - (ii) any other document to be signed by the Seller necessary for stamping or registering the transfer
- (vv) "Transport Infrastructure" has the meaning defined in the Transport Infrastructure Act 1994; and
- (ww) "Withholding Law" means Schedule 1 to the Taxation Administration Act 1953 (Cth).
- **1.2** Words and phrases defined in the *Body Corporate and Community Management Act 1997* have the same meaning in this contract unless the context indicates otherwise.

2. PURCHASE PRICE

2.1 **GST**

- (1) Unless otherwise specified in this contract, the Purchase Price includes any GST payable on the supply of the Property to the Buyer.
- (2) If a party is required to make any other payment or reimbursement under this contract, that payment or reimbursement will be reduced by the amount of any input tax credits to which the other party (or the representative member for a GST group of which it is a member) is entitled.

2.2 Deposit

EF003 12/19

(1) The Buyer must pay the Deposit to the Deposit Holder at the times shown in the Reference Schedule. The Deposit Holder will hold the Deposit until a party becomes entitled to it.

- (2) The Buyer will be in default if it:
 - (a) does not pay the Deposit when required;
 - (b) pays the Deposit by a post-dated cheque; or
 - (c) pays the Deposit by cheque which is dishonoured on presentation.
- (3) The Seller may recover from the Buyer as a liquidated debt any part of the Deposit which is not paid when required.

2.3 Investment of Deposit

If:

- (1) the Deposit Holder is instructed by either the Seller or the Buyer; and
- (2) it is lawful to do so;

the Deposit Holder must:

- (3) invest as much of the Deposit as has been paid with any Financial Institution in an interest-bearing account in the names of the parties; and
- (4) provide the parties' tax file numbers to the Financial Institution (if they have been supplied).

2.4 Entitlement to Deposit and Interest

- (1) The party entitled to receive the Deposit is:
 - (a) if this contract settles, the Seller;
 - (b) if this contract is terminated without default by the Buyer, the Buyer;
 - (c) if this contract is terminated owing to the Buyer's default, the Seller.
- (2) The interest on the Deposit must be paid to the person who is entitled to the Deposit.
- (3) If this contract is terminated, the Buyer has no further claim once it receives the Deposit and interest unless the termination is due to the Seller's default or breach of warranty.
- (4) The Deposit is invested at the risk of the party who is ultimately entitled to it.

2.5 Payment of Balance Purchase Price

- On the Settlement Date, the Buyer must pay the Balance Purchase Price by Bank cheque as the Seller or the Seller's Solicitor directs.
- (2) Despite any other provision of this contract, reference to a "Bank cheque" in clause 2.5:
 - includes a cheque drawn by a Building Society or Credit Union on itself;
 - (b) does not include a cheque drawn by a Building Society or Credit Union on a Bank;
 and the Seller is not obliged to accept a cheque

and the Seller is not obliged to accept a cheque referred to in clause 2.5(2)(b) on the Settlement Date.

- (3) If both the following apply:
 - (a) the sale is not an excluded transaction under section 14-215 of the Withholding Law; and
 - (b) the Seller has not given the Buyer on or before settlement for each person comprising the Seller either:
 - (i) an ATO Clearance Certificate; or
 - (ii) a variation notice under section 14-235 of the Withholding Law which remains current at the Settlement Date varying the CGT Withholding Amount to nil,

then:

(c) for clause 2.5(1), the Seller irrevocably directs the Buyer to draw a bank cheque for the CGT Withholding Amount in favour of the Commissioner of Taxation or, if the Buyer's Solicitor requests, the Buyer's Solicitor's Trust Account;

- (d) the Buyer must lodge a Foreign Resident Capital Gains Withholding Purchaser Notification Form with the ATO for each person comprising the Buyer and give copies to the Seller with the payment reference numbers (PRN) on or before settlement;
- (e) the Seller must return the bank cheque in paragraph (c) to the Buyer's Solicitor (or if there is no Buyer's Solicitor, the Buyer) at settlement; and
- (f) the Buyer must pay the CGT Withholding Amount to the ATO in accordance with section 14-200 of the Withholding Law and give the Seller evidence that it has done so within 2 Business Days of settlement occurring.
- (4) For clause 2.5(3) and section 14-215 of the Withholding Law, the market value of the CGT asset is taken to be the Purchase Price less any GST included in the Purchase Price for which the Buyer is entitled to an input tax credit unless:
 - (a) the Property includes items in addition to the Lot and Improvements; and
 - (b) no later than 2 Business Days prior to the Settlement Date, the Seller gives the Buyer a valuation of the Lot and Improvements prepared by a registered valuer,

in which case the market value of the Lot and Improvements will be as stated in the valuation.

- (5) If the Buyer is required to pay the GST Withholding Amount to the Commissioner of Taxation at settlement pursuant to section 14-250 of the Withholding Law:
 - (a) the Seller must give the Buyer a notice in accordance with section 14-255(1) of the Withholding Law;
 - (b) prior to settlement the Buyer must lodge with the ATO:
 - a GST Property Settlement Withholding Notification form ("Form 1"); and
 - (ii) a GST Property Settlement Date Confirmation form ("Form 2");
 - (c) on or before settlement, the Buyer must give the Seller copies of:
 - (i) the Form 1;
 - (ii) confirmation from the ATO that the Form 1 has been lodged specifying the Buyer's lodgement reference number and payment reference number;
 - (iii) confirmation from the ATO that the Form 2 has been lodged; and
 - (iv) a completed ATO payment slip for the Withholding Amount.
 - (d) the Seller irrevocably directs the Buyer to draw a bank cheque for the GST Withholding Amount in favour of the Commissioner of Taxation and deliver it to the Seller at settlement; and
 - (e) the Seller must pay the GST Withholding Amount to the ATO in compliance with section 14-250 of the Withholding Law promptly after settlement.
- (6) The Buyer warrants that the statements made by the Buyer in the Reference Schedule under GST Withholding Obligations are true and correct.

2.6 Adjustments to Balance Purchase Price

(1) The Seller is liable for Outgoings and is entitled to Rent up to and including the Settlement Date. The Buyer is liable for Outgoings and is entitled to Rent after the Settlement Date.

- (2) Subject to clauses 2.6(3), 2.6(5) and 2.6(17), Outgoings for periods including the Settlement Date must be adjusted:
 - (a) for those paid, on the amount paid;
 - (b) for those assessed but unpaid, on the amount payable (excluding any discount); and
 - (c) for those not assessed:
 - on the amount the relevant authority or the Body Corporate advises will be assessed (excluding any discount); or
 - (ii) if no advice on the assessment to be made is available, on the amount of the latest assessment (excluding any discount).
- (3) If there is no separate assessment of rates for the Lot at the Settlement Date and the Local Government informs the Buyer that it will not apportion rates between the Buyer and the Seller, then:
 - (a) the amount of rates to be adjusted is that proportion of the assessment equal to the ratio of the interest schedule lot entitlement of the Lot to the aggregate interest schedule lot entitlement of the Scheme; and
 - (b) if an assessment of rates includes charges imposed on a "per lot" basis, then the portion of those charges to be adjusted is the amount assessed divided by the number of lots in that assessment.
- (4) The Seller is liable for land tax assessed on the Lot for the financial year current at the Settlement Date. If land tax is unpaid at the Settlement Date and the Office of State Revenue advises that it will issue a final clearance for the Lot on payment of a specified amount, then the Buyer may deduct the specified amount from the Balance Purchase Price at settlement and must pay it promptly to the Office of State Revenue.
- (5) Any Outgoings assessable on the amount of water used must be adjusted on the charges that would be assessed on the total water usage for the assessment period, determined by assuming that the actual rate of usage shown by the meter reading made before settlement continues throughout the assessment period. The Buyer must obtain and pay for the meter reading.
- (6) If any Outgoings are assessed but unpaid at the Settlement Date, then the Buyer may deduct the amount payable from the Balance Purchase Price at settlement and pay it promptly to the relevant authority or the Body Corporate, as appropriate. If an amount is deducted under this clause, the relevant Outgoing will be treated as paid at the Settlement Date for the purposes of clause 2.6(2).
- (7) Arrears of Rent for any rental period ending on or before the Settlement Date belong to the Seller and are not adjusted at settlement.
- (8) Unpaid Rent for the rental period including both the Settlement Date and the following day ("Current Period") is not adjusted until it is paid.
- (9) Rent already paid for the Current Period or beyond must be adjusted at settlement.
- (10) If Rent payments are reassessed after the Settlement Date for periods including the Settlement Date, any additional Rent payment from a Tenant or refund due to a Tenant must be apportioned under clauses 2.6(7), 2.6(8) and 2.6(9) and 2.6(10).
- (11) Payments under clause 2.6(10) must be made within 14 days after notification by one party to the other but only after any additional payment from a Tenant has been received.

(12) The Seller is liable for:

- (a) any Special Contribution for which a levy notice has been issued on or before the Contract Date;
 and
- (b) any other Body Corporate Debt (including any penalty or recovery cost resulting from nonpayment of a Body Corporate Debt) owing in respect of the Lot at settlement.

The Buyer is liable for any Special Contribution levied after the Contract Date.

- (13) If an amount payable by the Seller under clause 2.6(12) is unpaid at the Settlement Date, the Buyer may deduct the specified amount from the Balance Purchase Price at settlement and must pay it promptly to the Body Corporate.
- (14) For the purposes of clause 2.6(12), an amount payable under an exclusive use by-law will be treated as levied on the date it is due.
- (15) The cost of Bank cheques payable at settlement:
 - (a) to the Seller or its mortgagee are the responsibility of the Buyer; and
 - (b) to parties other than the Seller or its mortgagee are the responsibility of the Seller.
- (16) The Seller is not entitled to require payment of the Balance Purchase Price by means other than Bank cheque without the consent of the Buyer.
- (17) Upon written request by the Buyer, the Seller will, prior to settlement, give the Buyer a written statement, supported by reasonable evidence, of —
 - (a) all Outgoings and all Rent for the Property to the extent they are not capable of discovery by search or enquiry at any office of public record or pursuant to the provisions of any statute; and
 - (b) any other information which the Buyer may reasonably require for the purpose of calculating or apportioning any Outgoings or Rent under this clause 2.6.

If the Seller becomes aware of a change to the information provided the Seller will as soon as practicably provide the updated information to the Buyer.

3. FINANCE

- 3.1 This contract is conditional on the Buyer obtaining approval of a loan for the Finance Amount from the Financier by the Finance Date on terms satisfactory to the Buyer. The Buyer must take all reasonable steps to obtain approval.
- 3.2 The Buyer must give notice to the Seller that:
 - approval has not been obtained by the Finance Date and the Buyer terminates this contract; or
 - (2) the finance condition has been either satisfied or waived by the Buyer.
- 3.3 The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 3.2 by 5pm on the Finance Date. This is the Seller's only remedy for the Buyer's failure to give notice.
- 3.4 The Seller's right under clause 3.3 is subject to the Buyer's continuing right to give written notice to the Seller of satisfaction, termination or waiver pursuant to clause 3.2

4. BUILDING AND PEST INSPECTION REPORTS AND POOL SAFETY

4.1 Building and Pest Inspection

(1) This contract is conditional on the Buyer obtaining a written building report from a Building Inspector and a written pest report from a Pest Inspector (which may be a single report) on the Property by the Inspection Date on terms satisfactory to the Buyer. The Buyer must take all reasonable steps to obtain the reports (subject to the right of the Buyer to elect to obtain only one of the reports).

- (2) The Buyer must give notice to the Seller that:
 - (a) a satisfactory Inspector's report under clause 4.1(1) has not been obtained by the Inspection Date and the Buyer terminates this contract. The Buyer must act reasonably; or
 - (b) clause 4.1(1) has been either satisfied or waived by the Buyer.
- (3) If the Buyer terminates this contract and the Seller asks the Buyer for a copy of the building and pest reports, the Buyer must give a copy of each report to the Seller without delay.
- (4) The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 4.1(2) by 5pm on the Inspection Date. This is the Seller's only remedy for the Buyer's failure to give notice.
- (5) The Seller's right under clause 4.1(4) is subject to the Buyer's continuing right to give written notice to the Seller of satisfaction, termination or waiver pursuant to clause 4.1(2).

4.2 Pool Safety for non-shared pool on Lot

- (1) This clause 4.2 applies if:
 - (a) there is a pool on the Lot and the answer to Q2 of the Reference Schedule is No or Q2 is not completed (for a non-shared pool on the Lot);
 and
 - (b) this contract is not a contract of a type referred to in section 160(1)(b) of the Property Occupations Act 2014.
- (2) This contract is conditional upon:
 - (a) the issue of a Pool Safety Certificate; or
 - (b) a Pool Safety Inspector issuing a Notice of Nonconformity stating the works required before a Pool Safety Certificate can be issued;
 - by the Pool Safety Inspection Date.
- (3) The Buyer is responsible for arranging an inspection by a Pool Safety Inspector at the Buyer's cost. The Seller authorises:
 - (a) the Buyer to arrange the inspection; and
 - (b) the Pool Safety Inspector to advise the Buyer of the results of the inspection and to give the Buyer a copy of any notice issued.
- (4) If a Pool Safety Certificate has not been issued by the Pool Safety Inspection Date, the Buyer may give notice to the Seller that the Buyer:
 - (a) terminates this contract; or
 - (b) waives the benefit of this clause 4.2

The Buyer must act reasonably.

- (5) The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 4.2(4) by 5pm on the Pool Safety Inspection Date.
- (6) The Seller's right under clause 4.2(5) is subject to the Buyer's continuing right to give written notice to the Seller of termination or waiver pursuant to clause 4.2(4).
- (7) The right of a party to terminate under this clause 4.2, ceases upon receipt by that party of a copy of a current Pool Safety Certificate.
- (8) If the Buyer terminates this contract under clause 4.2(4)(a), and the Seller has not obtained a copy of the Notice of Nonconformity issued by the Pool Safety Inspector, the Seller may request a copy and the Buyer must provide this to the Seller without delay.

5. SETTLEMENT

5.1 Time and Date

 Settlement must occur between 9am and 4pm AEST on the Settlement Date.

INITIALS (Note: initials not required if signed with Electronic Signature)

(2) If the parties do not agree on where settlement is to occur, it must take place in the Place for Settlement at the office of a Solicitor or Financial Institution nominated by the Seller, or, if the Seller does not make a nomination, at the Land Registry Office in or nearest to the Place for Settlement.

5.2 Transfer Documents

- (1) The Transfer Documents must be prepared by the Buyer's Solicitor and delivered to the Seller a reasonable time before the Settlement Date.
- (2) If the Buyer pays the Seller's reasonable expenses, it may require the Seller to produce the Transfer Documents at the Office of State Revenue nearest the Place for Settlement for stamping before settlement.

5.3 Documents and Keys at Settlement

- In exchange for payment of the Balance Purchase Price, the Seller must deliver to the Buyer at settlement:
 - (a) any instrument of title for the Lot required to register the transfer to the Buyer; and
 - (b) unstamped Transfer Documents capable of immediate registration after stamping; and
 - (c) any instrument necessary to release any Encumbrance over the Property in compliance with the Seller's obligation in clause 7.2; and
 - (d) if requested by the Buyer not less than 2 clear Business Days before the Settlement Date, the Keys; and
 - (e) if there are Tenancies:
 - (i) the Seller's copy of any Tenancy agreements;
 - (ii) a notice to each tenant advising of the sale in the form required by law; and
 - (iii) any notice required by law to transfer to the Buyer the Seller's interest in any Bond; and
 - (f) if the answer to Q2 in the Reference Schedule is Yes, a copy of a current Compliance or Exemption Certificate, if not already provided to the Buyer.
- (2) If the Keys are not delivered at settlement under clause 5.3(1)(d), the Seller must deliver the Keys to the Buyer. The Seller may discharge its obligation under this provision by authorising the Seller's Agent to release the Keys to the Buyer.

5.4 Assignment of Covenants and Warranties

At settlement, the Seller assigns to the Buyer the benefit of all:

- (1) covenants by the tenants under the Tenancies;
- (2) guarantees and Bonds (subject to the requirements of the Residential Tenancies and Rooming Accommodation Act 2008) supporting the Tenancies; and
- (3) manufacturers' warranties regarding the Included Chattels;
- (4) builders' warranties on the Improvements; to the extent that they are assignable and the Buyer accepts the assignment. However, the right to recover arrears of Rent is not assigned to the Buyer and section 117 of the *Property Law Act 1974* does not apply.

5.5 Possession of Property and Title to Included Chattels

On the Settlement Date, in exchange for the Balance Purchase Price, the Seller must give the Buyer vacant possession of the Lot and Exclusive Use Areas except for the Tenancies. Title to the Included Chattels passes at settlement.

5.6 Reservations

 The Seller must remove the Reserved Items from the Property before settlement.

- (2) The Seller must repair at its expense any damage done to the Property in removing the Reserved` Items. If the Seller fails to do so, the Buyer may repair that damage.
- (3) Any Reserved Items not removed before settlement will be considered abandoned and the Buyer may, without limiting its other rights, complete this contract and appropriate those Reserved Items or dispose of them in any way.
- (4) The Seller indemnifies the Buyer against any damages and expenses resulting from the Buyer's actions under clauses 5.6(2) or 5.6(3).

6. TIME

6.1 Time of the Essence

Time is of the essence of this contract, except regarding any agreement between the parties on a time of day for settlement.

6.2 Suspension of Time

- (1) This clause 6.2 applies if a party is unable to perform a settlement obligation solely as a consequence of a Delay Event but does not apply where the inability is attributable to: —
 - (a) damage to, destruction of or diminution in value of the Property or other property of the Seller or Buyer; or
 - (b) termination or variation of any agreement between a party and another person whether relating to the provision of finance, the release of an Encumbrance, the sale or purchase of another property or otherwise.
- (2) Time for the performance of the parties' Settlement Obligations is suspended and ceases to be of the essence of the contract and the parties are deemed not to be in breach of their Settlement Obligations.
- (3) An Affected Party must take reasonable steps to minimise the effect of the Delay Event on its ability to perform its Settlement Obligations.
- (4) When an Affected Party is no longer prevented from performing its Settlement Obligations due to the Delay Event, the Affected Party must give the other party a notice of that fact, promptly.
- (5) When the Suspension Period ends, whether notice under clause 6.2(4) has been given or not, either party may give the other party a Notice to Settle.
- (6) A Notice to Settle must be in writing and state:
 - (a) that the Suspension Period has ended; and
 - (b) a date, being not less than 5 nor more than 10 Business Days after the date the Notice to Settle is given, which shall become the Settlement Date:
 - (c) that time is of the essence.
- (7) When Notice to Settle is given, time is again of the essence of the contract.
- (8) In this clause 6.2:
 - (a) "Affected Party" means a party referred to in clause 6.2(1);
 - (b) "Delay Event" means:
 - a tsunami, flood, cyclone, earthquake, bushfire or other act of nature;
 - (ii) riot, civil commotion, war, invasion or a terrorist act;
 - (iii) an imminent threat of an event in paragraphs (i) or (ii); or
 - (iv) compliance with any lawful direction or order by a Government Agency;

- (c) "Government Agency" means the government of the Commonwealth of Australia or an Australian State, Territory or local government and includes their authorities, agencies, government owned corporations and authorised officers, courts and tribunals;
 - (d) "Settlement Obligations" means, in the case of the Buyer, its obligations under clauses 2.5(1) and 5.1(1) and, in the case of the Seller, its obligations under clauses 5.1(1), 5.3(1)(a) – (e) and 5.5:
- (e) "Suspension Period" means the period during which the Affected Party (or if both the Buyer and Seller are Affected Parties, either of them) remains unable to perform a settlement obligation solely as a consequence of a Delay Event.

7. MATTERS AFFECTING THE PROPERTY

7.1 Title

The Lot is sold subject to:

- the Body Corporate and Community Management Act 1997 and the by-Laws of the Body Corporate;
- any reservations or conditions on the title or the original Deed of Grant.

7.2 Encumbrances

The Property is sold free of all Encumbrances other than the Title Encumbrances, Tenancies, statutory easements implied by part 6A of the *Land Title Act 1994* and interests registered on the common property for the Scheme.

7.3 Requisitions

The Buyer may not deliver any requisitions or enquiries on title.

7.4 Seller's Warranties

- (1) The Seller warrants that, except as disclosed in this contract, at settlement:
 - it will be the registered owner of an estate in fee simple in the Lot and will own the Improvements and Included Chattels;
 - it will be capable of completing this contract (unless the Seller dies or becomes mentally incapable after the Contract Date); and
 - (c) there will be no unsatisfied judgment, order (except for an order referred to in clause 7.6(1)(b)) or writ affecting the Property.
- (2) The Seller warrants that, except as disclosed in this contract, at the Contract Date and at settlement there are no current or threatened claims, notices or proceedings that may lead to a judgment, order or writ affecting the Property.
- (3) The Seller warrants that, except as disclosed in this contract, at the Contract Date:
 - (a) there is no unregistered lease, easement or other right capable of registration and which is required to be registered to give indefeasibility affecting the common property or Body Corporate assets;
 - (b) there is no proposal to record a new community management statement for the Scheme and it has not received a notice of a meeting of the Body Corporate to be held after the Contract Date or notice of any proposed resolution or a decision of the Body Corporate to consent to the recording of a new community management statement for the Scheme;
 - (c) all Body Corporate consents to improvements made to common property and which benefit the Lot, or the registered owner of the Lot, are in force; and

- (d) the Additional Body Corporate Information is correct (if completed).
- (4) If the Seller breaches a warranty in clause 7.4(1) or clause 7.4(2), the Buyer may terminate this contract by notice to the Seller.
- (5) If:
 - (a) the Seller breaches a warranty in clause 7.4(3);
 - (b) the Additional Body Corporate Information is not completed;

and, as a result, the Buyer is materially prejudiced, the Buyer may terminate this contract by notice to the Seller given within 14 days after the Contract Date but may not claim damages or compensation.

- (6) Clauses 7.4(4) and 7.4(5) do not restrict any statutory rights the Buyer may have which cannot be excluded by this contract.
- (7) (a) The Seller warrants that, except as disclosed in this contract or a notice given by the Seller to the Buyer under the Environmental Protection Act 1994 ("EPA"), at the Contract Date:
 - there is no outstanding obligation on the Seller to give notice to the administering authority under EPA of notifiable activity being conducted on the Land; and
 - (ii) the Seller is not aware of any facts or circumstances that may lead to the Land being classified as contaminated land within the meaning of EPA.
 - (b) If the Seller breaches a warranty in clause 7.4(7), the Buyer may:
 - terminate this contract by notice in writing to the Seller given no later than 2 Business Days before the Settlement Date; or
 - (ii) complete this contract and claim compensation, but only if the Buyer claims it in writing before the Settlement Date.
- (8) The Seller does not warrant that the Present Use is lawful.

7.5 Survey and Mistake

- (1) The Buyer may survey the Lot.
- (2) If there is:
 - (a) an error in the boundaries or area of the Lot;
 - (b) an encroachment by structures onto or from the Lot; or
 - (c) a mistake or omission in describing the Lot or the Seller's title to it;

which is:

- (d) immaterial; or
- (e) material, but the Buyer elects to complete this

the Buyer's only remedy against the Seller is for compensation, but only if claimed by the Buyer in writing on or before settlement.

- (3) The Buyer may not delay settlement or withhold any part of the Balance Purchase Price because of any compensation claim under clause 7.5(2).
- (4) If there is a material error, encroachment or mistake, the Buyer may terminate this contract before settlement.

7.6 Requirements of Authorities

- (1) Subject to clause 7.6(5), any valid notice or order by any competent authority or Court requiring work to be done or money spent in relation to the Property ("Work or Expenditure") must be fully complied with:
 - (a) if issued before the Contract Date, by the Seller before the Settlement Date;

- (b) if issued on or after the Contract Date, by the Buyer.
- (2) If any Work or Expenditure that is the Seller's responsibility under clause 7.6(1)(a) is not done before the Settlement Date, the Buyer is entitled to claim the reasonable cost of work done by the Buyer in accordance with the notice or order referred to in clause 7.6(1) from the Seller after settlement as a debt.
- (3) Any Work or Expenditure that is the Buyer's responsibility under clause 7.6(1)(b), which is required to be done before the Settlement Date, must be done by the Seller unless the Buyer directs the Seller not to and indemnifies the Seller against any liability for not carrying out the work. If the Seller does the work, or spends the money, the reasonable cost of that Work or Expenditure must be added to the Balance Purchase Price.
- (4) The Buyer may terminate this contract by notice to the Seller if there is an outstanding notice at the Contract Date under section 246AG of the *Building Act 1975* that affects the Property. The Buyer may terminate this contract by notice to the Seller if there is an outstanding notice at the Contract Date under sections 247 or 248 of the *Building Act 1975* or sections 167 or 168 of the *Planning Act 2016* that affects the Property or Land.
- (5) Clause 7.6(1) does not apply to orders disclosed under section 83 of the Neighbourhood Disputes (Dividing Fences and Trees) Act 2011.

7.7 Property Adversely Affected

- (1) If at the Contract Date:
 - (a) the Present Use is not lawful under the relevant town planning scheme;
 - (b) the Land is affected by a proposal of any competent authority to alter the dimensions of any Transport Infrastructure or locate Transport Infrastructure on the Land;
 - (c) access or any service to the Land passes unlawfully through other land;
 - (d) any competent authority has issued a current notice to treat, or notice of intention to resume, regarding any part of the Land;
 - (e) there is an outstanding condition of a development approval attaching to the Land under section 73 of the Planning Act 2016 or section 96 of the Economic Development Queensland Act 2012 which, if complied with, would constitute a material mistake or omission in the Seller's title under clause 7.5(2)(c);
 - (f) the Property is affected by the *Queensland Heritage Act 1992* or is included in the World Heritage List;
 - (g) the Property is declared acquisition land under the *Queensland Reconstruction Authority Act* 2011; or
 - (h) there is a charge against the Lot under section 104 of the Foreign Acquisitions and Takeovers Act 1975 (Cth),
 - and that has not been disclosed in this contract, the Buyer may terminate this contract by notice to the Seller given on or before settlement.
- (2) If no notice is given under clause 7.7(1), the Buyer will be treated as having accepted the Property subject to all of the matters referred to in that clause.
- (3) The Seller authorises the Buyer to:
 - inspect records held by any authority, including Security Interests on the PPSR, relating to the Property or the Land; and
 - (b) apply for a certificate of currency of the Body Corporate's insurance from any insurer.

7.8 Dividing Fences

Notwithstanding any provision in the *Neighbourhood Disputes (Dividing Fences and Trees) Act 2011*, the Seller need not contribute to the cost of building any dividing fence between the Lot and any adjoining land owned by it. The Buyer waives any right to claim contribution from the Seller.

8. RIGHTS AND OBLIGATIONS UNTIL SETTLEMENT

8.1 Risk

The Property is at the Buyer's risk from 5pm on the first Business Day after the Contract Date.

8.2 Access

After reasonable notice to the Seller, the Buyer and its consultants may enter the Property:

- (1) once to read any meter;
- (2) for inspections under clause 4;
- (3) once to inspect the Property before settlement; and
- (4) once to value the Property before settlement.

8.3 Seller's Obligations After Contract Date

- (1) The Seller must use the Property reasonably until settlement. The Seller must not do anything regarding the Property or Tenancies that may significantly alter them or result in later expense for the Buyer.
- (2) The Seller must promptly upon receiving any notice, proceeding or order that affects the Property or requires work on the Property, give a copy to the Buyer.
- (3) Without limiting clause 8.3(1), the Seller must not without the prior written consent of the Buyer, give any notice or seek or consent to any order that affects the Property or make any agreement affecting the Property that binds the Buyer to perform.

8.4 Body Corporate Meetings

- (1) The Seller must promptly give the Buyer a copy of:
 - (a) any notice it receives of a proposed meeting of the Body Corporate to be held after the Contract Date; and
 - (b) resolutions passed at that meeting and prior to settlement.
- (2) The Buyer may terminate this contract by notice in writing to the Seller given before settlement if it is materially prejudiced by:
 - (a) any resolution of the Body Corporate passed after the Contract Date, other than a resolution, details of which are disclosed to the Buyer in this contract; or
 - (b) where the Scheme is a subsidiary scheme, any resolution of a Body Corporate of a higher scheme.
- (3) In clause 8.4(2) a resolution includes a decision of the Body Corporate Committee to consent to recording a new community management statement.
- (4) If the Buyer is not given a copy of the resolutions before settlement, it may sue the Seller for damages.

8.5 Information Regarding the Property

Upon written request of the Buyer but in any event before settlement, the Seller must give the Buyer:

- copies of all documents relating to any unregistered interests in the Property;
- full details of the Tenancies to allow the Buyer to properly manage the Property after settlement;
- (3) sufficient details (including the date of birth of each Seller who is an individual) to enable the Buyer to undertake a search of the PPSR;

8.6 Possession Before Settlement

If possession is given before settlement:

- the Buyer must maintain the Property in substantially its condition at the date of possession, fair wear and tear excepted;
- (2) entry into possession is under a licence personal to the Buyer revocable at any time and does not:
 - (a) create a relationship of landlord and tenant; or
 - (b) waive the Buyer's rights under this contract;
- (3) the Buyer must insure the Property to the Seller's satisfaction; and
- (4) the Buyer indemnifies the Seller against any expense or damages incurred by the Seller as a result of the Buyer's possession of the Property.

8.7 Seller's Obligations After Contract Date

- the Seller must promptly upon receiving any notice, proceeding or order that affects the Property or requires work on the Property, give a copy to the Buyer.
- (2) After the Contract Date, the Seller must not without the prior written consent of the Buyer, give any notice, seek or consent to any order or make an agreement that affects the Property.

9. PARTIES' DEFAULT

9.1 Seller and Buyer May Affirm or Terminate

Without limiting any other right or remedy of the parties including those under this contract, or any right at law or in equity, if the Seller or Buyer, as the case may be, fails to comply with an Essential Term, or makes a fundamental breach of an intermediate term, the Seller (in the case of the Buyer's default) or the Buyer (in the case of the Seller's default) may affirm or terminate this contract.

9.2 If Seller Affirms

If the Seller affirms this contract under clause 9.1, it may sue the Buyer for:

- (1) damages;
- (2) specific performance; or
- (3) damages and specific performance.

9.3 If Buyer Affirms

If the Buyer affirms this contract under clause 9.1, it may sue the Seller for:

- (1) damages:
- (2) specific performance; or
- (3) damages and specific performance.

9.4 If Seller Terminates

If the Seller terminates this contract under clause 9.1, it may do all or any of the following:

- (1) resume possession of the Property;
- (2) forfeit the Deposit and any interest earned;
- (3) sue the Buyer for damages;
- (4) resell the Property.

9.5 If Buyer Terminates

If the Buyer terminates this contract under clause 9.1, it may do all or any of the following:

- (1) recover the Deposit and any interest earned;
- (2) sue the Seller for damages.

9.6 Seller's Resale

- (1) If the Seller terminates this contract and resells the Property, the Seller may recover from the Buyer as liquidated damages:
 - (a) any deficiency in price on a resale; and
 - (b) its expenses connected with any repossession, any failed attempt to resell, and the resale,

provided the resale settles within 2 years of termination of this contract.

(2) Any profit on a resale belongs to the Seller.

9.7 Seller's Damages

The Seller may claim damages for any loss it suffers as a result of the Buyer's default, including its legal costs on an indemnity basis and the cost of any Work or Expenditure under clause 7.6(3).

9.8 Buyer's Damages

The Buyer may claim damages for any loss it suffers as a result of the Seller's default, including its legal costs on an indemnity basis.

9.9 Interest on Late Payments

- (1) The Buyer must pay interest at the Default Rate:
 - (a) on any amount payable under this contract which is not paid when due; and
 - (b) on any judgement for money payable under this contract.
- (2) Interest continues to accrue:
 - (a) under clause 9.9(1)(a), from the date it is due until paid; and
 - (b) under clause 9.9(1)(b), from the date of judgement until paid.
- (3) Any amount payable under clause 9.9(1)(a) in respect of a period prior to settlement must be paid by the Buyer at settlement. If this contract is terminated or if any amount remains unpaid after settlement, interest continues to accrue.
- (4) Nothing in this clause affects any other rights of the Seller under this contract or at law.

10. GENERAL

10.1 Seller's Agent

The Seller's Agent is appointed as the Seller's agent to introduce a Buyer.

10.2 Foreign Buyer Approval

The Buyer warrants that either:

- the Buyer's purchase of the Property is not a notifiable action; or
- (2) the Buyer has received a no objection notification, under the Foreign Acquisitions and Takeovers Act 1975 (Cth).

10.3 Duty

The Buyer must pay all duty on this contract.

10.4 Notices

- (1) Notices under this contract must be in writing.
- (2) Notices under this contract or notices required to be given by law may be given and received by the party's solicitor.
- (3) Notices under this contract or notices required to be given by law may be given by:
 - (a) delivering or posting to the other party or its solicitor; or
 - (b) sending it to the facsimile number of the other party or its solicitor stated in the Reference Schedule (or another facsimile number notified by the recipient to the sender); or
 - (c) sending it to the email address of the other party or its solicitor stated in the Reference Schedule (or another email address notified by the recipient to the sender).
- (4) Subject to clause 10.4(5), a notice given after this contract is entered into in accordance with clause 10.4(3) will be treated as given:
 - (a) 5 Business Days after posting;
 - (b) if sent by facsimile, at the time indicated on a clear transmission report; and
 - (c) if sent by email, at the time it is sent.



- (5) Notices given by facsimile, by personal delivery or by 10.10 Counterparts email between 5pm on a Business Day (the "first Business Day") and 9am on the next Business Day (the "second Business Day") will be treated as given or delivered at 9am on the second Business Day.
- (6) If two or more notices are treated as given at the same time under clause 10.4(5), they will be treated as given in the order in which they were sent or delivered.
- (7) Notices or other written communications by a party's solicitor (for example, varying the Inspection Date, Finance Date or Settlement Date) will be treated as given with that party's authority.
- (8) For the purposes of clause 10.4(3)(c) and clause 12.2 the notice or information may be contained within an email, as an attachment to an email or located in an electronic repository accessible by the recipient by clicking a link in an email.

10.5 Business Days

- (1) If anything is required to be done on a day that is not a Business Day, it must be done instead on the next Business Day.
- If the Finance Date or Inspection Date fall on a day that is not a Business Day, then it falls on the next Business Day.

10.6 Rights After Settlement

Despite settlement and registration of the transfer, any term of this contract that can take effect after settlement or registration remains in force.

10.7 Further Acts

If requested by the other party, each party must, at its own expense, do everything reasonably necessary to give effect to this contract.

10.8 Severance

If any term or part of a term of this contract is or becomes legally ineffective, invalid or unenforceable in any jurisdiction it will be severed and the effectiveness, validity or enforceability of the remainder will not be affected.

10.9 Interpretation

(1) Plurals and Genders

Reference to:

- (a) the singular includes the plural and the plural includes the singular;
- (b) one gender includes each other gender;
- (c) a person includes a Body Corporate; and
- (d) a party includes the party's executors, administrators, successors and permitted assigns.

(2) Parties

- (a) If a party consists of more than one person, this contract binds them jointly and each of them individually.
- (b) A party that is a trustee is bound both personally and in its capacity as a trustee.

(3) Statutes and Regulations

Reference to statutes includes all statutes amending, consolidating or replacing them.

(4) Inconsistencies

If there is any inconsistency between any provision added to this contract and the printed provisions, the added provision prevails.

(5) Headings

Headings are for convenience only and do not form part of this contract or affect its interpretation.

- (1) This contract may be executed in two or more counterparts, all of which will together be deemed to constitute one and the same contract.
- (2) A counterpart may be electronic and signed using an Electronic Signature.

11. ELECTRONIC SETTLEMENT

11.1 Application of Clause

- (1) Clause 11 applies if the Buyer, Seller and each Financial Institution involved in the transaction agree to an Electronic Settlement using the same ELNO System and overrides any other provision of this contract to the extent of any inconsistency.
- Acceptance of an invitation to an Electronic Workspace is taken to be an agreement for clause
- (3) Clause 11 (except clause 11.5(3)) ceases to apply if either party gives notice under clause 11.5 that settlement will not be an Electronic Settlement.

11.2 Completion of Electronic Workspace

- (1) The parties must:
 - (a) ensure that the Electronic Workspace is completed and all Electronic Conveyancing Documents and the Financial Settlement Schedule are Digitally Signed prior to settlement;
 - do everything else required in the Electronic Workspace or otherwise to enable settlement to occur on the Settlement Date.
- (2) If the parties cannot agree on a time for settlement, the time to be nominated in the Workspace is 4pm AEST.
- If any part of the Purchase Price is to be paid to discharge an Outgoing:
 - (a) the Buyer may, by notice in writing to the Seller. require that the amount is paid to the Buyer's Solicitor's trust account and the Buyer is responsible for paying the amount to the relevant authority;
 - (b) for amounts to be paid to destination accounts other than the Buyer's Solicitor's trust account, the Seller must give the Buyer a copy of the current account for the Outgoing to enable the Buyer to verify the destination account details in the Financial Settlement Schedule.
- (4) If the Deposit is required to discharge any Encumbrance or pay an Outgoing at settlement:
 - (a) the Deposit Holder must, if directed by the Seller at least 2 Business Days prior to Settlement, pay the Deposit (and any interest accrued on investment of the Deposit) less commission as clear funds to the Seller's Solicitor;
 - the Buyer and the seller authorise the Deposit Holder to make the payment in clause 11(4)(a);
 - the Seller's Solicitor will hold the money as Deposit Holder under the Contract;
 - the Seller and Buyer authorise the Seller's Solicitor to pay the money as directed by the Seller in accordance with the Financial Settlement Schedule.

11.3 Electronic Settlement

- (1) Clauses 5.1(2) and 5.2 do not apply.
- Payment of the Balance Purchase Price electronically as directed by the Seller's Solicitor in the Financial Settlement Schedule satisfies the Buyer's obligation in clause 2.5(1).

- (3) The Seller and Buyer will be taken to have complied with:
 - (a) clause 2.5(3)(c), (e) and (f); and
 - (b) clause 2.5(5)(d) and (e),

(as applicable) if at settlement the Financial Settlement Schedule specifies payment of the relevant amount to the account nominated by the Commissioner of Taxation.

- (4) The Seller will be taken to have complied with clause 5.3(1)(b), (c), (d), (e) and (f) if:
 - (a) in relation to documents which are suitable for Electronic Lodgement in the Land Registry at settlement, the documents are Digitally Signed within the Electronic Workspace; and
 - (b) in relation to any other document or thing, the Seller's Solicitor:
 - confirms in writing prior to settlement that it holds all relevant documents which are not suitable for Electronic Lodgement and all Keys (if requested under clause 5.3(1)(d)) in escrow on the terms contained in the QLS E-Conveyancing Guidelines; and
 - (ii) gives a written undertaking to send the documents and Keys (if applicable) to the Buyer or Buyer's Solicitor no later than the Business Day after settlement; and
 - (iii) if requested by the Buyer, provides copies of documents in the Seller's Solicitors possession.
- (5) A party is not in default to the extent it is prevented from complying with an obligation because the other party or the other party's Financial Institution has not done something in the Electronic Workspace.
- (6) Any rights under the contract or at law to terminate the contract may not be exercised during the time the Electronic Workspace is locked for Electronic Settlement.
- (7) Electronic Settlement is taken to occur when Financial Settlement is effected, whether or not Electronic Lodgement has occurred.

11.4 Computer System Unavailable

If settlement fails and cannot occur by 4pm AEST on the Settlement Date because a computer system operated by the Land Registry, Office of State Revenue, Reserve Bank, a Financial Institution or the relevant ELNO System is inoperative, neither party is in default and the Settlement Date is deemed to be the next Business Day. Time remains of the essence.

11.5 Withdrawal from Electronic Settlement

- Either party may elect not to proceed with an Electronic Settlement by giving written notice to the other party.
- (2) A notice under clause 11.5(1) may not be given later than 5 Business Days before the Settlement Date unless an Electronic Settlement cannot be effected because:
 - (a) the transaction is not a Qualifying Conveyancing Transaction; or
 - a party's solicitor is unable to complete the transaction due to death, a loss of legal capacity or appointment of a receiver or administrator (or similar) to their legal practice or suspension of their access to the ELNO System; or
 - (c) the Buyer's or Seller's Financial Institution is unable to use the relevant ELNO System to effect Electronic Settlement.

- (3) If clause 11.5(2) applies:
 - (a) the party giving the notice must provide satisfactory evidence of the reason for the withdrawal; and
 - (b) the Settlement Date will be extended to the date 5 Business Days after the Settlement Date.

11.6 Costs

Each party must pay its own fees and charges of using the relevant ELNO System for Electronic Settlement.

11.7 Definitions for clause 11

In clause 11:

"Digitally Sign" and "Digital Signature" have the meaning in the ECNL.

"ECNL" means the Electronic Conveyancing National Law (Queensland).

"Electronic Conveyancing Documents" has the meaning in the Land Title Act 1994.

"Electronic Lodgement" means lodgement of a document in the Land Registry in accordance with the ECNL.

"Electronic Settlement" means settlement facilitated by an ELNO System.

"Electronic Workspace" means a shared electronic workspace within an ELNO System that allows the Buyer and Seller to effect Electronic Lodgement and Financial Settlement.

"ELNO" has the meaning in the ECNL.

"ELNO System" means a system provided by the ELNO for facilitating Financial Settlement and Electronic Lodgement.

"Financial Settlement" means the exchange of value between Financial Institutions facilitated by an ELNO System in accordance with the Financial Settlement Schedule.

"Financial Settlement Schedule" means the electronic settlement schedule within the Electronic Workspace listing the source accounts and destination accounts.

"Qualifying Conveyancing Transaction" means a transaction that is not excluded for Electronic Settlement by the rules issued by the relevant ELNO, Office of State Revenue, Land Registry, or a Financial Institution involved in the transaction.

12. ELECTRONIC CONTRACT AND DISCLOSURE

12.1 Electronic Signing

If this contract is signed by any person using an Electronic Signature, the Buyer and the Seller:

- (a) agree to enter into this contract in electronic form; and
- (b) consent to either or both parties signing the contract using an Electronic Signature.

12.2 Pre-contract Disclosure

The Buyer consents to the Seller's use of electronic communication to give any notice or information required by law to be given to the Buyer and which was given before the Buyer signed this contract.



Annexure A

1. MODIFICATIONS TO TERMS OF CONTRACT

- 1.1 The Terms of Contract for Houses and Residential Land which form part of this Contract (the Standard Conditions) are amended as follows:
- (a) This Contract is not subject to finance and Clause 3 is deleted in its entirety;
- (b) This Contract is not subject to a Building, Pest or Pool Inspection and clause 4 is deleted in its entirety; (c) Clause 5.1 (1) is amended by replacing the words 4 pm AEST with 5 pm AEST;
- (d) The following clauses are deleted: clauses 7.4 (2) and (3); clauses 7.5 (2), (3) and (4); clause 7.6 (4) and clauses 7.7 (1) and (2);
- (e) Clause 7.4 (4) is amended to delete the words or clause 7.4 (2);

(f) A new clause 7.5 (2) is inserted as follows

The Buyer cannot not object to, delay settlement, terminate this Contract, deduct or retain any amount from the Purchase Price or make any requisition or claim for compensation about any error in the boundaries or area of the Land, any encroachment onto or from the Land or a mistake or omission in describing the Property or the Seller's Title to it (whether material or otherwise).

- 1.2 The Standard Conditions do not apply to this Contract to the extent of any inconsistency with these special conditions and the latter will prevail.
- 2. BUYER HAS SATISFIED ITSELF
- 2.1 The Buyer acknowledges and agrees that it has inspected or has had the opportunity to inspect the Property and has satisfied itself about:
- (a) The quality, condition and state of repair of the Property:
- (b) The purposes for which the Property can be used (including without limitation any restrictions on its use or development);
- (c) The fitness of the Property for the purposes the Buyer wants it;
- (d)The existence or otherwise of necessary approvals and licences from authorities relating to the Property including any failure to comply with any approvals or licences;
- (e) The carrying on of any notifiable activity on the land or the existence or otherwise of any hazardous contaminant on the land within the meaning of the Environmental Protection Act 1994;
- (f) The existence or otherwise of any legal and physical defects;
- (g) Accepts the Property on an as is where is basis, in its present condition and subject to any latent or patent defects.
- 2.2 The Buyer acknowledges and agrees that it has not relied on any representations, warranties or information provided or statements about the Property made by or on behalf of the Seller, the Seller's Agent or their respective employees or agents, other than as set out in this Contract.
- 3. Inventory
- 1 x Microwave
- 1x Fridge
- 1x Electric Jua
- 1x Toaster
- Assorted Cutlery + Crockery+ Glasses + Pots/Pans
- 1x Desk
- 1x Chair
- 2x cane chairs
- 2x small bedside tables
- 1x lamp
- 1x blanket box/coffee table
- 1x foldout sofa bed lounge
- 1x TV
- 1x outdoor setting (table + 4 x chairs)
- 1x double (split single) bed





Waiver/Shortening of Cooling off Period by a Proposed Buyer

Issued pursuant to section 167 of the Property Occupations Act 2014

NOTE FOR COMPLETION

Under the Property Occupations Act 2014 a buyer may:

- 1. waive their cooling off rights before entering into a relevant contract; or
- 2. shorten their cooling off rights after they enter into a relevant contract.

This form should be completed and signed either before or after the relevant contract is entered into depending on whether the buyer wishes to exercise option 1 or 2 above.

A "relevant contract" is defined under section 160 of the A	ct.
I/We	
(insert name/s of proposed buyer/s)	
understand that I am/ we are entitled to a five business da 160 and 166 of the Property Occupations Act 2014 in relat	y statutory cooling off period available under sections tion to a contract for the sale of the property located at:
ADDRESS: Marine Apartment 1503	
SUBURB: South Stradbroke	STATE: QLD POSTCODE: 4216
("the Property")	
I/ we wish to:	
waive my/our statutory cooling off rights.	
shorten my/our statutory cooling off rights from 5 bu	siness days to days. (insert relevant number of days)
I/ We agree that I have/ we have voluntarily agreed to wait	ve or shorten my/our statutory cooling off rights.
SIGNATURES	
Buyer Name:	Buyer Name:
Signature: *	Signature: SIGN HERE
Date:	Date:

INITIALS









Contract for Residential Lots in a Community Titles Scheme

Twelfth Edition

This document has been approved by The Real Estate Institute of Queensland Limited and the Queensland Law Society Incorporated as being suitable for the sale and purchase of Residential Lots in a Community Titles Scheme in Queensland except for new residential property in which case the issue of GST liability must be dealt with by special condition.

The Seller and Ruyer agree to sell and buy the Property under this contract

THE Sen	iei aliu buye	agree to sell and bi	uy tile Property u	nuci uns conuaci.		
REFER	RENCE SC	HEDULE				
Contrac	t Date:					
SELLE	R'S AGENT					
NAME: C	obalt Blue Re	sources Pty Ltd				
ABN: 81 600 9	50 016			LICENCE NO: 4204690		
	i: PO Box 16			4204090		
ADDITESS	PO B0X 16					
SUBURB;	Toowong				STATE: QLD	POSTCODE: 4066
PHONE:		MOBILE:	FAX:	EMAIL:		
3870720	3	0421878116		rwilson@rwilson.c	om.au	
SELLER	R					
NAME:					ABN:	
Coralspr	it Pty Ltd					
ADDRESS	: 2 Bega Road					
	· Z Deya Ruat					
SUBURB:	Kingston				STATE: QLD	POSTCODE: 4114
PHONE: 34583408	8	MOBILE:	FAX:	EMAIL: brian.lincez@rising	gsuncg.com.au	
NAME:					ABN:	
ADDRESS:	:				<u> </u>	
SUBURB:					STATE:	POSTCODE:
PHONE:		MOBILE:	FAX:	EMAIL:		_ :
SELLER NAME:	R'S SOLICIT	OR		←ora	any other solicitor noti	fied to the Buyer
	cCullough Lav					
REF:		CONTACT: Vicki Partanen				
ADDRESS:	2 Bayliss St					
SUBURB:	Toowong				STATE: QLD	POSTCODE: 4066
PHONE:	244	MOBILE:	FAX:	EMAIL:		
07387082	244			admin@lockettmcc	cuilougn.com.a <u>u</u>	

INITIAL 000015022655

BUYER NAME:				Δ	BN:
NAME.					J11.
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
NAME:				Α	BN:
ADDRESS:					
 SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
BUYER'S AGE	ENT (If applicable)				
ABN:			LICENCE NO:		
ADDRESS:					
SUBURB:	<u>. </u>			STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
BUYER'S SOL	LICITOR		←or	any other solicitor I	notified to the Seller
REF:	CONTACT:				
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
PROPERTY Lot: ADDRESS:	Marine Apartment 1503 (Couran Cove Island R	esort		
SUBURB:	South Stradbroke			STATE: QLI	POSTCODE: 4216
Description: Lot:	Lot 61	on: 🗹 BUP	GTP SP 106825		
Title Reference:	50220138			•	
Present Use:	Residential Unit				
Local Governmen	t: Gold Coast City				

INITIALS (Note: initials not required if signed with Electronic Signature)



Excluded Fixtures:						
Included Chattels: F	Refer to Anr	nexure A				
PRICE Deposit Holder:	Cobalt Bl	ue Resources Pt	y Ltd			
Deposit Holder's Tru	st Account:	Cobalt Blue Reso	ources Statutory Tr	ust Account		
Bank:	Westpac					
BSB: 034-061		Account No: 424	1-099			
Purchase Price:	\$	_				← Unless otherwise specified in this contract, the Purchase Price includes any GST payable on the supply of the Property to the Buyer.
Deposit:	10% of sa	le price	Initial Deposit pay specified below. Signing of Expres	-	e Buyer signs this c	ontract unless another time is
9	\$		Balance Deposit	(if any) payable on s days from the dat		
Default Interest Rate:	:	%				Contract Date published by the
FINANCE			•	, , , ,		
Finance Amount:	\$			Unless all of "Final completed, this conot apply.	nce Amount", "Finand Intract is not subject t	cier" and "Finance Date" are o finance and clause 3 does
Financier:						
Finance Date:						
BUILDING AND/C	OR PEST I	NSPECTION D	ATE:			
Inspection Date:					If "Inspection Date" is not subject to an insp does not apply.	s not completed, the contract is pection report and clause 4.1
MATTERS AFFEO		OPERTY			acconstappiy.	
Is the Property sold s		y Encumbrances?	✓ No □ Y	es, listed below:	to disclos will remai easemen easemen which ma Failure to Buyer to	IG TO SELLER: You are required to all Title Encumbrances which in after settlement (for example, its on your title and statutory its for sewerage and drainage by not appear on a title search), it disclose these may entitle the terminate the contract or to ation. It is NOT sufficient to state itle", "search will reveal", or similar
Tenancies: TENANT'S NAME: NIL			÷	If the property is so Otherwise complete	old with vacant posses e details from Reside	ssion from settlement, insert 'Nil'. ntial Tenancy Agreement.
TERM AND OPTIONS:						
STARTING DATE OF TE	RM: ENI	DING DATE OF TERM	M: RENT:		BOND:	

1NITIAL 000015022655

Managing Ager	nt:						
AGENCY NAME:							
N/A							
PROPERTY MANA	GER:						
N/A							
ADDRESS:							
SUBURB:				STATE:	POSTCODE:		
PHONE;	MOBILE:	FAX:	EMAIL:				
POOL SAFET	Y FOR NON-SHARE	D POOLS					
Complete the folio	owing questions if there is a	non-shared pool in the Lot	←WARNING	TO SELLER: Failure t	o comply with the Pool		
•	on-shared pool on the	•			e with substantial penalties.		
Yes			Exemption responsible	G TO BUYER: If there is Certificate at Settlemen e at its cost to obtain a l	nt, the Buyer becomes Pool Safety Certificate		
√ No	Clause 4.2 of this cor	tract does not apply	liable to pa	v any costs of rectificat	ne Buyer can also become ion necessary to comply		
Q2. If the answ Certificate f	er to Q1 is Yes, is there or the non-shared pool	a Compliance or Exempt at the time of contract?	Certificate.	with the Pool Safety Requirements to obtain a Pool Safe Certificate. The Buyer commits an offence and can be li to substantial penalties if the Buyer fails to comply with			
Yes	Clause 5.3(1)(f) appli	es	•		2 is not completed then		
☐ No	Clause 4.2 applies (e other excluded sales)	xcept for auction and som	e clause 4.2		z is not completed then		
Q3. If the answ Certificate b	er to Q2 is No, has a No been given prior to contr	ntice of No Pool Safety act?	← Note: This of the Build	← Note: This is an obligation of the Seller under section 16 of the Building Regulation 2006.			
Yes No							
Pool Safety Ins	pector: N/A			Safety Inspector must be ct 1975 and Building Re			
Pool Safety Ins	pection Date:		← Clause 4.2 on a sale b	← Clause 4.2(2) applies except where this contract is formed on a sale by auction and some other excluded sales.			
STATUTORY	WARRANTIES AND	CONTRACTUAL RIGH	ITS				
The Seller gives	notice to the Buyer of t	he following matters:					
•	•	Property or Body Corporate	Assets (section 223/a)/b))*			
NONE	tent Delects in Common i	Toperty of Body Corporate		 WARNING TO SELL Community Management Contract include warran 	ties by the Seller about the		
(b) Actual or Co	ntingent or Expected Liab	ilities of Body Corporate (se	ection 223(2)(c)(d))*	 Body Corporate and the Scheme land. Breach of a warranty may result in a damages claim or termination by the Buyer. Seller should consider whether to carry out an inspection of the Body Corporate records to complete this section. 			
(c) Circumstanc	es in Relation to Affairs o	f Body Corporate (section 2	23(3))*				
(d) Exceptions to	o Warranties in clause 7.4	1(3)*					
(e) Proposed Bo	ody Corporate Resolution	s (clause 8.4)*					

INITIALS (Note: initials not required if signed with Electronic Signature)

*Include in attachment if insufficient space



ADDITION	NAL BODY CORPORATE INFO	RMATION			
Interest Sch	nedule Lot Entitlement of Lot:	Refer to "Body Corporate Disc	losure Statement"		
Aggregate I	nterest Schedule Lot Entitlement:	Refer to "Body Corporate Disc	losure Statement"		
Contribution	Schedule Lot Entitlement of Lot:	Refer to "Body Corporate Disc	losure Statement"		
Aggregate C	Contribution Schedule Lot Entitlement:	Refer to "Body Corporate Disc	losure Statement"		
INSURAN	CE POLICIES				
Insurer:	See Body Corporate Disclosure Sta	atement	Policy No:		
Building:					
Public Liabilit	y:				
Other:					
					
POOL SAI	FETY FOR SHARED POOLS				
	e the following questions if there is a sha	ared pool on the Land.			
	a shared pool on the Land or on adj		with the Land?	✓ Yes	□No
B. If the ans	swer to A is Yes, is there a Complia e of contract?			✓ Yes	☐ No
C. If the ans	swer to B is No, has a Notice of No	Pool Safety Certificate been giv	en prior to contract?	Yes	☐ No
Note: This is	an obligation of the Seller under section	16 of the Building Regulation 2006.			
ELECTRIC	CAL SAFETY SWITCH AND SM	IOKE ALARM This section m	oust be completed unless the l	Lot is vacant	
The Seller of General Pul	gives notice to the Buyer that an Ap rpose Socket Outlets is: whichever is applicable) nstalled in the residence not installed in the residence		← WARNING: By giving fal information in this section a penalty. The Seller sho qualified advice about co and not rely on the Seller this section.	lse or misleading n, the Seller may ould seek expert ompleting this sed	incur and ction
-	gives notice to the Buyer that a Com whichever is applicable)	npliant Smoke Alarm(s) is/are:	← WARNING: Failure to ins Alarm is an offence unde Services Act 1990.		
	nstalled in the residence				
	ot installed in the residence				
	URHOOD DISPUTES (DIVIDING	-	CT 2011		
	jives notice to the Buyer in accorda abourhood Disputes (Dividing Fence d:		← WARNING: Failure to co Neighbourhood Disputes Trees Act) 2011 by givin	s (Dividing Fence ng a copy of an o	es and rder or
(select	whichever is applicable)		application to the Buyer to Buyer signing the con	ntract will entitle ti	he Buyer
Q	s not affected by any application to, queensland Civil and Administrative o a tree on the Lot or	or an order made by, the Tribunal (QCAT) in relation	to terminate the contract	: prior to Settlemi	ent.
re	affected by an application to, or ar elation to a tree on the Lot, a copy of the Buyer prior to the Buyer signin	of which has been given			

GST WIT	THHOLDING OBLIGATIONS	
Is the Buy	er registered for GST and acquiring the Lot for a creditable purpose?	← WARNING: the Buyer warrants in
(sele	ct whichever is applicable)	clause 2.5(6) that this information is true and correct.
	Yes	
	No	
the L	: An example of an acquisition for a creditable purpose would be the purchase of and by a building contractor, who is registered for GST, for the purposes of building use on the Land and selling it in the ordinary course of its business.]	
	gives notice to the Buyer in accordance with section 14-255(1)(a) of the glaw that:	← WARNING: All sellers of residential premises or potential residential land are required to complete this notice.
(sele	ct whichever is applicable)	Section 14-250 of the Withholding Lav applies to the sale of 'new residential
\checkmark	the Buyer <i>is not</i> required to make a payment under section 14-250 of the Withholding Law in relation to the supply of the Property	premises' or 'potential residential land (subject to some exceptions) and requires an amount to be withheld from
	the Buyer <i>is</i> required to make a payment under section 14-250 of the Withholding Law in relation to the supply of the Property. Under section 14-255(1) of the Withholding Law, the Seller is required to give further details prior to settlement.	the Purchase Price and paid to the ATO. The Seller should seek legal advice if unsure about completing this section.
The	REIQ Terms of Contract for Residential Lots in a Community Titles Sc Twelfth Edition Contain the Terms of this Contract.	heme (Pages 8-17)
SPECIA	L CONDITIONS	
1) In the e	vent that the buyer or seller signs the contract and transmits it electronically via	email or facsimile this contract shall
create a v	alid and binding obligation by both parties with the force and effect as if it was an	original.
SETTLE	MENT	
Settlemen	Date: 30 Days from the Contract date	←or the next Business Day if that is not a Business Day in the Place for Settlement.
Place for S	Settlement: Brisbane	← If Brisbane is inserted, this is a reference to Brisbane CBD.

INITIAL 000015022655

SIGNATURES

The contract may be subject to a 5 business day statutory cooling-off period. A termination penalty of 0.25% of the purchase price applies if the Buyer terminates the contract during the statutory cooling-off period. It is recommended the Buyer obtain an independent property valuation and independent legal advice about the contract and his or her cooling-off rights, before signing.

Witness: *	HER
Witness:	th - D
Electronic Signature)	the Buyer Signs using an
W	SIGN
witness:	HER
Witness:	
(Note: No witness is required if Electronic Signature)	the Seller signs using an
← Who acknowledges having received the look to hold that amount and any Balance Dep	oosit when received as
	Witness: (Note: No witness is required if Electronic Signature) Witness: (Note: No witness is required if Electronic Signature)

TERMS OF CONTRACT FOR RESIDENTIAL LOTS IN A COMMUNITY TITLES SCHEME

1. DEFINITIONS

1.1 In this contract:

- terms in **bold** in the Reference Schedule and the Disclosure Statement have the meanings shown opposite them unless the context requires otherwise; and
- (2) (a) "Approved Safety Switch" means a residual current device as defined in the Electrical Safety Regulation 2013;
 - (b) "ATO" means the Australian Taxation Office;
 - (c) "ATO Clearance Certificate" means a certificate issued under section 14-220(1) of the Withholding Law which is current on the date it is given to the Buyer;
 - (d) "Balance Purchase Price" means the Purchase Price, less the Deposit paid by the Buyer, adjusted under clause 2.6:
 - (e) "Bank" means an authorised deposit-taking institution within the meaning of the Banking Act 1959 (Cth).
 - (f) "Body Corporate" means the body corporate of the Scheme;
 - (g) "Body Corporate Debt" has the meaning in the Regulation Module but excludes the Body Corporate Levies for the period which includes the Settlement Date;
 - (h) "Body Corporate Levies" means regular periodic contributions levied on the owner of the Lot (including, if applicable, levied under an exclusive use by-law) excluding any Special Contribution:
 - "Bond" means a bond under the Residential Tenancies and Rooming Accommodation Act 2008;
 - "Building" means any building that forms part of the Lot or in which the Lot is situated;
 - (k) "Building Inspector" means a person licensed to carry out completed residential building inspections under the Queensland Building and Construction Commission Regulations 2003:
 - (I) "Business Day" means a day other than:
 - (i) a Saturday or Sunday
 - (ii) a public holiday in the Place for Settlement; and
 - (iii) a day in the period 27 to 31 December (inclusive).
 - (m) "CGT Withholding Amount" means the amount determined under section 14-200(3)(a) of the Withholding Law or, if a copy is provided to the Buyer prior to settlement, a lesser amount specified in a variation notice under section 14-235;
 - (n) "Compliance or Exemption Certificate" means:
 - (i) a Pool Safety Certificate; or
 - (ii) a building certificate that may be used instead of a Pool Safety Certificate under section 246AN(2) of the Building Act 1975;
 - (iii) an exemption from compliance on the grounds of impracticality under section 245B of the *Building Act 1975*;
 - (o) "Compliant Smoke Alarm" means a smoke alarm complying with the requirements for smoke alarms in domestic dwellings under the Fire and Emergency Services Act 1990;
 - (p) "Contract Date" or "Date of Contract" means the date inserted in the Reference Schedule;
 - (q) "Court" includes any tribunal established under statute;

- (r) "Disclosure Statement" means the statement under section 206 (Existing Lot) or section 213 (Proposed Lot) of the Body Corporate and Community Management Act 1997;
- (s) "Electronic Signature" means an electronic method of signing that identifies the person and indicates their intention to sign the contract;
- (t) "Encumbrances" includes:
 - (i) unregistered encumbrances
 - (ii) statutory encumbrances; and
 - (iii) Security Interests.
- (u) "Essential Term" includes, in the case of breach by:
 - (i) the Buyer: clauses 2.2, 2.5(1), 2.5(5),5.1 and 6.1; and
 - (ii) the Seller: clauses 2.5(5), 5.1, 5.3(1)(a) (d), 5.3(1)(e)(ii) & (iii), 5.3(1)(f), 5.5 and 6.1; but nothing in this definition precludes a Court from finding other terms to be essential;
- (v) "Exclusive Use Areas" means parts of the common property for the Scheme allocated to the Lot under an exclusive use by-law;
- (w) "Financial Institution" means a Bank, Building Society or Credit Union;
- (X) "General Purpose Socket Outlet" means an electrical socket outlet as defined in the Electrical Safety Regulation 2013;
- (y) "GST" means the goods and services tax under the GST Act;
- (z) "GST Act" means A New Tax System (Goods and Services Tax) Act 1999 (Cth) and includes other GST related legislation;
- (aa) "GST Withholding Amount" means the amount (if any) determined under section 14-250 of the Withholding Law required to be paid to the Commissioner of Taxation:
- (bb) "Improvements" means fixed structures in the Lot (such as stoves, hot water systems, fixed carpets, curtains, blinds and their fittings, clothes lines, fixed satellite dishes and television antennae, in-ground plants) but does not include the Reserved Items;
- (cc) "Keys" means keys, codes or devices in the Seller's possession or control for all locks or security systems on the Property or necessary to access the Property;
- (dd) "Land" means the scheme land for the Scheme;
- (ee) "Notice of No Pool Safety Certificate" means the Form 36 under the Building Regulation 2006 to the effect that there is no Pool Safety Certificate issued for the Land and/or the Lot;
- (ff) "Notice of nonconformity" means a Form 26 under the Building Regulation 2006 advising how the pool does not comply with the relevant pool safety standard;
- (gg) "Outgoings" means:
 - rates or charges on the Lot by any competent authority (for example, council rates, water rates, fire service levies) but excludes land tax; and
 - (ii) Body Corporate Levies.
- (hh) "Pest Inspector" means a person licensed to undertake termite inspections on completed buildings under the Queensland Building and Construction Commission Regulations 2003;
- (ii) "Pool Safety Certificate" has the meaning in section 231C(a) of the Building Act 1975;



- (jj) "Pool Safety Requirements" means the requirements for pool safety contained in the Building Act 1975 and Building Regulation 2006;
 - (kk) "Pool Safety Inspection Date" means the Pool Safety Inspection Date inserted in the Reference Schedule. If no date is inserted in the Reference Schedule, the Pool Safety Inspection Date is taken to be the earlier of the following:
 - (i) the Inspection Date for the Building and/or Pest Inspection; or
 - (ii) 2 Business Days before the Settlement Date;
 - (II) "Pool Safety Inspector" means a person authorised to give a Pool Safety Certificate;
- (mm) "PPSR" means the Personal Property Securities Register established under Personal Property Securities Act 2009 (Cth);
- (nn) "Property" means:
 - (i) the Lot;
 - (ii) the right to any Exclusive Use Areas;
 - (iii) the Improvements;
 - (iv) the Included Chattels;
- (oo) "Regulation Module" means the regulation module for the Scheme;
- (pp) "Rent" means any periodic amount, including outgoings, payable under the Tenancies;
- (qq) "Reserved Items" means the Excluded Fixtures and all Chattels in the Lot and Exclusive Use Areas other than the Included Chattels;
- (rr) "Scheme" means the community titles scheme containing the Lot;
- (ss) "Security Interests" means all security interests registered on the PPSR over Included Chattels and Improvements;
- (tt) "Special Contribution" means an amount levied by the Body Corporate under the Regulation Module for a liability for which no provision or inadequate provision has been made in the budget of the Body Corporate.
- (uu) "Transfer Documents" means:
 - the form of transfer under the Land Title Act 1994 required to transfer title in the Lot to the Buyer; and
 - (ii) any other document to be signed by the Seller necessary for stamping or registering the transfer;
- (vv) "Transport Infrastructure" has the meaning defined in the Transport Infrastructure Act 1994; and
- (ww) "Withholding Law" means Schedule 1 to the Taxation Administration Act 1953 (Cth).
- 1.2 Words and phrases defined in the Body Corporate and Community Management Act 1997 have the same meaning in this contract unless the context indicates otherwise.

2. PURCHASE PRICE

2.1 **GST**

- Unless otherwise specified in this contract, the Purchase Price includes any GST payable on the supply of the Property to the Buyer.
- (2) If a party is required to make any other payment or reimbursement under this contract, that payment or reimbursement will be reduced by the amount of any input tax credits to which the other party (or the representative member for a GST group of which it is a member) is entitled.

2.2 Deposit

(1) The Buyer must pay the Deposit to the Deposit Holder at the times shown in the Reference Schedule. The Deposit Holder will hold the Deposit until a party becomes entitled to it.

- (2) The Buyer will be in default if it:
 - (a) does not pay the Deposit when required;
 - (b) pays the Deposit by a post-dated cheque; or
 - (c) pays the Deposit by cheque which is dishonoured on presentation.
- (3) The Seller may recover from the Buyer as a liquidated debt any part of the Deposit which is not paid when required.

2.3 Investment of Deposit

lf:

- the Deposit Holder is instructed by either the Seller or the Buyer; and
- (2) it is lawful to do so;

the Deposit Holder must:

- (3) invest as much of the Deposit as has been paid with any Financial Institution in an interest-bearing account in the names of the parties; and
- (4) provide the parties' tax file numbers to the Financial Institution (if they have been supplied).

2.4 Entitlement to Deposit and Interest

- (1) The party entitled to receive the Deposit is:
 - (a) if this contract settles, the Seller;
 - (b) if this contract is terminated without default by the Buyer, the Buyer;
 - (c) if this contract is terminated owing to the Buyer's default, the Seller.
- (2) The interest on the Deposit must be paid to the person who is entitled to the Deposit.
- (3) If this contract is terminated, the Buyer has no further claim once it receives the Deposit and interest unless the termination is due to the Seller's default or breach of warranty.
- (4) The Deposit is invested at the risk of the party who is ultimately entitled to it.

2.5 Payment of Balance Purchase Price

- (1) On the Settlement Date, the Buyer must pay the Balance Purchase Price by Bank cheque as the Seller or the Seller's Solicitor directs.
- (2) Despite any other provision of this contract, reference to a "Bank cheque" in clause 2.5:
 - (a) includes a cheque drawn by a Building Society or Credit Union on itself;
 - (b) does not include a cheque drawn by a Building Society or Credit Union on a Bank;
 - and the Seller is not obliged to accept a cheque referred to in clause 2.5(2)(b) on the Settlement Date.
- (3) If both the following apply:
 - (a) the sale is not an excluded transaction under section 14-215 of the Withholding Law; and
 - (b) the Seller has not given the Buyer on or before settlement for each person comprising the Seller either:
 - (i) an ATO Clearance Certificate; or
 - (ii) a variation notice under section 14-235 of the Withholding Law which remains current at the Settlement Date varying the CGT Withholding Amount to nil,

then:

(c) for clause 2.5(1), the Seller irrevocably directs the Buyer to draw a bank cheque for the CGT Withholding Amount in favour of the Commissioner of Taxation or, if the Buyer's Solicitor requests, the Buyer's Solicitor's Trust Account;

- (d) the Buyer must lodge a Foreign Resident Capital Gains Withholding Purchaser Notification Form with the ATO for each person comprising the Buyer and give copies to the Seller with the payment reference numbers (PRN) on or before settlement;
- (e) the Seller must return the bank cheque in paragraph (c) to the Buyer's Solicitor (or if there is no Buyer's Solicitor, the Buyer) at settlement; and
- (f) the Buyer must pay the CGT Withholding Amount to the ATO in accordance with section 14-200 of the Withholding Law and give the Seller evidence that it has done so within 2 Business Days of settlement occurring.
- (4) For clause 2.5(3) and section 14-215 of the Withholding Law, the market value of the CGT asset is taken to be the Purchase Price less any GST included in the Purchase Price for which the Buyer is entitled to an input tax credit unless:
 - (a) the Property includes items in addition to the Lot and Improvements; and
 - (b) no later than 2 Business Days prior to the Settlement Date, the Seller gives the Buyer a valuation of the Lot and Improvements prepared by a registered valuer,

in which case the market value of the Lot and Improvements will be as stated in the valuation.

- (5) If the Buyer is required to pay the GST Withholding Amount to the Commissioner of Taxation at settlement pursuant to section 14-250 of the Withholding Law:
 - (a) the Seller must give the Buyer a notice in accordance with section 14-255(1) of the Withholding Law;
 - (b) prior to settlement the Buyer must lodge with the ATO:
 - (i) a GST Property Settlement Withholding Notification form ("Form 1"); and
 - (ii) a GST Property Settlement Date Confirmation form ("Form 2");
 - (c) on or before settlement, the Buyer must give the Seller copies of:
 - (i) the Form 1;
 - (ii) confirmation from the ATO that the Form 1 has been lodged specifying the Buyer's lodgement reference number and payment reference number;
 - (iii) confirmation from the ATO that the Form 2 has been lodged; and
 - (iv) a completed ATO payment slip for the Withholding Amount.
 - (d) the Seller irrevocably directs the Buyer to draw a bank cheque for the GST Withholding Amount in favour of the Commissioner of Taxation and deliver it to the Seller at settlement; and
 - (e) the Seller must pay the GST Withholding Amount to the ATO in compliance with section 14-250 of the Withholding Law promptly after settlement.
- (6) The Buyer warrants that the statements made by the Buyer in the Reference Schedule under GST Withholding Obligations are true and correct.

2.6 Adjustments to Balance Purchase Price

(1) The Seller is liable for Outgoings and is entitled to Rent up to and including the Settlement Date. The Buyer is liable for Outgoings and is entitled to Rent after the Settlement Date.

- (2) Subject to clauses 2.6(3), 2.6(5) and 2.6(17), Outgoings for periods including the Settlement Date must be adjusted:
 - (a) for those paid, on the amount paid;
 - (b) for those assessed but unpaid, on the amount payable (excluding any discount); and
 - (c) for those not assessed:
 - on the amount the relevant authority or the Body Corporate advises will be assessed (excluding any discount); or
 - (ii) if no advice on the assessment to be made is available, on the amount of the latest assessment (excluding any discount).
- (3) If there is no separate assessment of rates for the Lot at the Settlement Date and the Local Government informs the Buyer that it will not apportion rates between the Buyer and the Seller, then:
 - (a) the amount of rates to be adjusted is that proportion of the assessment equal to the ratio of the interest schedule lot entitlement of the Lot to the aggregate interest schedule lot entitlement of the Scheme; and
 - (b) if an assessment of rates includes charges imposed on a "per lot" basis, then the portion of those charges to be adjusted is the amount assessed divided by the number of lots in that assessment.
- (4) The Seller is liable for land tax assessed on the Lot for the financial year current at the Settlement Date. If land tax is unpaid at the Settlement Date and the Office of State Revenue advises that it will issue a final clearance for the Lot on payment of a specified amount, then the Buyer may deduct the specified amount from the Balance Purchase Price at settlement and must pay it promptly to the Office of State Revenue.
- (5) Any Outgoings assessable on the amount of water used must be adjusted on the charges that would be assessed on the total water usage for the assessment period, determined by assuming that the actual rate of usage shown by the meter reading made before settlement continues throughout the assessment period. The Buyer must obtain and pay for the meter reading.
- (6) If any Outgoings are assessed but unpaid at the Settlement Date, then the Buyer may deduct the amount payable from the Balance Purchase Price at settlement and pay it promptly to the relevant authority or the Body Corporate, as appropriate. If an amount is deducted under this clause, the relevant Outgoing will be treated as paid at the Settlement Date for the purposes of clause 2.6(2).
- (7) Arrears of Rent for any rental period ending on or before the Settlement Date belong to the Seller and are not adjusted at settlement.
- (8) Unpaid Rent for the rental period including both the Settlement Date and the following day ("Current Period") is not adjusted until it is paid.
- (9) Rent already paid for the Current Period or beyond must be adjusted at settlement.
- (10) If Rent payments are reassessed after the Settlement Date for periods including the Settlement Date, any additional Rent payment from a Tenant or refund due to a Tenant must be apportioned under clauses 2.6(7), 2.6(8) and 2.6(9) and 2.6(10).
- (11) Payments under clause 2.6(10) must be made within 14 days after notification by one party to the other but only after any additional payment from a Tenant has been received.



(12), The Seller is liable for:

- (a) any Special Contribution for which a levy notice has been issued on or before the Contract Date;
 and
- (b) any other Body Corporate Debt (including any penalty or recovery cost resulting from nonpayment of a Body Corporate Debt) owing in respect of the Lot at settlement.

The Buyer is liable for any Special Contribution levied after the Contract Date.

- (13) If an amount payable by the Seller under clause 2.6(12) is unpaid at the Settlement Date, the Buyer may deduct the specified amount from the Balance Purchase Price at settlement and must pay it promptly to the Body Corporate.
- (14) For the purposes of clause 2.6(12), an amount payable under an exclusive use by-law will be treated as levied on the date it is due.
- (15) The cost of Bank cheques payable at settlement:
 - (a) to the Seller or its mortgagee are the responsibility of the Buyer; and
 - (b) to parties other than the Seller or its mortgagee are the responsibility of the Seller.
- (16) The Seller is not entitled to require payment of the Balance Purchase Price by means other than Bank cheque without the consent of the Buyer.
- (17) Upon written request by the Buyer, the Seller will, prior to settlement, give the Buyer a written statement, supported by reasonable evidence, of —
 - (a) all Outgoings and all Rent for the Property to the extent they are not capable of discovery by search or enquiry at any office of public record or pursuant to the provisions of any statute; and
 - (b) any other information which the Buyer may reasonably require for the purpose of calculating or apportioning any Outgoings or Rent under this clause 2.6.

If the Seller becomes aware of a change to the information provided the Seller will as soon as practicably provide the updated information to the Buyer.

3. FINANCE

- **3.1** This contract is conditional on the Buyer obtaining approval of a loan for the Finance Amount from the Financier by the Finance Date on terms satisfactory to the Buyer. The Buyer must take all reasonable steps to obtain approval.
- 3.2 The Buyer must give notice to the Seller that:
 - approval has not been obtained by the Finance Date and the Buyer terminates this contract; or
 - (2) the finance condition has been either satisfied or waived by the Buyer.
- 3.3 The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 3.2 by 5pm on the Finance Date. This is the Seller's only remedy for the Buyer's failure to give notice.
- 3.4 The Seller's right under clause 3.3 is subject to the Buyer's continuing right to give written notice to the Seller of satisfaction, termination or waiver pursuant to clause 3.2.

4. BUILDING AND PEST INSPECTION REPORTS AND POOL SAFETY

4.1 Building and Pest Inspection

(1) This contract is conditional on the Buyer obtaining a written building report from a Building Inspector and a written pest report from a Pest Inspector (which may be a single report) on the Property by the Inspection Date on terms satisfactory to the Buyer. The Buyer must take all reasonable steps to obtain the reports (subject to the right of the Buyer to elect to obtain only one of the reports).

- (2) The Buyer must give notice to the Seller that:
 - (a) a satisfactory Inspector's report under clause 4.1(1) has not been obtained by the Inspection Date and the Buyer terminates this contract. The Buyer must act reasonably; or
 - (b) clause 4.1(1) has been either satisfied or waived by the Buyer.
- (3) If the Buyer terminates this contract and the Seller asks the Buyer for a copy of the building and pest reports, the Buyer must give a copy of each report to the Seller without delay.
- (4) The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 4.1(2) by 5pm on the Inspection Date. This is the Seller's only remedy for the Buyer's failure to give notice.
- (5) The Seller's right under clause 4.1(4) is subject to the Buyer's continuing right to give written notice to the Seller of satisfaction, termination or waiver pursuant to clause 4.1(2).

4.2 Pool Safety for non-shared pool on Lot

- (1) This clause 4.2 applies if:
 - (a) there is a pool on the Lot and the answer to Q2 of the Reference Schedule is No or Q2 is not completed (for a non-shared pool on the Lot); and
 - (b) this contract is not a contract of a type referred to in section 160(1)(b) of the *Property* Occupations Act 2014.
- (2) This contract is conditional upon:
 - (a) the issue of a Pool Safety Certificate; or
 - a Pool Safety Inspector issuing a Notice of Nonconformity stating the works required before a Pool Safety Certificate can be issued;
 - by the Pool Safety Inspection Date.
- (3) The Buyer is responsible for arranging an inspection by a Pool Safety Inspector at the Buyer's cost. The Seller authorises:
 - (a) the Buyer to arrange the inspection; and
 - (b) the Pool Safety Inspector to advise the Buyer of the results of the inspection and to give the Buyer a copy of any notice issued.
- (4) If a Pool Safety Certificate has not been issued by the Pool Safety Inspection Date, the Buyer may give notice to the Seller that the Buyer:
 - (a) terminates this contract; or
 - (b) waives the benefit of this clause 4.2

The Buyer must act reasonably.

- (5) The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 4.2(4) by 5pm on the Pool Safety Inspection Date.
- (6) The Seller's right under clause 4.2(5) is subject to the Buyer's continuing right to give written notice to the Seller of termination or waiver pursuant to clause 4.2(4).
- (7) The right of a party to terminate under this clause 4.2, ceases upon receipt by that party of a copy of a current Pool Safety Certificate.
- (8) If the Buyer terminates this contract under clause 4.2(4)(a), and the Seller has not obtained a copy of the Notice of Nonconformity issued by the Pool Safety Inspector, the Seller may request a copy and the Buyer must provide this to the Seller without delay.

5. SETTLEMENT

5.1 Time and Date

 Settlement must occur between 9am and 4pm AEST on the Settlement Date. (2) If the parties do not agree on where settlement is to occur, it must take place in the Place for Settlement at the office of a Solicitor or Financial Institution nominated by the Seller, or, if the Seller does not make a nomination, at the Land Registry Office in or nearest to the Place for Settlement.

5.2 Transfer Documents

- (1) The Transfer Documents must be prepared by the Buyer's Solicitor and delivered to the Seller a reasonable time before the Settlement Date.
- (2) If the Buyer pays the Seller's reasonable expenses, it may require the Seller to produce the Transfer Documents at the Office of State Revenue nearest the Place for Settlement for stamping before settlement.

5.3 Documents and Keys at Settlement

- In exchange for payment of the Balance Purchase Price, the Seller must deliver to the Buyer at settlement:
 - (a) any instrument of title for the Lot required to register the transfer to the Buyer; and
 - (b) unstamped Transfer Documents capable of immediate registration after stamping; and
 - (c) any instrument necessary to release any Encumbrance over the Property in compliance with the Seller's obligation in clause 7.2; and
 - (d) if requested by the Buyer not less than 2 clear Business Days before the Settlement Date, the Keys; and
 - (e) if there are Tenancies:
 - (i) the Seller's copy of any Tenancy agreements;
 - (ii) a notice to each tenant advising of the sale in the form required by law; and
 - (iii) any notice required by law to transfer to the Buyer the Seller's interest in any Bond; and
 - (f) if the answer to Q2 in the Reference Schedule is Yes, a copy of a current Compliance or Exemption Certificate, if not already provided to the Buyer.
- (2) If the Keys are not delivered at settlement under clause 5.3(1)(d), the Seller must deliver the Keys to the Buyer. The Seller may discharge its obligation under this provision by authorising the Seller's Agent to release the Keys to the Buyer.

5.4 Assignment of Covenants and Warranties

At settlement, the Seller assigns to the Buyer the benefit of all:

- covenants by the tenants under the Tenancies;
- (2) guarantees and Bonds (subject to the requirements of the Residential Tenancies and Rooming Accommodation Act 2008) supporting the Tenancies; and
- (3) manufacturers' warranties regarding the Included Chattels:
- (4) builders' warranties on the Improvements; to the extent that they are assignable and the Buyer accepts the assignment. However, the right to recover arrears of Rent is not assigned to the Buyer and section 117 of the *Property Law Act 1974* does not apply.

5.5 Possession of Property and Title to Included Chattels

On the Settlement Date, in exchange for the Balance Purchase Price, the Seller must give the Buyer vacant possession of the Lot and Exclusive Use Areas except for the Tenancies. Title to the Included Chattels passes at settlement.

5.6 Reservations

 The Seller must remove the Reserved Items from the Property before settlement.

- (2) The Seller must repair at its expense any damage done to the Property in removing the Reserved Items. If the Seller fails to do so, the Buyer may repair that damage.
- (3) Any Reserved Items not removed before settlement will be considered abandoned and the Buyer may, without limiting its other rights, complete this contract and appropriate those Reserved Items or dispose of them in any way.
- (4) The Seller indemnifies the Buyer against any damages and expenses resulting from the Buyer's actions under clauses 5.6(2) or 5.6(3).

6. TIME

6.1 Time of the Essence

Time is of the essence of this contract, except regarding any agreement between the parties on a time of day for settlement.

6.2 Suspension of Time

- (1) This clause 6.2 applies if a party is unable to perform a settlement obligation solely as a consequence of a Delay Event but does not apply where the inability is attributable to: –
 - (a) damage to, destruction of or diminution in value of the Property or other property of the Seller or Buyer; or
 - (b) termination or variation of any agreement between a party and another person whether relating to the provision of finance, the release of an Encumbrance, the sale or purchase of another property or otherwise.
- (2) Time for the performance of the parties' Settlement Obligations is suspended and ceases to be of the essence of the contract and the parties are deemed not to be in breach of their Settlement Obligations.
- (3) An Affected Party must take reasonable steps to minimise the effect of the Delay Event on its ability to perform its Settlement Obligations.
- (4) When an Affected Party is no longer prevented from performing its Settlement Obligations due to the Delay Event, the Affected Party must give the other party a notice of that fact, promptly.
- (5) When the Suspension Period ends, whether notice under clause 6.2(4) has been given or not, either party may give the other party a Notice to Settle.
- (6) A Notice to Settle must be in writing and state:
 - (a) that the Suspension Period has ended; and
 - (b) a date, being not less than 5 nor more than 10 Business Days after the date the Notice to Settle is given, which shall become the Settlement Date:
 - (c) that time is of the essence.
- (7) When Notice to Settle is given, time is again of the essence of the contract.
- (8) In this clause 6.2:
 - (a) "Affected Party" means a party referred to in clause 6.2(1);
 - (b) "Delay Event" means:
 - a tsunami, flood, cyclone, earthquake, bushfire or other act of nature;
 - (ii) riot, civil commotion, war, invasion or a terrorist act;
 - (iii) an imminent threat of an event in paragraphs (i) or (ii); or
 - (iv) compliance with any lawful direction or order by a Government Agency;

- (c) "Government Agency" means the government of the Commonwealth of Australia or an Australian State, Territory or local government and includes their authorities, agencies, government owned corporations and authorised officers, courts and tribunals;
 - (d) "Settlement Obligations" means, in the case of the Buyer, its obligations under clauses 2.5(1) and 5.1(1) and, in the case of the Seller, its obligations under clauses 5.1(1), 5.3(1)(a) – (e) and 5.5;
 - (e) "Suspension Period" means the period during which the Affected Party (or if both the Buyer and Seller are Affected Parties, either of them) remains unable to perform a settlement obligation solely as a consequence of a Delay Event.

7. MATTERS AFFECTING THE PROPERTY

7.1 Title

The Lot is sold subject to:

- (1) the Body Corporate and Community Management Act 1997 and the by-Laws of the Body Corporate;
- (2) any reservations or conditions on the title or the original Deed of Grant.

7.2 Encumbrances

The Property is sold free of all Encumbrances other than the Title Encumbrances, Tenancies, statutory easements implied by part 6A of the *Land Title Act 1994* and interests registered on the common property for the Scheme.

7.3 Requisitions

The Buyer may not deliver any requisitions or enquiries on title.

7.4 Seller's Warranties

- (1) The Seller warrants that, except as disclosed in this contract, at settlement:
 - (a) it will be the registered owner of an estate in fee simple in the Lot and will own the Improvements and Included Chattels;
 - (b) it will be capable of completing this contract (unless the Seller dies or becomes mentally incapable after the Contract Date); and
 - (c) there will be no unsatisfied judgment, order (except for an order referred to in clause 7.6(1)(b)) or writ affecting the Property.
- (2) The Seller warrants that, except as disclosed in this contract, at the Contract Date and at settlement there are no current or threatened claims, notices or proceedings that may lead to a judgment, order or writ affecting the Property.
- (3) The Seller warrants that, except as disclosed in this contract, at the Contract Date:
 - (a) there is no unregistered lease, easement or other right capable of registration and which is required to be registered to give indefeasibility affecting the common property or Body Corporate assets;
 - (b) there is no proposal to record a new community management statement for the Scheme and it has not received a notice of a meeting of the Body Corporate to be held after the Contract Date or notice of any proposed resolution or a decision of the Body Corporate to consent to the recording of a new community management statement for the Scheme;
 - (c) all Body Corporate consents to improvements made to common property and which benefit the Lot, or the registered owner of the Lot, are in force; and

- (d) the Additional Body Corporate Information is correct (if completed).
- (4) If the Seller breaches a warranty in clause 7.4(1) or clause 7.4(2), the Buyer may terminate this contract by notice to the Seller.
- (5) If:
 - (a) the Seller breaches a warranty in clause 7.4(3); or
 - (b) the Additional Body Corporate Information is not completed;

and, as a result, the Buyer is materially prejudiced, the Buyer may terminate this contract by notice to the Seller given within 14 days after the Contract Date but may not claim damages or compensation.

- (6) Clauses 7.4(4) and 7.4(5) do not restrict any statutory rights the Buyer may have which cannot be excluded by this contract.
- (7) (a) The Seller warrants that, except as disclosed in this contract or a notice given by the Seller to the Buyer under the *Environmental Protection Act 1994* (**'EPA**"), at the Contract Date:
 - there is no outstanding obligation on the Seller to give notice to the administering authority under EPA of notifiable activity being conducted on the Land; and
 - (ii) the Seller is not aware of any facts or circumstances that may lead to the Land being classified as contaminated land within the meaning of EPA.
 - (b) If the Seller breaches a warranty in clause 7.4(7), the Buyer may:
 - terminate this contract by notice in writing to the Seller given no later than 2 Business Days before the Settlement Date; or
 - (ii) complete this contract and claim compensation, but only if the Buyer claims it in writing before the Settlement Date.
- (8) The Seller does not warrant that the Present Use is lawful.

7.5 Survey and Mistake

- (1) The Buyer may survey the Lot.
- (2) If there is:
 - (a) an error in the boundaries or area of the Lot;
 - (b) an encroachment by structures onto or from the Lot; or
 - (c) a mistake or omission in describing the Lot or the Seller's title to it;

which is:

- (d) immaterial; or
- (e) material, but the Buyer elects to complete this contract;

the Buyer's only remedy against the Seller is for compensation, but only if claimed by the Buyer in writing on or before settlement.

- (3) The Buyer may not delay settlement or withhold any part of the Balance Purchase Price because of any compensation claim under clause 7.5(2).
- (4) If there is a material error, encroachment or mistake, the Buyer may terminate this contract before settlement.

7.6 Requirements of Authorities

- (1) Subject to clause 7.6(5), any valid notice or order by any competent authority or Court requiring work to be done or money spent in relation to the Property ("Work or Expenditure") must be fully complied with:
 - (a) if issued before the Contract Date, by the Seller before the Settlement Date;

- (b) if issued on or after the Contract Date, by the Buyer.
- (2) If any Work or Expenditure that is the Seller's responsibility under clause 7.6(1)(a) is not done before the Settlement Date, the Buyer is entitled to claim the reasonable cost of work done by the Buyer in accordance with the notice or order referred to in clause 7.6(1) from the Seller after settlement as a debt
- (3) Any Work or Expenditure that is the Buyer's responsibility under clause 7.6(1)(b), which is required to be done before the Settlement Date, must be done by the Seller unless the Buyer directs the Seller not to and indemnifies the Seller against any liability for not carrying out the work. If the Seller does the work, or spends the money, the reasonable cost of that Work or Expenditure must be added to the Balance Purchase Price.
- (4) The Buyer may terminate this contract by notice to the Seller if there is an outstanding notice at the Contract Date under section 246AG of the Building Act 1975 that affects the Property. The Buyer may terminate this contract by notice to the Seller if there is an outstanding notice at the Contract Date under sections 247 or 248 of the Building Act 1975 or sections 167 or 168 of the Planning Act 2016 that affects the Property or Land.
- (5) Clause 7.6(1) does not apply to orders disclosed under section 83 of the *Neighbourhood Disputes* (Dividing Fences and Trees) Act 2011.

7.7 Property Adversely Affected

- (1) If at the Contract Date:
 - (a) the Present Use is not lawful under the relevant town planning scheme;
 - (b) the Land is affected by a proposal of any competent authority to alter the dimensions of any Transport Infrastructure or locate Transport Infrastructure on the Land;
 - (c) access or any service to the Land passes unlawfully through other land;
 - (d) any competent authority has issued a current notice to treat, or notice of intention to resume, regarding any part of the Land;
 - (e) there is an outstanding condition of a development approval attaching to the Land under section 73 of the Planning Act 2016 or section 96 of the Economic Development Queensland Act 2012 which, if complied with, would constitute a material mistake or omission in the Seller's title under clause 7.5(2)(c);
 - (f) the Property is affected by the Queensland Heritage Act 1992 or is included in the World Heritage List;
 - (g) the Property is declared acquisition land under the Queensland Reconstruction Authority Act 2011: or
 - (h) there is a charge against the Lot under section 104 of the Foreign Acquisitions and Takeovers Act 1975 (Cth),
 - and that has not been disclosed in this contract, the Buyer may terminate this contract by notice to the Seller given on or before settlement.
- (2) If no notice is given under clause 7.7(1), the Buyer will be treated as having accepted the Property subject to all of the matters referred to in that clause.
- (3) The Seller authorises the Buyer to:
 - inspect records held by any authority, including Security Interests on the PPSR, relating to the Property or the Land; and
 - (b) apply for a certificate of currency of the Body Corporate's insurance from any insurer.

7.8 Dividing Fences

Notwithstanding any provision in the *Neighbourhood Disputes (Dividing Fences and Trees) Act 2011*, the Seller need not contribute to the cost of building any dividing fence between the Lot and any adjoining land owned by it. The Buyer waives any right to claim contribution from the Seller.

8. RIGHTS AND OBLIGATIONS UNTIL SETTLEMENT

8.1 Risk

The Property is at the Buyer's risk from 5pm on the first Business Day after the Contract Date.

8.2 Access

After reasonable notice to the Seller, the Buyer and its consultants may enter the Property:

- (1) once to read any meter;
- (2) for inspections under clause 4;
- (3) once to inspect the Property before settlement; and
- (4) once to value the Property before settlement.

8.3 Seller's Obligations After Contract Date

- (1) The Seller must use the Property reasonably until settlement. The Seller must not do anything regarding the Property or Tenancies that may significantly alter them or result in later expense for the Buyer.
- (2) The Seller must promptly upon receiving any notice, proceeding or order that affects the Property or requires work on the Property, give a copy to the Buyer.
- (3) Without limiting clause 8.3(1), the Seller must not without the prior written consent of the Buyer, give any notice or seek or consent to any order that affects the Property or make any agreement affecting the Property that binds the Buyer to perform.

8.4 Body Corporate Meetings

- (1) The Seller must promptly give the Buyer a copy of:
 - (a) any notice it receives of a proposed meeting of the Body Corporate to be held after the Contract Date; and
 - (b) resolutions passed at that meeting and prior to settlement.
- (2) The Buyer may terminate this contract by notice in writing to the Seller given before settlement if it is materially prejudiced by:
 - (a) any resolution of the Body Corporate passed after the Contract Date, other than a resolution, details of which are disclosed to the Buyer in this contract; or
 - (b) where the Scheme is a subsidiary scheme, any resolution of a Body Corporate of a higher scheme.
- (3) In clause 8.4(2) a resolution includes a decision of the Body Corporate Committee to consent to recording a new community management statement.
- (4) If the Buyer is not given a copy of the resolutions before settlement, it may sue the Seller for damages.

8.5 Information Regarding the Property

Upon written request of the Buyer but in any event before settlement, the Seller must give the Buyer:

- copies of all documents relating to any unregistered interests in the Property;
- full details of the Tenancies to allow the Buyer to properly manage the Property after settlement;
- (3) sufficient details (including the date of birth of each Seller who is an individual) to enable the Buyer to undertake a search of the PPSR;

8.6 Possession Before Settlement

If possession is given before settlement:

- the Buyer must maintain the Property in substantially its condition at the date of possession, fair wear and tear excepted;
- (2) entry into possession is under a licence personal to the Buyer revocable at any time and does not:
 - (a) create a relationship of landlord and tenant; or
 - (b) waive the Buyer's rights under this contract;
- the Buyer must insure the Property to the Seller's satisfaction; and
- (4) the Buyer indemnifies the Seller against any expense or damages incurred by the Seller as a result of the Buyer's possession of the Property.

8.7 Seller's Obligations After Contract Date

- the Seller must promptly upon receiving any notice, proceeding or order that affects the Property or requires work on the Property, give a copy to the Buyer.
- (2) After the Contract Date, the Seller must not without the prior written consent of the Buyer, give any notice, seek or consent to any order or make an agreement that affects the Property.

9. PARTIES' DEFAULT

9.1 Seller and Buyer May Affirm or Terminate

Without limiting any other right or remedy of the parties including those under this contract, or any right at law or in equity, if the Seller or Buyer, as the case may be, fails to comply with an Essential Term, or makes a fundamental breach of an intermediate term, the Seller (in the case of the Buyer's default) or the Buyer (in the case of the Seller's default) may affirm or terminate this contract.

9.2 If Seller Affirms

If the Seller affirms this contract under clause 9.1, it may sue the Buyer for:

- (1) damages:
- (2) specific performance; or
- (3) damages and specific performance.

9.3 If Buyer Affirms

If the Buyer affirms this contract under clause 9.1, it may sue the Seller for:

- (1) damages;
- (2) specific performance; or
- (3) damages and specific performance.

9.4 If Seller Terminates

If the Seller terminates this contract under clause 9.1, it may do all or any of the following:

- (1) resume possession of the Property;
- (2) forfeit the Deposit and any interest earned;
- (3) sue the Buyer for damages;
- (4) resell the Property.

9.5 If Buyer Terminates

If the Buyer terminates this contract under clause 9.1, it may do all or any of the following:

- (1) recover the Deposit and any interest earned;
- (2) sue the Seller for damages.

9.6 Seller's Resale

- If the Seller terminates this contract and resells the Property, the Seller may recover from the Buyer as liquidated damages:
 - (a) any deficiency in price on a resale; and
 - (b) its expenses connected with any repossession, any failed attempt to resell, and the resale,

provided the resale settles within 2 years of termination of this contract.

(2) Any profit on a resale belongs to the Seller.

9.7 Seller's Damages

The Seller may claim damages for any loss it suffers as a result of the Buyer's default, including its legal costs on an indemnity basis and the cost of any Work or Expenditure under clause 7.6(3).

9.8 Buyer's Damages

The Buyer may claim damages for any loss it suffers as a result of the Seller's default, including its legal costs on an indemnity basis.

9.9 Interest on Late Payments

- (1) The Buyer must pay interest at the Default Rate:
 - (a) on any amount payable under this contract which is not paid when due; and
 - on any judgement for money payable under this contract.
- (2) Interest continues to accrue:
 - (a) under clause 9.9(1)(a), from the date it is due until paid; and
 - (b) under clause 9.9(1)(b), from the date of judgement until paid.
- (3) Any amount payable under clause 9.9(1)(a) in respect of a period prior to settlement must be paid by the Buyer at settlement. If this contract is terminated or if any amount remains unpaid after settlement, interest continues to accrue.
- (4) Nothing in this clause affects any other rights of the Seller under this contract or at law.

10. GENERAL

10.1 Seller's Agent

The Seller's Agent is appointed as the Seller's agent to introduce a Buyer.

10.2 Foreign Buyer Approval

The Buyer warrants that either:

- the Buyer's purchase of the Property is not a notifiable action; or
- (2) the Buyer has received a no objection notification, under the *Foreign Acquisitions and Takeovers Act 1975* (Cth).

10.3 Duty

The Buyer must pay all duty on this contract.

10.4 Notices

- (1) Notices under this contract must be in writing.
- (2) Notices under this contract or notices required to be given by law may be given and received by the party's solicitor.
- (3) Notices under this contract or notices required to be given by law may be given by:
 - (a) delivering or posting to the other party or its solicitor; or
 - (b) sending it to the facsimile number of the other party or its solicitor stated in the Reference Schedule (or another facsimile number notified by the recipient to the sender); or
 - (c) sending it to the email address of the other party or its solicitor stated in the Reference Schedule (or another email address notified by the recipient to the sender).
- (4) Subject to clause 10.4(5), a notice given after this contract is entered into in accordance with clause 10.4(3) will be treated as given:
 - (a) 5 Business Days after posting;
 - (b) if sent by facsimile, at the time indicated on a clear transmission report; and
 - (c) if sent by email, at the time it is sent.

- (5) Notices given by facsimile, by personal delivery or by 10.10 Counterparts email between 5pm on a Business Day (the "first Business Day") and 9am on the next Business Day (the "second Business Day") will be treated as given or delivered at 9am on the second Business Day.
- (6) If two or more notices are treated as given at the same time under clause 10.4(5), they will be treated as given in the order in which they were sent or delivered.
- (7) Notices or other written communications by a party's solicitor (for example, varying the Inspection Date, Finance Date or Settlement Date) will be treated as given with that party's authority.
- For the purposes of clause 10.4(3)(c) and clause 12.2 the notice or information may be contained within an email, as an attachment to an email or located in an electronic repository accessible by the recipient by clicking a link in an email.

10.5 Business Days

- (1) If anything is required to be done on a day that is not a Business Day, it must be done instead on the next Business Day.
- If the Finance Date or Inspection Date fall on a day that is not a Business Day, then it falls on the next Business Day.

10.6 Rights After Settlement

Despite settlement and registration of the transfer, any term of this contract that can take effect after settlement or registration remains in force.

10.7 Further Acts

If requested by the other party, each party must, at its own expense, do everything reasonably necessary to give effect to this contract.

10.8 Severance

If any term or part of a term of this contract is or becomes legally ineffective, invalid or unenforceable in any jurisdiction it will be severed and the effectiveness, validity or enforceability of the remainder will not be affected.

10.9 Interpretation

(1) Plurals and Genders

Reference to:

- (a) the singular includes the plural and the plural includes the singular;
- one gender includes each other gender;
- (c) a person includes a Body Corporate; and
- (d) a party includes the party's executors, administrators, successors and permitted assigns.

(2) Parties

- (a) If a party consists of more than one person, this contract binds them jointly and each of them individually.
- (b) A party that is a trustee is bound both personally and in its capacity as a trustee.

(3) Statutes and Regulations

Reference to statutes includes all statutes amending, consolidating or replacing them.

(4) Inconsistencies

If there is any inconsistency between any provision added to this contract and the printed provisions, the added provision prevails.

(5) Headings

Headings are for convenience only and do not form part of this contract or affect its interpretation.

- (1) This contract may be executed in two or more counterparts, all of which will together be deemed to constitute one and the same contract.
- A counterpart may be electronic and signed using an Electronic Signature.

11. FLECTRONIC SETTLEMENT

11.1 Application of Clause

- (1) Clause 11 applies if the Buyer, Seller and each Financial Institution involved in the transaction agree to an Electronic Settlement using the same ELNO System and overrides any other provision of this contract to the extent of any inconsistency.
- Acceptance of an invitation to an Electronic Workspace is taken to be an agreement for clause 11.1(1)
- (3) Clause 11 (except clause 11.5(3)) ceases to apply if either party gives notice under clause 11.5 that settlement will not be an Electronic Settlement.

11.2 Completion of Electronic Workspace

- (1) The parties must:
 - (a) ensure that the Electronic Workspace is completed and all Electronic Conveyancing Documents and the Financial Settlement Schedule are Digitally Signed prior to settlement; and
 - do everything else required in the Electronic (b) Workspace or otherwise to enable settlement to occur on the Settlement Date.
- (2) If the parties cannot agree on a time for settlement, the time to be nominated in the Workspace is 4pm AEST.
- (3) If any part of the Purchase Price is to be paid to discharge an Outgoing:
 - (a) the Buyer may, by notice in writing to the Seller, require that the amount is paid to the Buyer's Solicitor's trust account and the Buyer is responsible for paying the amount to the relevant authority;
 - for amounts to be paid to destination accounts other than the Buyer's Solicitor's trust account, the Seller must give the Buyer a copy of the current account for the Outgoing to enable the Buyer to verify the destination account details in the Financial Settlement Schedule.
- (4) If the Deposit is required to discharge any Encumbrance or pay an Outgoing at settlement:
 - (a) the Deposit Holder must, if directed by the Seller at least 2 Business Days prior to Settlement, pay the Deposit (and any interest accrued on investment of the Deposit) less commission as clear funds to the Seller's Solicitor;
 - the Buyer and the seller authorise the Deposit Holder to make the payment in clause 11(4)(a);
 - the Seller's Solicitor will hold the money as Deposit Holder under the Contract;
 - the Seller and Buyer authorise the Seller's Solicitor to pay the money as directed by the Seller in accordance with the Financial Settlement Schedule.

11.3 Electronic Settlement

- (1) Clauses 5.1(2) and 5.2 do not apply.
- Payment of the Balance Purchase Price electronically as directed by the Seller's Solicitor in the Financial Settlement Schedule satisfies the Buyer's obligation in clause 2.5(1).



- (3) The Seller and Buyer will be taken to have complied with
 - (a) clause 2.5(3)(c), (e) and (f); and
 - (b) clause 2.5(5)(d) and (e),

(as applicable) if at settlement the Financial Settlement Schedule specifies payment of the relevant amount to the account nominated by the Commissioner of Taxation.

- (4) The Seller will be taken to have complied with clause 5.3(1)(b), (c), (d), (e) and (f) if:
 - (a) in relation to documents which are suitable for Electronic Lodgement in the Land Registry at settlement, the documents are Digitally Signed within the Electronic Workspace; and
 - (b) in relation to any other document or thing, the Seller's Solicitor:
 - (i) confirms in writing prior to settlement that it holds all relevant documents which are not suitable for Electronic Lodgement and all Keys (if requested under clause 5.3(1)(d)) in escrow on the terms contained in the QLS E-Conveyancing Guidelines; and
 - (ii) gives a written undertaking to send the documents and Keys (if applicable) to the Buyer or Buyer's Solicitor no later than the Business Day after settlement; and
 - (iii) if requested by the Buyer, provides copies of documents in the Seller's Solicitors possession.
- (5) A party is not in default to the extent it is prevented from complying with an obligation because the other party or the other party's Financial Institution has not done something in the Electronic Workspace.
- (6) Any rights under the contract or at law to terminate the contract may not be exercised during the time the Electronic Workspace is locked for Electronic Settlement.
- (7) Electronic Settlement is taken to occur when Financial Settlement is effected, whether or not Electronic Lodgement has occurred.

11.4 Computer System Unavailable

If settlement fails and cannot occur by 4pm AEST on the Settlement Date because a computer system operated by the Land Registry, Office of State Revenue, Reserve Bank, a Financial Institution or the relevant ELNO System is inoperative, neither party is in default and the Settlement Date is deemed to be the next Business Day. Time remains of the essence.

11.5 Withdrawal from Electronic Settlement

- Either party may elect not to proceed with an Electronic Settlement by giving written notice to the other party.
- (2) A notice under clause 11.5(1) may not be given later than 5 Business Days before the Settlement Date unless an Electronic Settlement cannot be effected because:
 - (a) the transaction is not a Qualifying Conveyancing Transaction; or
 - (b) a party's solicitor is unable to complete the transaction due to death, a loss of legal capacity or appointment of a receiver or administrator (or similar) to their legal practice or suspension of their access to the ELNO System; or
 - (c) the Buyer's or Seller's Financial Institution is unable to use the relevant ELNO System to effect Electronic Settlement.

- (3) If clause 11.5(2) applies:
 - the party giving the notice must provide satisfactory evidence of the reason for the withdrawal; and
 - (b) the Settlement Date will be extended to the date 5 Business Days after the Settlement Date.

11.6 Costs

Each party must pay its own fees and charges of using the relevant ELNO System for Electronic Settlement.

11.7 Definitions for clause 11

In clause 11:

"Digitally Sign" and "Digital Signature" have the meaning in the ECNL.

"ECNL" means the Electronic Conveyancing National Law (Queensland).

"Electronic Conveyancing Documents" has the meaning in the Land Title Act 1994.

"Electronic Lodgement" means lodgement of a document in the Land Registry in accordance with the ECNL.

"Electronic Settlement" means settlement facilitated by an ELNO System.

"Electronic Workspace" means a shared electronic workspace within an ELNO System that allows the Buyer and Seller to effect Electronic Lodgement and Financial Settlement.

"ELNO" has the meaning in the ECNL.

"ELNO System" means a system provided by the ELNO for facilitating Financial Settlement and Electronic Lodgement.

"Financial Settlement" means the exchange of value between Financial Institutions facilitated by an ELNO System in accordance with the Financial Settlement Schedule.

"Financial Settlement Schedule" means the electronic settlement schedule within the Electronic Workspace listing the source accounts and destination accounts.

"Qualifying Conveyancing Transaction" means a transaction that is not excluded for Electronic Settlement by the rules issued by the relevant ELNO, Office of State Revenue, Land Registry, or a Financial Institution involved in the transaction.

12. ELECTRONIC CONTRACT AND DISCLOSURE

12.1 Electronic Signing

If this contract is signed by any person using an Electronic Signature, the Buyer and the Seller:

- (a) agree to enter into this contract in electronic form;
- (b) consent to either or both parties signing the contract using an Electronic Signature.

12.2 Pre-contract Disclosure

The Buyer consents to the Seller's use of electronic communication to give any notice or information required by law to be given to the Buyer and which was given before the Buyer signed this contract.



Annexure A

1. MODIFICATIONS TO TERMS OF CONTRACT

- 1.1 The Terms of Contract for Houses and Residential Land which form part of this Contract (the Standard Conditions) are amended as follows:
- (a) This Contract is not subject to finance and Clause 3 is deleted in its entirety;
- (b) This Contract is not subject to a Building, Pest or Pool Inspection and clause 4 is deleted in its entirety;
- (c) Clause 5.1 (1) is amended by replacing the words 4 pm AEST with 5 pm AEST;
- (d) The following clauses are deleted: clauses 7.4 (2) and (3); clauses 7.5 (2), (3) and (4); clause 7.6 (4) and clauses 7.7 (1) and (2);
- (e) Clause 7.4 (4) is amended to delete the words or clause 7.4 (2);
- (f) A new clause 7.5 (2) is inserted as follows

The Buyer cannot not object to, delay settlement, terminate this Contract, deduct or retain any amount from the Purchase Price or make any requisition or claim for compensation about any error in the boundaries or area of the Land, any encroachment onto or from the Land or a mistake or omission in describing the Property or the Seller's Title to it (whether material or otherwise).

- 1.2 The Standard Conditions do not apply to this Contract to the extent of any inconsistency with these special conditions and the latter will prevail.
- 2. BUYER HAS SATISFIED ITSELF
- 2.1 The Buyer acknowledges and agrees that it has inspected or has had the opportunity to inspect the Property and has satisfied itself about:
- (a)The quality, condition and state of repair of the Property;
- (b) The purposes for which the Property can be used (including without limitation any restrictions on its use or development);
- (c) The fitness of the Property for the purposes the Buyer wants it;
- (d)The existence or otherwise of necessary approvals and licences from authorities relating to the Property including any failure to comply with any approvals or licences;
- (e)The carrying on of any notifiable activity on the land or the existence or otherwise of any hazardous contaminant on the land within the meaning of the Environmental Protection Act 1994;
- (f) The existence or otherwise of any legal and physical defects;
- (g) Accepts the Property on an as is where is basis, in its present condition and subject to any latent or patent defects.
- 2.2 The Buyer acknowledges and agrees that it has not relied on any representations, warranties or information provided or statements about the Property made by or on behalf of the Seller, the Seller's Agent or their respective employees or agents, other than as set out in this Contract.
- 3. Inventory
- 1 x Microwave
- 1x Fridge
- 1x Electric Jug
- 1x Toaster
- Assorted Cutlery + Crockery+ Glasses + Pots/Pans
- 1x Desk
- 1x Chair
- 2x cane chairs
- 2x small bedside tables
- 1x lamp
- 1x blanket box/coffee table
- 1x foldout sofa bed lounge
- 1x TV
- 1x outdoor setting (table + 4x chairs)
- 1x double (split single) bed





Waiver/Shortening of Cooling off Period by a Proposed Buyer

Issued pursuant to section 167 of the Property Occupations Act 2014

NOTE FOR COMPLETION

Under the Property Occupations Act 2014 a buyer may:

- 1. waive their cooling off rights before entering into a relevant contract; or
- 2. shorten their cooling off rights after they enter into a relevant contract.

This form should be completed and signed either before or after the relevant contract is entered into depending on whether the buyer wishes to exercise option 1 or 2 above.

whether the buyer wishes to exercise option 1 or 2 above.	
A "relevant contract" is defined under section 160 of the Ac	t.
I/We	
(insert name/s of proposed buyer/s)	
understand that I am/ we are entitled to a five business day 160 and 166 of the Property Occupations Act 2014 in relati	statutory cooling off period available under sections on to a contract for the sale of the property located at:
Course Cours Island Depart	
	STATE: QLD POSTCODE: 4216
("the Property")	
 I/ we wish to: ✓ waive my/our statutory cooling off rights. ☐ shorten my/our statutory cooling off rights from 5 bus 	siness days to days.
I/ We agree that I have/ we have voluntarily agreed to waiv	(insert relevant number of days) e or shorten my/our statutory cooling off rights.
SIGNATURES	
Buyer Name:	Buyer Name:
Signature: 🐣	Signature: SIGN HERE
Date:	Date:

INITIALS









Contract for Residential Lots in a Community Titles Scheme

Twelfth Edition

This document has been approved by The Real Estate Institute of Queensland Limited and the Queensland Law Society Incorporated as being suitable for the sale and purchase of Residential Lots in a Community Titles Scheme in Queensland except for new residential property in which case the issue of GST liability must be dealt with by special condition.

The Seller and Buyer agree to sell and buy the Property under this contract.

	•	3				
REFER	RENCE SC	HEDULE				
Contrac	t Date:					
SELLE	R'S AGENT					
NAME: C	obalt Blue Re	sources Pty Ltd				
ABN: 81 600 9	50 016			LICENCE NO: 4204690		
ADDRESS	PO Box 16					
SUBURB:	Toowong				STATE: QLD	POSTCODE: 4066
PHONE: 3870720		MOBILE: 0421878116	FAX:	EMAIL: rwilson@rwilson.co		
SELLEI NAME: Coralspr					ABN:	
ADDRESS	² Bega Road	l				
SUBURB:	Kingston				STATE: QLD	POSTCODE: 4114
PHONE: 34583408	8	MOBILE:	FAX:	EMAIL: brian.lincez@rising	suncg.com.au_	
NAME:					ABN:	
ADDRESS:	:					
SUBURB:					STATE:	POSTCODE:
PHONE:		MOBILE:	FAX:	EMAIL:		
NAME:	R'S SOLICIT			← or a	ny other solicitor noti	fied to the Buyer
	IcCullough Lav					
REF:		CONTACT: Vicki Partanen				
ADDRESS:	2 Bayliss St					
SUBURB:	Toowong				STATE: QLD	POSTCODE: 4066
PHONE: 07387082	244	MOBILE:	FAX:	EMAIL: admin@lockettmcc	ullough.com.au	

INITIALS (Note: initials not required if signed with Electronic Signature)



BUYER NAME:				AB	N:
NAME.				AL	14.
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
NAME:				AB	N:
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
BUYER'S AGE	NT (If applicable)				
ABN:			LICENCE NO:		
ADDRESS:					_
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:	_	
BUYER'S SOLI	ICITOR		← or al	ny other solicitor no	otified to the Seller
REF:	CONTACT:				
ADDRESS:					
SUBURB:				STATE:	POSTCODE:
PHONE:	MOBILE:	FAX:	EMAIL:		
PROPERTY Lot: ADDRESS:	Marine Apartment 1504 (Couran Cove Island R	esort		
		_			DOSTOODE: 404.5
SUBURB: Description: Lot:	South Stradbroke Lot 62	on: 🗸 BUP		STATE: QLD	POSTCODE: 4216
Scheme:			nity Titles Scheme:		
Title Reference:	50220138				
Present Use:	Residential Unit				
Local Government	: Gold Coast City				

Excluded Fixtures	:					
Included Chattels:	Refer t	o Annexure A				
PRICE Deposit Holder:	Cob	alt Blue Resources Pt	y Ltd			
Deposit Holder's T	rust Acc	ount: Cobalt Blue Reso	ources Statutory	Trust Account		
Bank:	Wes					
BSB: ₀₃₄₋₀₆₁		Account No: 424	1-099			
Purchase Price:	\$					← Unless otherwise specified in this contract, the Purchase Price includes any GST payable on the supply of the Property to the Buyer.
Deposit:	\$ 10%	of sale price	specified below	ayable on the day the	he Buyer signs this c	contract unless another time is
	\$			t (if any) payable or ss days from the da		
Default Interest Ra	te:	%	← If no figure is ir Queensland La	serted, the Contract w Society Inc will ap	Rate applying at the C ply.	Contract Date published by the
FINANCE						
Finance Amount:	\$			←Unless all of "Fina completed, this cont not apply.	ance Amount", "Financ ontract is not subject t	cier" and "Finance Date" are o finance and clause 3 does
Financier:						
Finance Date:						
BUILDING AND		EST INSPECTION D		(If "Inspection Date" is not subject to an insp does not apply.	s not completed, the contract is pection report and clause 4.1
MATTERS AFFI	es:	G PROPERTY to any Encumbrances?	√ No	Yes, listed below:		IG TO SELLER: You are required
					to disclos will reman easemen easemen which ma Failure to Buyer to i compens	the all Title Encumbrances which in after settlement (for example, its on your title and statutory its for sewerage and drainage by not appear on a title search), of disclose these may entitle the terminate the contract or to ation. It is NOT sufficient to state itle", "search will reveal", or similate.
Tenancies: TENANT'S NAME: NIL TERM AND OPTIONS:	:			←If the property is s Otherwise comple	old with vacant posses te details from Reside	ssion from settlement, insert 'Nil'. ntial Tenancy Agreement.
STARTING DATE OF 1	TERM:	ENDING DATE OF TERM	#: RENT:		BOND:	
			Ψ		—	

INITIALS (Note: initials not required if signed with Electronic Signature)



Ma	anaging Age	nt:						
AG	SENCY NAME:							
N/	'A							
PR	OPERTY MANA	AGER:						
N/	<u>'A</u>							
AD	DRESS:		<u></u>					
SU	IBURB:				S	TATE:	POSTCODE:	
PH	IONE:	MOBILE:	FAX:	EMAIL:				
P	OOL SAFE	TY FOR NON-SHARE	POOLS					
Co	mnlete the foll	owing questions if there is a	non-shared nool in the Lot	← WARNING	← WARNING TO SELLER: Failure to comply with the Pool			
	•	non-shared pool on the L	•				with substantial penalties	
Ų.	L. IS lilete a l	ion-snared poor on the L	.ot?				no Compliance or	
	Yes			Exemption responsible	n Certificate at le at its cost to	Settlement obtain a P	t, the Buyer becomes ool Safety Certificate	
	√ No	Clause 4.2 of this cont	ract does not apply	within 90 d	days after settl	settlement. The Buyer can also become ts of rectification necessary to comply		
02	If the answ	ver to 01 is Ves is there:	a Compliance or Exemption	with the P	ool Safety Red	uirements	to obtain a Pool Safety	
Ų.	If the answer to Q1 is Yes, is there a Compliance or Exemption Certificate for the non-shared pool at the time of contract?			Ceruncaic	Certificate. The Buyer commits an offence and can be liable to substantial penalties if the Buyer fails to comply with this			
	Yes	Clause 5.3(1)(f) applie	\$	requireme				
	☐ No	. , . ,	cept for auction and some			. ot and Q2	is not completed then	
		other excluded sales)	cept for auction and some	clause 4.2	аррнез.			
Q3	3. If the answ	er to Q2 is No, has a Not	← Note: This	← Note: This is an obligation of the Seller under section 16				
	Certificate been given prior to contract?			or the Buil	of the Building Regulation 2006.			
	☐ Yes							
	☐ No							
Po	ol Safety Ins	pector: N/A		← The Pool : ———— Building A	← The Pool Safety Inspector must be licensed under the —— Building Act 1975 and Building Regulation 2006.			
Po	ol Safety Ins	pection Date:	← Clause 4.2 on a sale i	← Clause 4.2(2) applies except where this contract is formed on a sale by auction and some other excluded sales.				
СТ	ATUTODY	WADDANITIES AND	CONTRACTUAL DICUTS		-,			
			CONTRACTUAL RIGHTS	•				
The	e Seller gives	s notice to the Buyer of th	ie following matters:					
(a)	Latent or Pa	tent Defects in Common P	roperty or Body Corporate As	sets (section 223(a		TO 05//5		
	NONE			← WARNING TO SELLER: The Body Corporate and Community Management Act 1997 and the				
							es by the Seller about the	
(b)	Actual or Co	entingent or Expected Liabi	on 223(2)(c)(d))*	23(2)(c)(d))* Body Corporate and the Scheme land. Bre of a warranty may result in a damages clais termination by the Buyer. Seller should cor whether to carry out an inspection of the Bo		n a damages claim or		
` '	NONE	,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,					
				Corporate records to complete this section.				
(-)								
(c) Circumstances in Relation to Affairs of Body Corporate (section 223(3))* NONE								
	INONE							
(d)		o Warranties in clause 7.4						
	NONE							
(e)	Proposed Body Corporate Resolutions (clause 8.4)*							
	NONE							

*Include in attachment if insufficient space

ADDITIO	DNAL BODT CORPORATE INFO	RIVIATION				
Interest S	chedule Lot Entitlement of Lot:	Refer to "Body Corporate Discl	losure Statement"			
Aggregate	e Interest Schedule Lot Entitlement:	Refer to "Body Corporate Disci	losure Statement"			
Contribution	on Schedule Lot Entitlement of Lot:	Refer to "Body Corporate Discl	losure Statement"			
Aggregate	Contribution Schedule Lot Entitlement:	Refer to "Body Corporate Discl	osure Statement"			
INSURA	NCE POLICIES		 -			
Insurer:	See Body Corporate Disclosure Sta	atement	Policy No:			
Building:		-	1 0110) 1101			
-						
	ility:					-
Other:			 			
DOO! 0	AFETY FOR CUAREN BOOL C					
	AFETY FOR SHARED POOLS ete the following questions if there is a sha	ared pool on the Land				
	e a shared pool on the Land or on adj	•	with the Land?	⊘ Y	/es [□ No
B. If the a	nswer to A is Yes, is there a Complia me of contract?			✓ Y		☐ No
	nswer to B is No, has a Notice of No	Pool Safety Certificate been giv	en prior to contract?	Пү	′es [□ No
	is an obligation of the Seller under section	•				_
	_		unt he accomplated unless that	l at ia casa		
	ICAL SAFETY SWITCH AND SM		ust be completed unless the L WARNING: By giving fal.			
General P	r gives notice to the Buyer that an Ap Purpose Socket Outlets is: ct whichever is applicable) installed in the residence not installed in the residence	proved Salety Switch for the	information in this section a penalty. The Seller sho qualified advice about co and not rely on the Seller this section.	n, the Seller ould seek ex ompleting thi	r may in: kpert an: is sectio	d on
(selec	gives notice to the Buyer that a Comet whichever is applicable)	npliant Smoke Alarm(s) is/are:	← WARNING: Failure to ins Alarm is an offence unde Services Act 1990.			
\checkmark	installed in the residence					
LI NEIOUB	not installed in the residence	S ECHOES AND TOESS AS	T 2044			
	OURHOOD DISPUTES (DIVIDING	•				
	gives notice to the Buyer in accorda ghbourhood Disputes (Dividing Fence and:		 WARNING: Failure to converge to the Neighbourhood Disputes Trees Act) 2011 by givin application to the Buyer 	s (Dividing F ng a copy of	⊏ences a ∙an orde	and er or
·	ct whichever is applicable)		to Buyer signing the con to terminate the contract	tract will en	title the	Buyer
\checkmark	is not affected by any application to, Queensland Civil and Administrative to a tree on the Lot or	or an order made by, the Tribunal (QCAT) in relation	to terminate the contract	prior to Set	.aement	•
	is affected by an application to, or ar relation to a tree on the Lot, a copy of to the Buyer prior to the Buyer signin	of which has been given				



GST WITHHOLDI	NG OBLIGATIONS	
Is the Buyer register	red for GST and acquiring the Lot for a creditable purpose?	← WARNING: the Buyer warrants in
(select whicheve	er is applicable)	clause 2.5(6) that this information is true and correct.
Yes		
✓ No		
the Land by a bu	ole of an acquisition for a creditable purpose would be the purchase of uilding contractor, who is registered for GST, for the purposes of building and and selling it in the ordinary course of its business.]	
The Seller gives not Withholding Law that	ice to the Buyer in accordance with section 14-255(1)(a) of the at:	 WARNING: All sellers of residential premises or potential residential land are required to complete this notice.
(select whicheve	r is applicable)	Section 14-250 of the Withholding Law applies to the sale of 'new residential
	is not required to make a payment under section 14-250 of olding Law in relation to the supply of the Property	premises' or 'potential residential land' (subject to some exceptions) and requires an amount to be withheld from
Withholdir 14-255(1)	is required to make a payment under section 14-250 of the ng Law in relation to the supply of the Property. Under section of the Withholding Law, the Seller is required to give further or to settlement.	the Purchase Price and paid to the ATO. The Seller should seek legal advice if unsure about completing this section.
The REIQ Te	rms of Contract for Residential Lots in a Community Titles Sc Twelfth Edition Contain the Terms of this Contract.	heme (Pages 8-17)
SPECIAL COND	DITIONS	
1) In the event that t	the buyer or seller signs the contract and transmits it electronically via ending obligation by both parties with the force and effect as if it was an	email or facsimile this contract shall
SETTLEMENT		
	20 Days from the Contract date	←or the next Business Day if that is not
Settlement Date:	30 Days from the Contract date	a Business Day in the Place for Settlement.
Place for Settlement	Brisbane	← If Brisbane is inserted, this is a reference to Brisbane CBD.

SIGNATURES

The contract may be subject to a 5 business day statutory cooling-off period. A termination penalty of 0.25% of the purchase price applies if the Buyer terminates the contract during the statutory cooling-off period. It is recommended the Buyer obtain an independent property valuation and independent legal advice about the contract and his or her cooling-off rights, before signing.

Buyer: ×	Witness: *	HER
Buyer: By placing my signature above, I warrant that I am the Buyer named in the Reference Schedule or authorised by the Buyer to sign.	Witness: (Note: No witness is required if the Electronic Signature)	Buyer signs using an
Seller: X	Witness: **	SIGN
Seller: By placing my signature above, I warrant that I am the Seller named in the Reference Schedule or authorised by the Seller to sign.	Witness: (Note: No witness is required if the Electronic Signature)	: Seller signs using an
Deposit Holder:	← Who acknowledges having received the Init to hold that amount and any Balance Depos Deposit Holder for the parties as provided in	it when received as

TERMS OF CONTRACT FOR RESIDENTIAL LOTS IN A COMMUNITY TITLES SCHEME

1. DEFINITIONS

- 1.1 In this contract:
 - (1) terms in **bold** in the Reference Schedule and the Disclosure Statement have the meanings shown opposite them unless the context requires otherwise; and
 - (2) (a) "Approved Safety Switch" means a residual current device as defined in the Electrical Safety Regulation 2013;
 - (b) "ATO" means the Australian Taxation Office;
 - (c) "ATO Clearance Certificate" means a certificate issued under section 14-220(1) of the Withholding Law which is current on the date it is given to the Buyer;
 - (d) "Balance Purchase Price" means the Purchase Price, less the Deposit paid by the Buyer, adjusted under clause 2.6;
 - (e) "Bank" means an authorised deposit-taking institution within the meaning of the Banking Act 1959 (Cth).
 - (f) "Body Corporate" means the body corporate of the Scheme;
 - (g) "Body Corporate Debt" has the meaning in the Regulation Module but excludes the Body Corporate Levies for the period which includes the Settlement Date;
 - (h) "Body Corporate Levies" means regular periodic contributions levied on the owner of the Lot (including, if applicable, levied under an exclusive use by-law) excluding any Special Contribution:
 - "Bond" means a bond under the Residential Tenancies and Rooming Accommodation Act 2008;
 - "Building" means any building that forms part of the Lot or in which the Lot is situated;
 - (k) "Building Inspector" means a person licensed to carry out completed residential building inspections under the Queensland Building and Construction Commission Regulations 2003:
 - (I) "Business Day" means a day other than:
 - (i) a Saturday or Sunday
 - (ii) a public holiday in the Place for Settlement; and
 - (iii) a day in the period 27 to 31 December (inclusive).
 - (m) "CGT Withholding Amount" means the amount determined under section 14-200(3)(a) of the Withholding Law or, if a copy is provided to the Buyer prior to settlement, a lesser amount specified in a variation notice under section 14-235;
 - (n) "Compliance or Exemption Certificate" means:
 - (i) a Pool Safety Certificate; or
 - (ii) a building certificate that may be used instead of a Pool Safety Certificate under section 246AN(2) of the *Building Act 1975*; or
 - (iii) an exemption from compliance on the grounds of impracticality under section 245B of the *Building Act 1975*;
 - (o) "Compliant Smoke Alarm" means a smoke alarm complying with the requirements for smoke alarms in domestic dwellings under the Fire and Emergency Services Act 1990;
 - (p) "Contract Date" or "Date of Contract" means the date inserted in the Reference Schedule;
 - (q) "Court" includes any tribunal established under statute;

- (r) "Disclosure Statement" means the statement under section 206 (Existing Lot) or section 213 (Proposed Lot) of the Body Corporate and Community Management Act 1997;
- (s) "Electronic Signature" means an electronic method of signing that identifies the person and indicates their intention to sign the contract;
- (t) "Encumbrances" includes:
 - (i) unregistered encumbrances
 - (ii) statutory encumbrances; and
 - (iii) Security Interests.
- (u) "Essential Term" includes, in the case of breach by:
 - (i) the Buyer: clauses 2.2, 2.5(1), 2.5(5),5.1 and 6.1; and
 - (ii) the Seller: clauses 2.5(5), 5.1, 5.3(1)(a) (d), 5.3(1)(e)(ii) & (iii), 5.3(1)(f), 5.5 and 6.1; but nothing in this definition precludes a Court from finding other terms to be essential;
- (v) "Exclusive Use Areas" means parts of the common property for the Scheme allocated to the Lot under an exclusive use by-law;
- (w) "Financial Institution" means a Bank, Building Society or Credit Union;
- (x) "General Purpose Socket Outlet" means an electrical socket outlet as defined in the Electrical Safety Regulation 2013;
- (y) "GST" means the goods and services tax under the GST Act;
- (z) "GST Act" means A New Tax System (Goods and Services Tax) Act 1999 (Cth) and includes other GST related legislation;
- (aa) "GST Withholding Amount" means the amount (if any) determined under section 14-250 of the Withholding Law required to be paid to the Commissioner of Taxation;
- (bb) "Improvements" means fixed structures in the Lot (such as stoves, hot water systems, fixed carpets, curtains, blinds and their fittings, clothes lines, fixed satellite dishes and television antennae, in-ground plants) but does not include the Reserved Items;
- (cc) "Keys" means keys, codes or devices in the Seller's possession or control for all locks or security systems on the Property or necessary to access the Property;
- (dd) "Land" means the scheme land for the Scheme;
- (ee) "Notice of No Pool Safety Certificate" means the Form 36 under the *Building Regulation 2006* to the effect that there is no Pool Safety Certificate issued for the Land and/or the Lot;
- (ff) "Notice of nonconformity" means a Form 26 under the Building Regulation 2006 advising how the pool does not comply with the relevant pool safety standard;
- (gg) "Outgoings" means:
 - (i) rates or charges on the Lot by any competent authority (for example, council rates, water rates, fire service levies) but excludes land tax; and
 - (ii) Body Corporate Levies.
- (hh) "Pest Inspector" means a person licensed to undertake termite inspections on completed buildings under the Queensland Building and Construction Commission Regulations 2003;
- (ii) "Pool Safety Certificate" has the meaning in section 231C(a) of the Building Act 1975;

- (jj) "Pool Safety Requirements" means the requirements for pool safety contained in the Building Act 1975 and Building Regulation 2006;
- (kk) "Pool Safety Inspection Date" means the Pool Safety Inspection Date inserted in the Reference Schedule. If no date is inserted in the Reference Schedule, the Pool Safety Inspection Date is taken to be the earlier of the following:
 - (i) the Inspection Date for the Building and/or Pest Inspection; or
 - (ii) 2 Business Days before the Settlement Date;
- (II) "Pool Safety Inspector" means a person authorised to give a Pool Safety Certificate;
- (mm) "PPSR" means the Personal Property Securities Register established under Personal Property Securities Act 2009 (Cth);
- (nn) "Property" means:
 - (i) the Lot;
 - (ii) the right to any Exclusive Use Areas;
 - (iii) the Improvements;
 - (iv) the Included Chattels;
- (oo) "Regulation Module" means the regulation module for the Scheme;
- (pp) "Rent" means any periodic amount, including outgoings, payable under the Tenancies;
- (qq) "Reserved Items" means the Excluded Fixtures and all Chattels in the Lot and Exclusive Use Areas other than the Included Chattels;
- (rr) "Scheme" means the community titles scheme containing the Lot;
- (ss) "Security Interests" means all security interests registered on the PPSR over Included Chattels and Improvements;
- (tt) "Special Contribution" means an amount levied by the Body Corporate under the Regulation Module for a liability for which no provision or inadequate provision has been made in the budget of the Body Corporate.
- (uu) "Transfer Documents" means:
 - (i) the form of transfer under the Land Title Act 1994 required to transfer title in the Lot to the Buyer; and
 - (ii) any other document to be signed by the Seller necessary for stamping or registering the transfer;
- (vv) "Transport Infrastructure" has the meaning defined in the *Transport Infrastructure Act 1994*; and
- (ww) "Withholding Law" means Schedule 1 to the Taxation Administration Act 1953 (Cth).
- **1.2** Words and phrases defined in the *Body Corporate and Community Management Act 1997* have the same meaning in this contract unless the context indicates otherwise.

2. PURCHASE PRICE

2.1 GST

- (1) Unless otherwise specified in this contract, the Purchase Price includes any GST payable on the supply of the Property to the Buyer.
- (2) If a party is required to make any other payment or reimbursement under this contract, that payment or reimbursement will be reduced by the amount of any input tax credits to which the other party (or the representative member for a GST group of which it is a member) is entitled.

2.2 Deposit

(1) The Buyer must pay the Deposit to the Deposit Holder at the times shown in the Reference Schedule. The Deposit Holder will hold the Deposit until a party becomes entitled to it.

- (2) The Buyer will be in default if it:
 - (a) does not pay the Deposit when required;
 - (b) pays the Deposit by a post-dated cheque; or
 - (c) pays the Deposit by cheque which is dishonoured on presentation.
- (3) The Seller may recover from the Buyer as a liquidated debt any part of the Deposit which is not paid when required.

2.3 Investment of Deposit

lf:

- the Deposit Holder is instructed by either the Seller or the Buyer; and
- (2) it is lawful to do so;

the Deposit Holder must:

- (3) invest as much of the Deposit as has been paid with any Financial Institution in an interest-bearing account in the names of the parties; and
- (4) provide the parties' tax file numbers to the Financial Institution (if they have been supplied).

2.4 Entitlement to Deposit and Interest

- (1) The party entitled to receive the Deposit is:
 - (a) if this contract settles, the Seller;
 - (b) if this contract is terminated without default by the Buyer, the Buyer;
 - (c) if this contract is terminated owing to the Buyer's default, the Seller.
- (2) The interest on the Deposit must be paid to the person who is entitled to the Deposit.
- (3) If this contract is terminated, the Buyer has no further claim once it receives the Deposit and interest unless the termination is due to the Seller's default or breach of warranty.
- (4) The Deposit is invested at the risk of the party who is ultimately entitled to it.

2.5 Payment of Balance Purchase Price

- (1) On the Settlement Date, the Buyer must pay the Balance Purchase Price by Bank cheque as the Seller or the Seller's Solicitor directs.
- (2) Despite any other provision of this contract, reference to a "Bank cheque" in clause 2.5:
 - (a) includes a cheque drawn by a Building Society or Credit Union on itself;
 - (b) does not include a cheque drawn by a Building Society or Credit Union on a Bank;
 - and the Seller is not obliged to accept a cheque referred to in clause 2.5(2)(b) on the Settlement Date.
- (3) If both the following apply:
 - (a) the sale is not an excluded transaction under section 14-215 of the Withholding Law; and
 - (b) the Seller has not given the Buyer on or before settlement for each person comprising the Seller either:
 - (i) an ATO Clearance Certificate; or
 - (ii) a variation notice under section 14-235 of the Withholding Law which remains current at the Settlement Date varying the CGT Withholding Amount to nil,

then:

(c) for clause 2.5(1), the Seller irrevocably directs the Buyer to draw a bank cheque for the CGT Withholding Amount in favour of the Commissioner of Taxation or, if the Buyer's Solicitor requests, the Buyer's Solicitor's Trust Account;

- (d) the Buyer must lodge a Foreign Resident Capital Gains Withholding Purchaser Notification Form with the ATO for each person comprising the Buyer and give copies to the Seller with the payment reference numbers (PRN) on or before settlement;
- (e) the Seller must return the bank cheque in paragraph (c) to the Buyer's Solicitor (or if there is no Buyer's Solicitor, the Buyer) at settlement; and
- (f) the Buyer must pay the CGT Withholding Amount to the ATO in accordance with section 14-200 of the Withholding Law and give the Seller evidence that it has done so within 2 Business Days of settlement occurring.
- (4) For clause 2.5(3) and section 14-215 of the Withholding Law, the market value of the CGT asset is taken to be the Purchase Price less any GST included in the Purchase Price for which the Buyer is entitled to an input tax credit unless:
 - (a) the Property includes items in addition to the Lot and Improvements; and
 - (b) no later than 2 Business Days prior to the Settlement Date, the Seller gives the Buyer a valuation of the Lot and Improvements prepared by a registered valuer,

in which case the market value of the Lot and Improvements will be as stated in the valuation.

- (5) If the Buyer is required to pay the GST Withholding Amount to the Commissioner of Taxation at settlement pursuant to section 14-250 of the Withholding Law:
 - (a) the Seller must give the Buyer a notice in accordance with section 14-255(1) of the Withholding Law;
 - (b) prior to settlement the Buyer must lodge with the ATO:
 - a GST Property Settlement Withholding Notification form ("Form 1"); and
 - (ii) a GST Property Settlement Date Confirmation form ("Form 2");
 - (c) on or before settlement, the Buyer must give the Seller copies of:
 - (i) the Form 1;
 - (ii) confirmation from the ATO that the Form 1 has been lodged specifying the Buyer's lodgement reference number and payment reference number;
 - (iii) confirmation from the ATO that the Form 2 has been lodged; and
 - (iv) a completed ATO payment slip for the Withholding Amount.
 - (d) the Seller irrevocably directs the Buyer to draw a bank cheque for the GST Withholding Amount in favour of the Commissioner of Taxation and deliver it to the Seller at settlement; and
 - (e) the Seller must pay the GST Withholding Amount to the ATO in compliance with section 14-250 of the Withholding Law promptly after settlement.
- (6) The Buyer warrants that the statements made by the Buyer in the Reference Schedule under GST Withholding Obligations are true and correct.

2.6 Adjustments to Balance Purchase Price

(1) The Seller is liable for Outgoings and is entitled to Rent up to and including the Settlement Date. The Buyer is liable for Outgoings and is entitled to Rent after the Settlement Date.

- (2) Subject to clauses 2.6(3), 2.6(5) and 2.6(17), Outgoings for periods including the Settlement Date must be adjusted:
 - (a) for those paid, on the amount paid;
 - (b) for those assessed but unpaid, on the amount payable (excluding any discount); and
 - (c) for those not assessed:
 - on the amount the relevant authority or the Body Corporate advises will be assessed (excluding any discount); or
 - (ii) if no advice on the assessment to be made is available, on the amount of the latest assessment (excluding any discount).
- (3) If there is no separate assessment of rates for the Lot at the Settlement Date and the Local Government informs the Buyer that it will not apportion rates between the Buyer and the Seller, then:
 - (a) the amount of rates to be adjusted is that proportion of the assessment equal to the ratio of the interest schedule lot entitlement of the Lot to the aggregate interest schedule lot entitlement of the Scheme; and
 - (b) if an assessment of rates includes charges imposed on a "per lot" basis, then the portion of those charges to be adjusted is the amount assessed divided by the number of lots in that assessment.
- (4) The Seller is liable for land tax assessed on the Lot for the financial year current at the Settlement Date. If land tax is unpaid at the Settlement Date and the Office of State Revenue advises that it will issue a final clearance for the Lot on payment of a specified amount, then the Buyer may deduct the specified amount from the Balance Purchase Price at settlement and must pay it promptly to the Office of State Revenue.
- (5) Any Outgoings assessable on the amount of water used must be adjusted on the charges that would be assessed on the total water usage for the assessment period, determined by assuming that the actual rate of usage shown by the meter reading made before settlement continues throughout the assessment period. The Buyer must obtain and pay for the meter reading.
- (6) If any Outgoings are assessed but unpaid at the Settlement Date, then the Buyer may deduct the amount payable from the Balance Purchase Price at settlement and pay it promptly to the relevant authority or the Body Corporate, as appropriate. If an amount is deducted under this clause, the relevant Outgoing will be treated as paid at the Settlement Date for the purposes of clause 2.6(2).
- (7) Arrears of Rent for any rental period ending on or before the Settlement Date belong to the Seller and are not adjusted at settlement.
- (8) Unpaid Rent for the rental period including both the Settlement Date and the following day ("Current Period") is not adjusted until it is paid.
- (9) Rent already paid for the Current Period or beyond must be adjusted at settlement.
- (10) If Rent payments are reassessed after the Settlement Date for periods including the Settlement Date, any additional Rent payment from a Tenant or refund due to a Tenant must be apportioned under clauses 2.6(7), 2.6(8) and 2.6(9) and 2.6(10).
- (11) Payments under clause 2.6(10) must be made within 14 days after notification by one party to the other but only after any additional payment from a Tenant has been received.

- (12) The Seller is liable for:
 - (a) any Special Contribution for which a levy notice has been issued on or before the Contract Date; and
 - (b) any other Body Corporate Debt (including any penalty or recovery cost resulting from nonpayment of a Body Corporate Debt) owing in respect of the Lot at settlement.

The Buyer is liable for any Special Contribution levied after the Contract Date.

- (13) If an amount payable by the Seller under clause 2.6(12) is unpaid at the Settlement Date, the Buyer may deduct the specified amount from the Balance Purchase Price at settlement and must pay it promptly to the Body Corporate.
- (14) For the purposes of clause 2.6(12), an amount payable under an exclusive use by-law will be treated as levied on the date it is due.
- (15) The cost of Bank cheques payable at settlement:
 - (a) to the Seller or its mortgagee are the responsibility of the Buyer; and
 - (b) to parties other than the Seller or its mortgagee are the responsibility of the Seller.
- (16) The Seller is not entitled to require payment of the Balance Purchase Price by means other than Bank cheque without the consent of the Buyer.
- (17) Upon written request by the Buyer, the Seller will, prior to settlement, give the Buyer a written statement, supported by reasonable evidence, of –
 - (a) all Outgoings and all Rent for the Property to the extent they are not capable of discovery by search or enquiry at any office of public record or pursuant to the provisions of any statute; and
 - (b) any other information which the Buyer may reasonably require for the purpose of calculating or apportioning any Outgoings or Rent under this clause 2.6.

If the Seller becomes aware of a change to the information provided the Seller will as soon as practicably provide the updated information to the Buyer.

3. FINANCE

- **3.1** This contract is conditional on the Buyer obtaining approval of a loan for the Finance Amount from the Financier by the Finance Date on terms satisfactory to the Buyer. The Buyer must take all reasonable steps to obtain approval.
- 3.2 The Buyer must give notice to the Seller that:
 - approval has not been obtained by the Finance Date and the Buyer terminates this contract; or
 - the finance condition has been either satisfied or waived by the Buyer.
- 3.3 The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 3.2 by 5pm on the Finance Date. This is the Seller's only remedy for the Buyer's failure to give notice.
- 3.4 The Seller's right under clause 3.3 is subject to the Buyer's continuing right to give written notice to the Seller of satisfaction, termination or waiver pursuant to clause 3.2.

4. BUILDING AND PEST INSPECTION REPORTS AND POOL SAFETY

4.1 Building and Pest Inspection

(1) This contract is conditional on the Buyer obtaining a written building report from a Building Inspector and a written pest report from a Pest Inspector (which may be a single report) on the Property by the Inspection Date on terms satisfactory to the Buyer. The Buyer must take all reasonable steps to obtain the reports (subject to the right of the Buyer to elect to obtain only one of the reports).

- (2) The Buyer must give notice to the Seller that:
 - (a) a satisfactory Inspector's report under clause 4.1(1) has not been obtained by the Inspection Date and the Buyer terminates this contract. The Buyer must act reasonably; or
 - (b) clause 4.1(1) has been either satisfied or waived by the Buyer.
- (3) If the Buyer terminates this contract and the Seller asks the Buyer for a copy of the building and pest reports, the Buyer must give a copy of each report to the Seller without delay.
- (4) The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 4.1(2) by 5pm on the Inspection Date. This is the Seller's only remedy for the Buyer's failure to give notice.
- (5) The Seller's right under clause 4.1(4) is subject to the Buyer's continuing right to give written notice to the Seller of satisfaction, termination or waiver pursuant to clause 4.1(2).

4.2 Pool Safety for non-shared pool on Lot

- (1) This clause 4.2 applies if:
 - (a) there is a pool on the Lot and the answer to Q2 of the Reference Schedule is No or Q2 is not completed (for a non-shared pool on the Lot); and
 - (b) this contract is not a contract of a type referred to in section 160(1)(b) of the *Property* Occupations Act 2014.
- (2) This contract is conditional upon:
 - (a) the issue of a Pool Safety Certificate; or
 - (b) a Pool Safety Inspector issuing a Notice of Nonconformity stating the works required before a Pool Safety Certificate can be issued;

by the Pool Safety Inspection Date.

- (3) The Buyer is responsible for arranging an inspection by a Pool Safety Inspector at the Buyer's cost. The Seller authorises:
 - (a) the Buyer to arrange the inspection; and
 - (b) the Pool Safety Inspector to advise the Buyer of the results of the inspection and to give the Buyer a copy of any notice issued.
- (4) If a Pool Safety Certificate has not been issued by the Pool Safety Inspection Date, the Buyer may give notice to the Seller that the Buyer:
 - (a) terminates this contract; or
 - (b) waives the benefit of this clause 4.2

The Buyer must act reasonably.

- (5) The Seller may terminate this contract by notice to the Buyer if notice is not given under clause 4.2(4) by 5pm on the Pool Safety Inspection Date.
- (6) The Seller's right under clause 4.2(5) is subject to the Buyer's continuing right to give written notice to the Seller of termination or waiver pursuant to clause 4.2(4).
- (7) The right of a party to terminate under this clause 4.2, ceases upon receipt by that party of a copy of a current Pool Safety Certificate.
- (8) If the Buyer terminates this contract under clause 4.2(4)(a), and the Seller has not obtained a copy of the Notice of Nonconformity issued by the Pool Safety Inspector, the Seller may request a copy and the Buyer must provide this to the Seller without delay.

5. SETTLEMENT

5.1 Time and Date

 Settlement must occur between 9am and 4pm AEST on the Settlement Date.

INITIALS (Note: initials not required if signed with Electronic Signature)

(2) If the parties do not agree on where settlement is to occur, it must take place in the Place for Settlement at the office of a Solicitor or Financial Institution nominated by the Seller, or, if the Seller does not make a nomination, at the Land Registry Office in or nearest to the Place for Settlement.

5.2 Transfer Documents

- (1) The Transfer Documents must be prepared by the Buyer's Solicitor and delivered to the Seller a reasonable time before the Settlement Date.
- (2) If the Buyer pays the Seller's reasonable expenses, it may require the Seller to produce the Transfer Documents at the Office of State Revenue nearest the Place for Settlement for stamping before settlement

5.3 Documents and Keys at Settlement

- In exchange for payment of the Balance Purchase Price, the Seller must deliver to the Buyer at settlement:
 - (a) any instrument of title for the Lot required to register the transfer to the Buyer; and
 - (b) unstamped Transfer Documents capable of immediate registration after stamping; and
 - (c) any instrument necessary to release any Encumbrance over the Property in compliance with the Seller's obligation in clause 7.2; and
 - (d) if requested by the Buyer not less than 2 clear Business Days before the Settlement Date, the Keys; and
 - (e) if there are Tenancies:
 - (i) the Seller's copy of any Tenancy agreements;
 - (ii) a notice to each tenant advising of the sale in the form required by law; and
 - (iii) any notice required by law to transfer to the Buyer the Seller's interest in any Bond; and
 - (f) if the answer to Q2 in the Reference Schedule is Yes, a copy of a current Compliance or Exemption Certificate, if not already provided to the Buyer.
- (2) If the Keys are not delivered at settlement under clause 5.3(1)(d), the Seller must deliver the Keys to the Buyer. The Seller may discharge its obligation under this provision by authorising the Seller's Agent to release the Keys to the Buyer.

5.4 Assignment of Covenants and Warranties

At settlement, the Seller assigns to the Buyer the benefit of all:

- (1) covenants by the tenants under the Tenancies;
- (2) guarantees and Bonds (subject to the requirements of the Residential Tenancies and Rooming Accommodation Act 2008) supporting the Tenancies; and
- (3) manufacturers' warranties regarding the Included Chattels;
- (4) builders' warranties on the Improvements; to the extent that they are assignable and the Buyer accepts the assignment. However, the right to recover arrears of Rent is not assigned to the Buyer and section 117 of the *Property Law Act 1974* does not apply.

5.5 Possession of Property and Title to Included Chattels

On the Settlement Date, in exchange for the Balance Purchase Price, the Seller must give the Buyer vacant possession of the Lot and Exclusive Use Areas except for the Tenancies. Title to the Included Chattels passes at settlement.

5.6 Reservations

 The Seller must remove the Reserved Items from the Property before settlement.

- (2) The Seller must repair at its expense any damage, done to the Property in removing the Reserved Items. If the Seller fails to do so, the Buyer may repair that damage.
- (3) Any Reserved Items not removed before settlement will be considered abandoned and the Buyer may, without limiting its other rights, complete this contract and appropriate those Reserved Items or dispose of them in any way.
- (4) The Seller indemnifies the Buyer against any damages and expenses resulting from the Buyer's actions under clauses 5.6(2) or 5.6(3).

6. TIME

6.1 Time of the Essence

Time is of the essence of this contract, except regarding any agreement between the parties on a time of day for settlement.

6.2 Suspension of Time

- (1) This clause 6.2 applies if a party is unable to perform a settlement obligation solely as a consequence of a Delay Event but does not apply where the inability is attributable to: —
 - (a) damage to, destruction of or diminution in value of the Property or other property of the Seller or Buyer; or
 - (b) termination or variation of any agreement between a party and another person whether relating to the provision of finance, the release of an Encumbrance, the sale or purchase of another property or otherwise.
- (2) Time for the performance of the parties' Settlement Obligations is suspended and ceases to be of the essence of the contract and the parties are deemed not to be in breach of their Settlement Obligations.
- (3) An Affected Party must take reasonable steps to minimise the effect of the Delay Event on its ability to perform its Settlement Obligations.
- (4) When an Affected Party is no longer prevented from performing its Settlement Obligations due to the Delay Event, the Affected Party must give the other party a notice of that fact, promptly.
- (5) When the Suspension Period ends, whether notice under clause 6.2(4) has been given or not, either party may give the other party a Notice to Settle.
- (6) A Notice to Settle must be in writing and state:
 - (a) that the Suspension Period has ended; and
 - a date, being not less than 5 nor more than 10 Business Days after the date the Notice to Settle is given, which shall become the Settlement Date;
 - (c) that time is of the essence.
- (7) When Notice to Settle is given, time is again of the essence of the contract.
- (8) In this clause 6.2:
 - (a) "Affected Party" means a party referred to in clause 6.2(1);
 - (b) "Delay Event" means:
 - a tsunami, flood, cyclone, earthquake, bushfire or other act of nature;
 - (ii) riot, civil commotion, war, invasion or a terrorist act;
 - (iii) an imminent threat of an event in paragraphs (i) or (ii); or
 - (iv) compliance with any lawful direction or order by a Government Agency;

- (c) "Government Agency" means the government of the Commonwealth of Australia or an Australian State, Territory or local government and includes their authorities, agencies, government owned corporations and authorised officers, courts and tribunals;
- (d) "Settlement Obligations" means, in the case of the Buyer, its obligations under clauses 2.5(1) and 5.1(1) and, in the case of the Seller, its obligations under clauses 5.1(1), 5.3(1)(a) – (e) and 5.5;
- (e) "Suspension Period" means the period during which the Affected Party (or if both the Buyer and Seller are Affected Parties, either of them) remains unable to perform a settlement obligation solely as a consequence of a Delay Event.

7. MATTERS AFFECTING THE PROPERTY

7.1 Title

The Lot is sold subject to:

- (1) the Body Corporate and Community Management Act 1997 and the by-Laws of the Body Corporate;
- any reservations or conditions on the title or the original Deed of Grant.

7.2 Encumbrances

The Property is sold free of all Encumbrances other than the Title Encumbrances, Tenancies, statutory easements implied by part 6A of the *Land Title Act 1994* and interests registered on the common property for the Scheme.

7.3 Requisitions

The Buyer may not deliver any requisitions or enquiries on title.

7.4 Seller's Warranties

- (1) The Seller warrants that, except as disclosed in this contract, at settlement:
 - it will be the registered owner of an estate in fee simple in the Lot and will own the Improvements and Included Chattels;
 - (b) it will be capable of completing this contract (unless the Seller dies or becomes mentally incapable after the Contract Date); and
 - (c) there will be no unsatisfied judgment, order (except for an order referred to in clause 7.6(1)(b)) or writ affecting the Property.
- (2) The Seller warrants that, except as disclosed in this contract, at the Contract Date and at settlement there are no current or threatened claims, notices or proceedings that may lead to a judgment, order or writ affecting the Property.
- (3) The Seller warrants that, except as disclosed in this contract, at the Contract Date:
 - (a) there is no unregistered lease, easement or other right capable of registration and which is required to be registered to give indefeasibility affecting the common property or Body Corporate assets;
 - (b) there is no proposal to record a new community management statement for the Scheme and it has not received a notice of a meeting of the Body Corporate to be held after the Contract Date or notice of any proposed resolution or a decision of the Body Corporate to consent to the recording of a new community management statement for the Scheme;
 - (c) all Body Corporate consents to improvements made to common property and which benefit the Lot, or the registered owner of the Lot, are in force; and

- (d) the Additional Body Corporate Information is correct (if completed).
- (4) If the Seller breaches a warranty in clause 7.4(1) or clause 7.4(2), the Buyer may terminate this contract by notice to the Seller.
- (5) If:
 - (a) the Seller breaches a warranty in clause 7.4(3);
 - (b) the Additional Body Corporate Information is not completed;

and, as a result, the Buyer is materially prejudiced, the Buyer may terminate this contract by notice to the Seller given within 14 days after the Contract Date but may not claim damages or compensation.

- (6) Clauses 7.4(4) and 7.4(5) do not restrict any statutory rights the Buyer may have which cannot be excluded by this contract.
- (7) (a) The Seller warrants that, except as disclosed in this contract or a notice given by the Seller to the Buyer under the Environmental Protection Act 1994 ("EPA"), at the Contract Date:
 - there is no outstanding obligation on the Seller to give notice to the administering authority under EPA of notifiable activity being conducted on the Land; and
 - (ii) the Seller is not aware of any facts or circumstances that may lead to the Land being classified as contaminated land within the meaning of EPA.
 - (b) If the Seller breaches a warranty in clause 7.4(7), the Buyer may:
 - terminate this contract by notice in writing to the Seller given no later than 2 Business Days before the Settlement Date; or
 - (ii) complete this contract and claim compensation, but only if the Buyer claims it in writing before the Settlement Date.
- (8) The Seller does not warrant that the Present Use is lawful.

7.5 Survey and Mistake

- (1) The Buyer may survey the Lot.
- (2) If there is:
 - (a) an error in the boundaries or area of the Lot;
 - (b) an encroachment by structures onto or from the Lot; or
 - (c) a mistake or omission in describing the Lot or the Seller's title to it;

which is:

- (d) immaterial; or
- (e) material, but the Buyer elects to complete this contract;

the Buyer's only remedy against the Seller is for compensation, but only if claimed by the Buyer in writing on or before settlement.

- (3) The Buyer may not delay settlement or withhold any part of the Balance Purchase Price because of any compensation claim under clause 7.5(2).
- (4) If there is a material error, encroachment or mistake, the Buyer may terminate this contract before settlement.

7.6 Requirements of Authorities

- (1) Subject to clause 7.6(5), any valid notice or order by any competent authority or Court requiring work to be done or money spent in relation to the Property ("Work or Expenditure") must be fully complied with:
 - (a) if issued before the Contract Date, by the Seller before the Settlement Date;

- (b) if issued on or after the Contract Date, by the Buyer.
- (2) If any Work or Expenditure that is the Seller's responsibility under clause 7.6(1)(a) is not done before the Settlement Date, the Buyer is entitled to claim the reasonable cost of work done by the Buyer in accordance with the notice or order referred to in clause 7.6(1) from the Seller after settlement as a debt
- (3) Any Work or Expenditure that is the Buyer's responsibility under clause 7.6(1)(b), which is required to be done before the Settlement Date, must be done by the Seller unless the Buyer directs the Seller not to and indemnifies the Seller against any liability for not carrying out the work. If the Seller does the work, or spends the money, the reasonable cost of that Work or Expenditure must be added to the Balance Purchase Price.
- (4) The Buyer may terminate this contract by notice to the Seller if there is an outstanding notice at the Contract Date under section 246AG of the Building Act 1975 that affects the Property. The Buyer may terminate this contract by notice to the Seller if there is an outstanding notice at the Contract Date under sections 247 or 248 of the Building Act 1975 or sections 167 or 168 of the Planning Act 2016 that affects the Property or Land.
- (5) Clause 7.6(1) does not apply to orders disclosed under section 83 of the Neighbourhood Disputes (Dividing Fences and Trees) Act 2011.

7.7 Property Adversely Affected

- (1) If at the Contract Date:
 - (a) the Present Use is not lawful under the relevant town planning scheme;
 - (b) the Land is affected by a proposal of any competent authority to alter the dimensions of any Transport Infrastructure or locate Transport Infrastructure on the Land;
 - (c) access or any service to the Land passes unlawfully through other land;
 - (d) any competent authority has issued a current notice to treat, or notice of intention to resume, regarding any part of the Land;
 - (e) there is an outstanding condition of a development approval attaching to the Land under section 73 of the Planning Act 2016 or section 96 of the Economic Development Queensland Act 2012 which, if complied with, would constitute a material mistake or omission in the Seller's title under clause 7.5(2)(c);
 - (f) the Property is affected by the *Queensland*Heritage Act 1992 or is included in the World
 Heritage List;
 - (g) the Property is declared acquisition land under the Queensland Reconstruction Authority Act 2011; or
 - (h) there is a charge against the Lot under section 104 of the Foreign Acquisitions and Takeovers Act 1975 (Cth),
 - and that has not been disclosed in this contract, the Buyer may terminate this contract by notice to the Seller given on or before settlement.
- (2) If no notice is given under clause 7.7(1), the Buyer will be treated as having accepted the Property subject to all of the matters referred to in that clause.
- (3) The Seller authorises the Buyer to:
 - inspect records held by any authority, including Security Interests on the PPSR, relating to the Property or the Land; and
 - (b) apply for a certificate of currency of the Body Corporate's insurance from any insurer.

7.8 Dividing Fences

Notwithstanding any provision in the *Neighbourhood Disputes (Dividing Fences and Trees) Act 2011*, the Seller need not contribute to the cost of building any dividing fence between the Lot and any adjoining land owned by it. The Buyer waives any right to claim contribution from the Seller.

8. RIGHTS AND OBLIGATIONS UNTIL SETTLEMENT

8.1 Risk

The Property is at the Buyer's risk from 5pm on the first Business Day after the Contract Date.

8.2 Access

After reasonable notice to the Seller, the Buyer and its consultants may enter the Property:

- (1) once to read any meter;
- (2) for inspections under clause 4;
- (3) once to inspect the Property before settlement; and
- (4) once to value the Property before settlement.

8.3 Seller's Obligations After Contract Date

- (1) The Seller must use the Property reasonably until settlement. The Seller must not do anything regarding the Property or Tenancies that may significantly alter them or result in later expense for the Buyer.
- (2) The Seller must promptly upon receiving any notice, proceeding or order that affects the Property or requires work on the Property, give a copy to the Buyer.
- (3) Without limiting clause 8.3(1), the Seller must not without the prior written consent of the Buyer, give any notice or seek or consent to any order that affects the Property or make any agreement affecting the Property that binds the Buyer to perform.

8.4 Body Corporate Meetings

- (1) The Seller must promptly give the Buyer a copy of:
 - (a) any notice it receives of a proposed meeting of the Body Corporate to be held after the Contract Date; and
 - (b) resolutions passed at that meeting and prior to settlement.
- (2) The Buyer may terminate this contract by notice in writing to the Seller given before settlement if it is materially prejudiced by:
 - (a) any resolution of the Body Corporate passed after the Contract Date, other than a resolution, details of which are disclosed to the Buyer in this contract; or
 - (b) where the Scheme is a subsidiary scheme, any resolution of a Body Corporate of a higher scheme.
- (3) In clause 8.4(2) a resolution includes a decision of the Body Corporate Committee to consent to recording a new community management statement.
- (4) If the Buyer is not given a copy of the resolutions before settlement, it may sue the Seller for damages.

8.5 Information Regarding the Property

Upon written request of the Buyer but in any event before settlement, the Seller must give the Buyer:

- copies of all documents relating to any unregistered interests in the Property;
- (2) full details of the Tenancies to allow the Buyer to properly manage the Property after settlement;
- (3) sufficient details (including the date of birth of each Seller who is an individual) to enable the Buyer to undertake a search of the PPSR;

8.6 Possession Before Settlement

If possession is given before settlement:

- the Buyer must maintain the Property in substantially its condition at the date of possession, fair wear and tear excepted;
- (2) entry into possession is under a licence personal to the Buyer revocable at any time and does not:
 - (a) create a relationship of landlord and tenant; or
 - (b) waive the Buyer's rights under this contract;
- (3) the Buyer must insure the Property to the Seller's satisfaction; and
- (4) the Buyer indemnifies the Seller against any expense or damages incurred by the Seller as a result of the Buyer's possession of the Property.

8.7 Seller's Obligations After Contract Date

- the Seller must promptly upon receiving any notice, proceeding or order that affects the Property or requires work on the Property, give a copy to the Buyer.
- (2) After the Contract Date, the Seller must not without the prior written consent of the Buyer, give any notice, seek or consent to any order or make an agreement that affects the Property.

9. PARTIES' DEFAULT

9.1 Seller and Buyer May Affirm or Terminate

Without limiting any other right or remedy of the parties including those under this contract, or any right at law or in equity, if the Seller or Buyer, as the case may be, fails to comply with an Essential Term, or makes a fundamental breach of an intermediate term, the Seller (in the case of the Buyer's default) or the Buyer (in the case of the Seller's default) may affirm or terminate this contract.

9.2 If Seller Affirms

If the Seller affirms this contract under clause 9.1, it may sue the Buyer for:

- (1) damages;
- (2) specific performance; or
- (3) damages and specific performance.

9.3 If Buyer Affirms

If the Buyer affirms this contract under clause 9.1, it may sue the Seller for:

- (1) damages;
- (2) specific performance; or
- (3) damages and specific performance.

9.4 If Seller Terminates

If the Seller terminates this contract under clause 9.1, it may do all or any of the following:

- (1) resume possession of the Property:
- (2) forfeit the Deposit and any interest earned;
- (3) sue the Buyer for damages;
- (4) resell the Property.

9.5 If Buyer Terminates

If the Buyer terminates this contract under clause 9.1, it may do all or any of the following:

- (1) recover the Deposit and any interest earned;
- (2) sue the Seller for damages.

9.6 Seller's Resale

- (1) If the Seller terminates this contract and resells the Property, the Seller may recover from the Buyer as liquidated damages:
 - (a) any deficiency in price on a resale; and
 - (b) its expenses connected with any repossession, any failed attempt to resell, and the resale,

provided the resale settles within 2 years of termination of this contract.

(2) Any profit on a resale belongs to the Seller.

9.7 Seller's Damages

The Seller may claim damages for any loss it suffers as a result of the Buyer's default, including its legal costs on an indemnity basis and the cost of any Work or Expenditure under clause 7.6(3).

9.8 Buyer's Damages

The Buyer may claim damages for any loss it suffers as a result of the Seller's default, including its legal costs on an indemnity basis.

9.9 Interest on Late Payments

- (1) The Buyer must pay interest at the Default Rate:
 - (a) on any amount payable under this contract which is not paid when due; and
 - (b) on any judgement for money payable under this contract.
- (2) Interest continues to accrue:
 - (a) under clause 9.9(1)(a), from the date it is due until paid; and
 - (b) under clause 9.9(1)(b), from the date of judgement until paid.
- (3) Any amount payable under clause 9.9(1)(a) in respect of a period prior to settlement must be paid by the Buyer at settlement. If this contract is terminated or if any amount remains unpaid after settlement, interest continues to accrue.
- (4) Nothing in this clause affects any other rights of the Seller under this contract or at law.

10. GENERAL

10.1 Seller's Agent

The Seller's Agent is appointed as the Seller's agent to introduce a Buyer.

10.2 Foreign Buyer Approval

The Buyer warrants that either:

- (1) the Buyer's purchase of the Property is not a notifiable action; or
- (2) the Buyer has received a no objection notification, under the *Foreign Acquisitions and Takeovers Act 1975* (Cth).

10.3 Duty

The Buyer must pay all duty on this contract.

10.4 Notices

- (1) Notices under this contract must be in writing.
- (2) Notices under this contract or notices required to be given by law may be given and received by the party's solicitor.
- (3) Notices under this contract or notices required to be given by law may be given by:
 - (a) delivering or posting to the other party or its solicitor; or
 - (b) sending it to the facsimile number of the other party or its solicitor stated in the Reference Schedule (or another facsimile number notified by the recipient to the sender); or
 - (c) sending it to the email address of the other party or its solicitor stated in the Reference Schedule (or another email address notified by the recipient to the sender).
- (4) Subject to clause 10.4(5), a notice given after this contract is entered into in accordance with clause 10.4(3) will be treated as given:
 - (a) 5 Business Days after posting;
 - (b) if sent by facsimile, at the time indicated on a clear transmission report; and
 - (c) if sent by email, at the time it is sent.



- (5) Notices given by facsimile, by personal delivery or by 10.10 Counterparts email between 5pm on a Business Day (the "first Business Day") and 9am on the next Business Day (the "second Business Day") will be treated as given or delivered at 9am on the second Business Day.
- (6) If two or more notices are treated as given at the same time under clause 10.4(5), they will be treated as given in the order in which they were sent or delivered.
- (7) Notices or other written communications by a party's solicitor (for example, varying the Inspection Date, Finance Date or Settlement Date) will be treated as given with that party's authority.
- (8) For the purposes of clause 10.4(3)(c) and clause 12.2 the notice or information may be contained within an email, as an attachment to an email or located in an electronic repository accessible by the recipient by clicking a link in an email.

10,5 Business Days

- (1) If anything is required to be done on a day that is not a Business Day, it must be done instead on the next Business Day.
- If the Finance Date or Inspection Date fall on a day that is not a Business Day, then it falls on the next Business Day.

10.6 Rights After Settlement

Despite settlement and registration of the transfer, any term of this contract that can take effect after settlement or registration remains in force.

10.7 Further Acts

If requested by the other party, each party must, at its own expense, do everything reasonably necessary to give effect to this contract.

10.8 Severance

If any term or part of a term of this contract is or becomes legally ineffective, invalid or unenforceable in any jurisdiction it will be severed and the effectiveness, validity or enforceability of the remainder will not be affected.

10.9 Interpretation

(1) Plurals and Genders

Reference to:

- (a) the singular includes the plural and the plural includes the singular;
- (b) one gender includes each other gender;
- (c) a person includes a Body Corporate; and
- (d) a party includes the party's executors. administrators, successors and permitted assigns.

(2) Parties

- (a) If a party consists of more than one person, this contract binds them jointly and each of them individually.
- (b) A party that is a trustee is bound both personally and in its capacity as a trustee.

(3) Statutes and Regulations

Reference to statutes includes all statutes amending, consolidating or replacing them.

(4) Inconsistencies

If there is any inconsistency between any provision added to this contract and the printed provisions, the added provision prevails.

(5) **Headings**

Headings are for convenience only and do not form part of this contract or affect its interpretation.

- (1) This contract may be executed in two or more counterparts, all of which will together be deemed to constitute one and the same contract.
- (2) A counterpart may be electronic and signed using an Electronic Signature.

11. ELECTRONIC SETTLEMENT

11.1 Application of Clause

- (1) Clause 11 applies if the Buyer, Seller and each Financial Institution involved in the transaction agree to an Electronic Settlement using the same ELNO System and overrides any other provision of this contract to the extent of any inconsistency.
- (2) Acceptance of an invitation to an Electronic Workspace is taken to be an agreement for clause
- (3) Clause 11 (except clause 11.5(3)) ceases to apply if either party gives notice under clause 11.5 that settlement will not be an Electronic Settlement.

11.2 Completion of Electronic Workspace

- (1) The parties must:
 - (a) ensure that the Electronic Workspace is completed and all Electronic Conveyancing Documents and the Financial Settlement Schedule are Digitally Signed prior to settlement;
 - do everything else required in the Electronic Workspace or otherwise to enable settlement to occur on the Settlement Date.
- (2) If the parties cannot agree on a time for settlement, the time to be nominated in the Workspace is 4pm AEST.
- If any part of the Purchase Price is to be paid to discharge an Outgoing:
 - (a) the Buyer may, by notice in writing to the Seller, require that the amount is paid to the Buyer's Solicitor's trust account and the Buyer is responsible for paying the amount to the relevant authority;
 - for amounts to be paid to destination accounts other than the Buyer's Solicitor's trust account, the Seller must give the Buyer a copy of the current account for the Outgoing to enable the Buyer to verify the destination account details in the Financial Settlement Schedule.
- (4) If the Deposit is required to discharge any Encumbrance or pay an Outgoing at settlement:
 - (a) the Deposit Holder must, if directed by the Seller at least 2 Business Days prior to Settlement, pay the Deposit (and any interest accrued on investment of the Deposit) less commission as clear funds to the Seller's Solicitor;
 - (b) the Buyer and the seller authorise the Deposit Holder to make the payment in clause 11(4)(a);
 - the Seller's Solicitor will hold the money as Deposit Holder under the Contract;
 - the Seller and Buyer authorise the Seller's Solicitor to pay the money as directed by the Seller in accordance with the Financial Settlement Schedule.

11.3 Electronic Settlement

- Clauses 5.1(2) and 5.2 do not apply.
- Payment of the Balance Purchase Price electronically as directed by the Seller's Solicitor in the Financial Settlement Schedule satisfies the Buyer's obligation in clause 2.5(1).

- (3) The Seller and Buyer will be taken to have complied with:
 - (a) clause 2.5(3)(c), (e) and (f); and
 - (b) clause 2.5(5)(d) and (e),

(as applicable) if at settlement the Financial Settlement Schedule specifies payment of the relevant amount to the account nominated by the Commissioner of Taxation.

- (4) The Seller will be taken to have complied with clause 5.3(1)(b), (c), (d), (e) and (f) if:
 - (a) in relation to documents which are suitable for Electronic Lodgement in the Land Registry at settlement, the documents are Digitally Signed within the Electronic Workspace; and
 - (b) in relation to any other document or thing, the Seller's Solicitor:
 - confirms in writing prior to settlement that it holds all relevant documents which are not suitable for Electronic Lodgement and all Keys (if requested under clause 5.3(1)(d)) in escrow on the terms contained in the QLS E-Conveyancing Guidelines; and
 - (ii) gives a written undertaking to send the documents and Keys (if applicable) to the Buyer or Buyer's Solicitor no later than the Business Day after settlement; and
 - (iii) if requested by the Buyer, provides copies of documents in the Seller's Solicitors possession.
- (5) A party is not in default to the extent it is prevented from complying with an obligation because the other party or the other party's Financial Institution has not done something in the Electronic Workspace.
- (6) Any rights under the contract or at law to terminate the contract may not be exercised during the time the Electronic Workspace is locked for Electronic Settlement.
- (7) Electronic Settlement is taken to occur when Financial Settlement is effected, whether or not Electronic Lodgement has occurred.

11.4 Computer System Unavailable

If settlement fails and cannot occur by 4pm AEST on the Settlement Date because a computer system operated by the Land Registry, Office of State Revenue, Reserve Bank, a Financial Institution or the relevant ELNO System is inoperative, neither party is in default and the Settlement Date is deemed to be the next Business Day. Time remains of the essence.

11.5 Withdrawal from Electronic Settlement

- Either party may elect not to proceed with an Electronic Settlement by giving written notice to the other party.
- (2) A notice under clause 11.5(1) may not be given later than 5 Business Days before the Settlement Date unless an Electronic Settlement cannot be effected because:
 - (a) the transaction is not a Qualifying Conveyancing Transaction; or
 - (b) a party's solicitor is unable to complete the transaction due to death, a loss of legal capacity or appointment of a receiver or administrator (or similar) to their legal practice or suspension of their access to the ELNO System; or
 - (c) the Buyer's or Seller's Financial Institution is unable to use the relevant ELNO System to effect Electronic Settlement.

- (3) If clause 11.5(2) applies:
 - the party giving the notice must provide satisfactory evidence of the reason for the withdrawal; and
 - (b) the Settlement Date will be extended to the date 5 Business Days after the Settlement Date.

11.6 Costs

Each party must pay its own fees and charges of using the relevant ELNO System for Electronic Settlement.

11.7 Definitions for clause 11

In clause 11:

"Digitally Sign" and "Digital Signature" have the meaning in the ECNL.

"ECNL" means the Electronic Conveyancing National Law (Queensland).

"Electronic Conveyancing Documents" has the meaning in the Land Title Act 1994.

"Electronic Lodgement" means lodgement of a document in the Land Registry in accordance with the FCNL.

"Electronic Settlement" means settlement facilitated by an ELNO System.

"Electronic Workspace" means a shared electronic workspace within an ELNO System that allows the Buyer and Seller to effect Electronic Lodgement and Financial Settlement.

"ELNO" has the meaning in the ECNL.

"ELNO System" means a system provided by the ELNO for facilitating Financial Settlement and Electronic Lodgement.

"Financial Settlement" means the exchange of value between Financial Institutions facilitated by an ELNO System in accordance with the Financial Settlement Schedule.

"Financial Settlement Schedule" means the electronic settlement schedule within the Electronic Workspace listing the source accounts and destination accounts.

"Qualifying Conveyancing Transaction" means a transaction that is not excluded for Electronic Settlement by the rules issued by the relevant ELNO, Office of State Revenue, Land Registry, or a Financial Institution involved in the transaction.

12. ELECTRONIC CONTRACT AND DISCLOSURE

12.1 Electronic Signing

If this contract is signed by any person using an Electronic Signature, the Buyer and the Seller:

- (a) agree to enter into this contract in electronic form; and
- (b) consent to either or both parties signing the contract using an Electronic Signature.

12.2 Pre-contract Disclosure

The Buyer consents to the Seller's use of electronic communication to give any notice or information required by law to be given to the Buyer and which was given before the Buyer signed this contract.



C/- SSKB

P O Box 8319, GCMC QLD 9726 Australia

Client Solutions Ph: (07) 5504 2000 Fax: (07) 5504 2001

18 August 2020

COURAN COVE RESORT MARINE APTS G.T.P. 106784 Registered for GST

Tax Invoice

ABN: 32 955 764 518

Geoff Molloy gjmolloy00@gmail.com

C205 Ref

COURAN COVE RESORT MARINE APTS G.T.P. 106784 Re Lot 62

74.10 Fee Paid

Above Fee includes GST Legal Action

Please find enclosed Body Corporate Information Certificate pursuant to Section 40 of the Building Units and Group Titles Act 1980. Please arrange for all outstanding levies to be adjusted at settlement and a cheque in payment forwarded to us urgently along with the required form advising of the new owner's details and advice of transfer. Please impress upon your client the necessity to return the form to us promptly to ensure their interest in the lot is duly recorded and they do not incur penalties or lost discount due to non-receipt of information and notices.

Island Resort (Facilities & Equipment) Pty Ltd and Island Resorts (Infrastructure) Pty Ltd v Couran Cove Resort Community Body Corporate (Supreme Court Claim No. 5871 of 2020).

Island Resorts (Developments) Pty Ltd v Couran Cove Community Body Corporate and The Proprietors of Couran Cove Resort – Eco Lodges GTP 106986 and The Proprietors of Couran Cove Resort - Marine Apartments GTP 106784 and The Proprietors of Couran Cove Resort - Lagoon Lodges GTP 106808 and The Proprietors of Couran Cove Resort - Broadwater Villas GTP 106807 (Supreme Court Claim No.6868 of 2020)

Please note that there is no Community Management Statement or Interest Schedule Lot Entitlements applicable to this Scheme which is under the Building Units & Group Titles Act 1980. Before settlement, please go to http://www.stratamax.com.au//cert to obtain an update on the figures contained in the attached certificate. Access ID: 106784 SSKBGC 874616.

A fee of \$26.10 inclusive of all fees applies for this service which is payable on-line via credit card. This Access ID will expire one month from the date of this letter.

Please also note that a Community Facility Levy and a Utility Infrastructure Administrative and Sinking fund levy and a Community Body Corporate Administration & Sinking Fund levy is adopted by the body corporate as part of the budget each year and is referred to under the special levies section in the certificate and are on-going charges. We have also attached a statement to assist you in following the levies processed on the lot.

Yours faithfully, SSKB

CONTRACTS REGISTER

Date: 18/08/20

COURAN COVE RESORT MARINE APTS G.T.P. 106784

Administration						
Contractor Name and Address SSKR (Cold Coast) Pty I to		Details of Duties		Delegated Powers	Basis of Remuneration	
SSKB (Gold Coast) Pty Lte	ı Str	ata Managemen	(\$1.00 Sec Fee per lot per annum excl. GST plus	
10 Frigo Crt					Disbursements \$1 per lot per	
BUNDALL QLD 4217					annum plus GST payable	
					Quarterly in advance	
Commencement Date	01/11/18	Expiry Date	31/10/21	Copy of Agreement on File Y	Termination Date	
Term of Contract	3 years			Options		
Estimated Cost of Contract	0.00			Name of Financier		
Est Renumeration of Contract	0.00			Date of Advice from Financier		
Workers Comp No				Date of Withdrawal of Financier		
Work Cover						
Contractor Name and Address	Detai	ls of Duties		Delegated Powers	Basis of Remuneration	
Workcover QLD					•	
GPO BOX 2772						
BRISBANE QLD 4001						
Commencement Date	01/07/20	Expiry Date	30/06/21	Copy of Agreement on File	Termination Date	
Term of Contract	Policy: WS	SB180859248		Options		
Estimated Cost of Contract	0.00			Name of Financier		
Est Renumeration of Contract	0.00			Date of Advice from Financier		
Workers Comp No				Date of Withdrawal of Financier		
Contractor Name and Address	Detai	s of Duties	·····	Delegated Powers	Basis of Remuneration	
Commencement Date		Expiry Date	<u> </u>	Copy of Agreement on File	Termination Date	
Term of Contract				Options		
Estimated Cost of Contract				Name of Financier		
Est Renumeration of Contract				Date of Advice from Financier		
Workers Comp No				Date of Withdrawal of Financier		

Contractor Name and Address	Detail	s of Duties		Delegated Powers	Basis of Remuneration	
				new particular		
	ł					
					ļ	
	l					
Commencement Date	<u> </u>	Expiry Date	I	Copy of Agreement on File	Termination Date	
Term of Contract		Expiry Date	L	Options Options	Termination Date	
Estimated Cost of Contract				Name of Financier		
Est Renumeration of Contract				Date of Advice from Financier		
Workers Comp No				Date of Withdrawal of Financier		
Workers comp ivo				Date of Williams of Timbreto		
Contractor Name and Address	Detail	of Duties		Delegated Powers	Basis of Remuneration	
	1					
Commencement Date		Expiry Date		Copy of Agreement on File	Termination Date	
Term of Contract		1 2-7	L	Options		
Estimated Cost of Contract				Name of Financier		
Est Renumeration of Contract				Date of Advice from Financier		
Workers Comp No				Date of Withdrawal of Financier		

Date: 18/08/20

Register Of Authorisations Affecting Common Property COURAN COVE RESORT MARINE APTS G.T.P. 106784

Lot No	Date of	Authority Given To	Description of Area	Conditions
	Resolution			
	A			

Name of Body Corporate	COUR	COURAN COVE RESORT MARINE APARTMENTS						
Lot No.	62	Туре	GROUP TITLES PLAN	Plan No	106825			

ANNEXURE - LEVY DETAILS

Paid Notice Outs COMMUNITY FACILITY F 01/06/19 to 30/09/19 317.88 01/06/19 * 63.58 01/07/19 30/04/19 01/10/19 to 31/01/20 322.62 01/12/19 * 64.52 13/01/20 04/12/19 01/02/20 to 31/05/20 322.56 01/02/20 * 64.51 01/03/20 21/01/20	tanding 29.18
01/06/19 to 30/09/19 317.88 01/06/19 * 63.58 01/07/19 30/04/19 01/10/19 to 31/01/20 322.62 01/12/19 * 64.52 13/01/20 04/12/19	29.18
01/10/19 to 31/01/20 322.62 01/12/19 * 64.52 13/01/20 04/12/19	29.18
	29.18
1 01/02/20 to 31/05/20 322.56 01/02/20 * 64.51 01/03/20 21/01/20	29.18
* · · • · · · · · · · · · · · · · · · ·	29.18
01/06/20****30/09/20 322.56 01/06/20 * 64.51 01/07/20 16/04/20 22	
UTILITY INFRAS - ADMIN	
01/06/19-30/09/19 1445.58 01/06/19 * 289.12 01/07/19 30/04/19	
01/10/19-31/01/20	
01/02/20-31/05/20	-
01/06/20-30/09/20	32.04
	ļ
UTLITY INSFRUSTRU SINK	
01/06/19 to 30/09/19 130.80 01/06/19 * 26.16 01/07/19 30/04/19	
01/10/19 to 31/01/20 84.12 01/12/19 * 16.82 13/01/20 04/12/19	1
01/02/20 to 31/05/20 84.06 01/02/20 * 16.81 01/03/20 21/01/20	
01/06/20****30/09/20 84.06 01/06/20 * 16.81 01/07/20 16/04/20 6	31.96
PBC CONTRIBUTION ADMIN LEVY	
01/06/19 to 30/09/19 500.64 01/06/19 * 100.13 01/07/19 30/04/19	
01/10/19 to 31/01/20 698.22 01/12/19 * 139.64 13/01/20 04/12/19	1
01/02/20 to 31/05/20 698.22 01/02/20 * 139.64 01/03/20 21/01/20	
01/06/20****30/09/20 698.22 01/06/20 * 139.64 01/07/20 16/04/20 44	50.14
PBC CONTRIBUTION SINK LEVY	Ì
01/06/19 to 30/09/19 41.16 01/06/19 * 8.23 01/07/19 30/04/19	
01/10/19 to 31/01/20 222.60 01/12/19 * 44.52 13/01/20 04/12/19	
01/02/20 to 31/05/20 222.54 01/02/20 * 44.51 01/03/20 21/01/20	72.54
01/06/20****30/09/20 222.54 01/06/20 * 44.51 01/07/20 16/04/20 8	73.54

Items marked with **** are for periods that are outside the current financial year. They may also be subject to ratification at the next General Meeting.

COURAN COVE RESORT MARINE APTS G.T.P. 106784 LOT 62 (d) SECTION 30 Amount unpaid under a By-Law referred to in Section 30 \$873.54 This amount (if any) was due and payable on (e) SECTION 124(2) Amount unpaid in respect of a contribution levied under Section 124(2) \$4,450.14 09/09/99 Date due and payable Date levy imposed (f) SECTION 33(3) or (4) Nil Amount recoverable under Section 33(3) or (4) (g) DISCOUNTS Any discounts determined pursuant to Section 38A(4) and applicable to contributions determined in accordance with sections 38A(1) or 38A(2) or 38A(3) are shown under INSTALMENT DETAILS on page two of this Certificate. **INTEREST** Pursuant to By-Law interest is charged daily on unpaid levies 0.00 Interest Rate (% per annum) Interest amount due and payable as at the date of this Certificate Nil Nil Daily interest amount accruing **OTHER** \$740.51 Other amount(s) unpaid OTHER LEVIES Fund Amount Due Date Amount Overdue Purpose 681.96 See Annexure-Levy Details 4450.14 See Annexure-Levy Details 873.54 See Annexure-Levy Details Any amounts paid in the form of cheques or other negotiable instruments are subject to clearance. Amounts shown as due and payable may include levies which have been generated but not yet due. **SUMMARY** 9.618.61 Administrative Fund (Regular) 9.618.61 Sinking Fund (Regular) 555.66 555.66 Total Regular Levies: 10,174.27 10,174.27 Administrative Fund (Special) 9.032.04 9.032.04 Sinking Fund (Special) Nil Nil 681.96 Section 38A(3) 681.96 873.54 Section 30 873.54 4.450.14 4,450.14 Section 124(2) Nil Section 33(3) or (4) Nil Nil Interest to date Nil 740.51 740.51 Other 8,234.82 Other Levies 8,234.82

CERTIFICATION

The Common Seal of the Body Corporate for COURAN COVE RESORT MARINE APTS Community Title Scheme was hereunto affixed on 18 August 2020 in the presence of SSKB (Gold Coast) Pty Ltd by its duly authorised officer being a person authorised under the Body Corporate and Community Management Act 1997 to attest the fixing of the Common Seal.



CERTIFICATE PURSUANT TO SECTION 40

Name of Body Corporate	COURAN COVE RESORT MARINE APTS G.T.P. 106784				
Location	Doeblien Drive	Registered			
	Couran Cove	Office			
	South Stradbroke Island	(if not Location)			

	STATUTORY FEE	
	Geoff Molloy gjmolloy00@gmail.com	AMOUNT \$74.10
		Paid
REFERENCE	C205	

LOT NO. 62 PROPRIETOR Coralsprit Pty Ltd ATF	LOT
--	-----

	OFFICE BEARERS PART 1. (Section 40(1)(a))						
OFFICE	NAME	ADDRESS					
CHAIRPERSON	Lachlan McIntosh (HUB2)	Level 27 / 10 Eagle Street BRISBANE QLD 4000					
SECRETARY	Michelle Puck (Non-Voting)	P O Box 8319 GOLD COAST MC QLD 9726					
TREASURER	lan Maynard (Non-Voting)	20 Gloucester St SPRING HILL QLD 4000					
COMMITTEE (1)	Steve Anderson	1617 Rosebank Way HOPE ISLAND QLD 4212					
(2)	Dominic Blewitt (HUB1)	email::dominic.blewitt@ bartercard.com;					
(3)	Col Elliott	1066 Killymoon Dve HOPE ISLAND QLD 4212					
(4)	Paul Eyschen	email::admin@ edgcapital.com.au;					
(5)	Jonathan Kuper	email::admin@ edgcapital.com.au;					
(6)							
(7)							
RODY COPPORATE MANAGER APPOINTED LINDER SECTION 50 OR 94							

BODY CORPORATE MANAGER APPOINTED UNDER SECTION 50 OR 94

SSKB (Gold Coast) Pty Ltd PO Box 8319 Telephone: 07 5504 2000 Facsimile: 07 5504 2001

INSURANCE						
TYPE	COMPANY	POLICY NO.	SUM INSURED	PREMIUM	DUE DATE	
BUILD CATASTROPHE/EM	CHU U/W Agencies P/L	QNC60606	Ref 106751	Ref106,751	01/06/21	
BUILDING	CHU U/W Agencies P/L	QNC60606	Ref 106751	Ref106,751	01/06/21	
COMMON AREA CONTENTS	CHU U/W Agencies P/L	QNC60606	Ref 106751	Ref106,751	01/06/21	
COST OF TEMP ACCOM	CHU U/W Agencies P/L	QNC60606	Ref 106751	Ref106,751	01/06/21	
EXT COVER RENT/ACCOM	CHU U/W Agencies P/L	QNC60606	Ref 106751	Ref106,751	01/06/21	
FIDELITY GUARANTEE	CHU U/W Agencies P/L	QCS60607	100,000	Included	01/06/21	
LOSS OF RENT	CHU U/W Agencies P/L	QNC60606	Ref 106751	Ref106,751	01/06/21	
OFFICE BEARERS	CHU U/W Agencies P/L	QCS60607	5,000,000	Included	01/06/21	
PUBLIC LIABILITY	CHU U/W Agencies P/L	QCS60607	20,000,000	7,363.77	01/06/21	
STAMP DUTY/GST/OTHER	CHU U/W Agencies P/L	QCS60607	0	1,828.41	01/06/21	
STORAGE/EVACUATION	CHU U/W Agencies P/L	QNC60606	Ref 106751	Ref106,751	01/06/21	
VOLUNTARY WORKERS	CHU U/W Agencies P/L	QCS60607	200,000	Included	01/06/21	

GOLD COAST M C QLD 9726

COURAN COVE RESORT MARINE APTS G.T.P. 106784 62 OT **LEVIES** PART 2. (Section 40(1)(c)) COURAN COVE RESORT MARINE APTS G.T.P. 106784 The proprietors certify as follows:-Financial year end 31/05/20 **Total Entitlements** Entitlements for subject lot 31/05/21 Levy year end 1152 6 (a) ADMINISTRATIVE FUND (Section 38A(1)) REGULAR PERIODIC CONTRIBUTIONS (Section 38A(5)) Total amount determined with respect to the subject lot \$5,737.74 Number of Instalments 4 PERIOD AMOUNT DUE DATE DATE PAID DISCOUNT IF PAID BY GENERATED 01/06/19 to 30/09/19 1637.70 01/06/19 327.54 01/07/19 30/04/19 INSTALMENT 01/10/19 to 31/01/20 1366.68 01/12/19 273.34 13/01/20 04/12/19 01/03/20 01/02/20 to 31/05/20 1366.68 01/02/20 273.34 21/01/20 **DETAILS** 01/06/20****30/09/20 1366.68 01/06/20 273.34 01/07/20 16/04/20 Date to which instalments are paid Amount outstanding 31/05/18 \$9,618,61 (including levies generated but not yet due) SPECIAL (LUMP SUM) CONTRIBUTIONS Total amount determined with respect to the subject lot \$6.140.94 Number of Instalments DISCOUNT AMOUNT DUE DATE DATE PAID IF PAID BY **GENERATED INSTALMENT DETAILS** Purpose for which contribution was made See Annexure 'Levy Details' Amount of the contribution unpaid (if not generated liability may still exist) \$9,032.04 (b) SINKING FUND (Section 38A(2)) REGULAR PERIODIC CONTRIBUTIONS (Section 38A(5)) \$308.70 Total amount determined with respect to the subject lot Number of Instalments 4 PERIOD AMOUNT DUE DATE DATE PAID DISCOUNT IF PAID BY **GENERATED** 123,48 01/06/19 24,70 01/07/19 30/04/19 01/06/19 to 30/09/19 INSTALMENT 01/10/19 to 31/01/20 61.74 01/12/19 12.35 13/01/20 04/12/19 **DETAILS** 01/02/20 to 31/05/20 61.74 01/02/20 12.35 01/03/20 21/01/20 01/06/20****30/09/20 61.74 01/06/20 12.35 01/07/20 16/04/20 Date to which instalments are paid Amount outstanding 30/09/18 \$555.66 (including levies generated but not yet due) SPECIAL (LUMP SUM) CONTRIBUTIONS Total amount determined with respect to the subject lot Nil n Number of Instalments AMOUNT DUE DATE DATE PAID DISCOUNT IF PAID BY GENERATED **INSTALMENT DETAILS** Purpose for which contribution was made Amount of the contribution unpaid (if not generated liability may still exist) Nil (c) SECTION 38A(3)

\$84.06

\$681.96

COURAN COVE RESORT MARINE APAR

Amount determined with respect to the subject lot

Purpose for which contribution was made

Date on which contribution is due and payable

Amount of the contribution unpaid

Date of determination

Page 1	INSPECTIONS REGISTER	18 August 2

COURAN COVE RESORT MARINE APTS G.T.P. 106784

A/c No	Category	Period	Next Due	Last Done	Inspected By	Result	Rating	Property Details
601	Insurance Valuation	5 Yearly	09/05/15	10/05/10	Csa Valuers			
603	Sinking Fund Report	5 Yearly	23/07/22	24/07/17	Napier & Blakeley			
605	General Safety Report - Star	Annually	20/04/21	20/04/20	Star Building Management Ser			
610	Pool Safety Certificate	Annually	14/02/21	14/02/20	David Eagles			
612	Asbestos Report	Once Only	13/08/12	13/08/12	Seymour Consultants			
631	Obcc Cladding Checklist Part 1	Once Only	04/04/19	04/04/19	Star Bms			
632	Obcc Cladding Checklist Part 2	Once Only	12/06/19	12/06/19	Gma			
633	Qbcc Cladding Checklist Part 3	Once Only	31/10/19	31/10/19	Safer Buildings			



COURAN COVE RESORT MARINE APARTMENTS G.T.P. 106784

STATEMENT

Legal Action

Coralsprit Pty Ltd ATF 2 Bega Rd KINGSTON QLD 4114

Statement Period						
01 Jun 19 to 18 Aug 20						
A/c No	62		Lot No	62		
Page Number		1 of 2				

Transfer Date: 21/03/17 Last Certificate Issued: 01/03/17 Legal Action Date Type Details Reference Debit Credit Balance Brought forward 15.705.60 15.705.60 01/06/19 01/06/19 To 30/09/19 10046529 41.16 15,746.76 PBC Contrib Sinking 01/06/19 10046721 1.637.70 17,384.46 01/06/19 To 30/09/19 Administrative Fund 10046913 1,445.58 18,830.04 01/06/19 01/06/19 Utility Infras Admin 01/06/19 01/06/19 To 30/09/19 10047105 123.48 18,953.52 Sinking Fund 19,084.32 01/06/19 Utility Infras Sink 01/06/19 To 30/09/19 10047297 130.80 01/06/19 PBC Contrib Admin 01/06/19 To 30/09/19 10047489 500.64 19,584.96 01/06/19 01/06/19 To 30/09/19 10047681 317.88 19,902.84 Community Facility 17/09/19 Other Lot62 Legal Recovery 220.00 20,122.84 M0002220 20,133.29 17/09/19 Other Legal Process-Cs M0002221 10.45 04/12/19 Administrative Fund 01/10/19 To 31/01/20 10047873 1,366.68 21,499.97 04/12/19 Community Facility 01/10/19 To 31/01/20 10048065 322.62 21,822.59 04/12/19 Utility Infras Admin 01/12/19 10048257 1,565.16 23,387.75 04/12/19 Sinking Fund 01/10/19 To 31/01/20 10048449 61.74 23,449.49 10048641 23,533.61 04/12/19 Utility Infras Sink 01/10/19 To 31/01/20 84.12 04/12/19 PBC Contrib Admin 01/10/19 To 31/01/20 10048833 698.22 24,231.83 04/12/19 01/10/19 To 31/01/20 10049025 222.60 24,454,43 PBC Contrib Sinking 10049217 1.366.68 25.821.11 21/01/20 01/02/20 To 31/05/20 Administrative Fund 21/01/20 01/02/20 To 31/05/20 10049409 322,56 26.143.67 Community Facility 21/01/20 01/02/20 10049601 1.565.10 27,708.77 Utility Infras Admin 27,770.51 01/02/20 To 31/05/20 10049793 61.74 21/01/20 Sinking Fund 27,854.57 01/02/20 To 31/05/20 10049985 84.06 21/01/20 Utility Infras Sink 28,552.79 21/01/20 PBC Contrib Admin 01/02/20 To 31/05/20 10050177 698.22 21/01/20 PBC Contrib Sinking 01/02/20 To 31/05/20 10050369 222.54 28,775.33 06/03/20 Receipt Administrative Fund R0013920 2,305.07 26,470.26 06/03/20 Receipt Community Facility RA013920 615.60 25,854.66 Utility Infras Admin RB013920 47.40 25,807.26 06/03/20 Receipt Utility Infras Sink RC013920 74.76 25,732.50 06/03/20 Receipt \$28,775.33 \$3,042.83 \$25,732.50

 Over 90 Days
 90 Days
 60 Days
 30 Days
 Current
 BALANCE DUE:
 \$28,181.64

 23,860.74
 0.00
 0.00
 4,320.90
 0.00
 Date Paid
 Amount Paid

******* PLEASE RETURN THIS SECTION WITH YOUR PAYMENT *******

Send payment to:-

COURAN COVE RESORT MARINE APTS SSKB COURAN COVE RESORT MARINE APTS

G.T.P. 106784

62 Coralsprit Pty Ltd Atf 01 Jun 19 to 18 Aug 20

COURAN COVE RESORT MARINE APARTMENTS G.T.P. 106784

STATEMENT

Coralsprit Pty Ltd ATF	
2 Bega Rd	
KINGSTON QLD 4114	

	Statement Period						
01 Jun 19 to 18 Aug 20							
A/c No	62		Lot No	62			
Page Number		2 of 2					

Date	Туре	Details	Reference	Debit	Credit	Balance
06/03/20 06/03/20 16/04/20 16/04/20 16/04/20 16/04/20 16/04/20 16/04/20	Receipt Receipt Administrative Fund Community Facility Utility Infras Admin Sinking Fund Utility Infras Sink PBC Contrib Admin PBC Contrib Sinking	01/06/20 To 30/09/20 01/06/20 01/06/20 To 30/09/20	RD013920 RE013920 I0050561 I0050753 I0050945 I0051329 I0051521 I0051713	28,775.33 1,366.68 322.56 1,565.10 61.74 84.06 698.22 222.54	3,042.83 1,707.06 164.70	25,732.50 24,025.44 23,860.74 25,227.42 25,549.98 27,115.08 27,176.82 27,260.88 27,959.10 28,181.64
				\$33,096.23	\$4,914.59	\$28,181.64



Policy No: Period of Insurance:

QCS60748

From:

01/06/19

To:

01/06/20 at 4.00 pm

DISCLAIMER NOTICE

This Certificate of Currency has been issued to SSKB for the records of the Body Corporate only. We strongly recommend that the Insurance Cover be confirmed directly with the Underwriter or Insurance Broker.

The Insured & Situation:
BC FOR COURAN COVE
RESORT-BROADWATER VILLAS GTP
106807
COURAN COVE RESORT
SOUTH STRADBROKE
QLD 4216

Certificate of Currency - Tax Invoice

Cover Select	ed	Sum Insured
POLICY 1	INSURED PROPERTY (Building) Loss of Rent/Temp Accommodation INSURED PROPERTY (Common Area Contents) FLOOD	Not selected Not selected Not selected Not selected
POLICY 2	LIABILITY TO OTHERS	Limit of Liability - 20,000,000
POLICY 3	VOLUNTARY WORKERS - Refer to Table of Benefits	200,000/2,000
POLICY 4	WORKERS COMPENSATION (ACT, TAS & WA ONLY)	Not selected
POLICY 5	FIDELITY GUARANTEE	100,000
POLICY 6	OFFICE BEARER'S LEGAL LIABILITY	Limit of Liability - 5,000,000
POLICY 7	MACHINERY BREAKDOWN Loss of Rent/Temp Accommodation (20%)	Not selected Not selected
POLICY 8	CATASTROPHE INSURANCE (Insured Property) Extended cover – Rent/Temp Accommodation Escalation in Cost of Temp Accommodation Cost of Storage and Evacuation	Not selected Not selected Not selected Not selected
POLICY 9	PART A - Government Audit Costs PART B - Appeal Expenses - common property health & safety breaches PART C - Legal Defence Expenses	25,000 100,000 50,000
POLICY 10	LOT OWNER'S FIXTURES AND IMPROVEMENTS (per lot)	Not selected

 Date of Issue:
 07/06/19

 Issue Fee Incl GST:
 0.00

 Issue Fee GST:
 0.00

This certificate confirms that on the date of issue noted above, a policy existed for the sums insured shown.

It is not intended to amend, extend, replace or override the policy terms and conditions contained in the actual policy document. This certificate is issued as a matter of information only and confers no rights on the certificate holder.

CHU Underwriting Agencies Pty Ltd is an underwriting intermediary acting on behalf of the insurers.

Our Ref: 0175738



Policy No: Period of Insurance:

QNC60606

From:

01/06/19

To:

01/06/20 at 4.00 pm

DISCLAIMER NOTICE

This Certificate of Currency has been issued to SSK8 for the records of the Body Corporate only. We strongly recommend that the Insurance Cover be confirmed directly with the Underwriter or Insurance Broker.

The Insured & Situation: BODY CORPORATE FOR COURAN COVE RESORT MCP106751,GTP106784,106808,106783,10 6807 COURAN COVE RESORT SOUTH STRADBROKE QLD 4216

Certificate of Currency - Tax Invoice

Cover Selected					
POLICY 1	INSURED PROPERTY (Building) Loss of Rent/Temp Accommodation (20.2%) INSURED PROPERTY (Common Area Contents) FLOOD	133,436,939 26,954,262 1,334,370 Not selected			
POLICY 2	LIABILITY TO OTHERS	Not selected			
POLICY 3	VOLUNTARY WORKERS - Refer to Table of Benefits	Not selected			
POLICY 4	WORKERS COMPENSATION (ACT, TAS & WA ONLY)	Not selected			
POLICY 5	FIDELITY GUARANTEE	Not selected			
POLICY 6	OFFICE BEARER'S LEGAL LIABILITY	Not selected			
POLICY 7	MACHINERY BREAKDOWN Loss of Rent/Temp Accommodation (20%)	Not selected Not selected			
POLICY 8	CATASTROPHE INSURANCE (Insured Property) Extended cover – Rent/Temp Accommodation (15%) Escalation in Cost of Temp Accommodation (5%) Cost of Storage and Evacuation (5%)	13,343,694 2,001,554 667,185 667,185			
POLICY 9	PART A - Government Audit Costs PART B - Appeal Expenses - common property health & safety breaches PART C - Legal Defence Expenses	Not selected Not selected Not selected			
POLICY 10	LOT OWNER'S FIXTURES AND IMPROVEMENTS (per lot)	250,000			

Date of Issue: Issue Fee Incl GST: Issue Fee GST: 07/06/19 0.00

This certificate confirms that on the date of issue noted above, a policy existed for the sums insured shown.

It is not intended to amend, extend, replace or override the policy terms and conditions contained in the actual policy document. This certificate is issued as a matter of information only and confers no rights on the certificate holder.

CHU Underwriting Agencies Pty Ltd is an underwriting intermediary acting on behalf of the insurers.

Our Ref: 0218273



Annexure A

1. MODIFICATIONS TO TERMS OF CONTRACT

- 1.1 The Terms of Contract for Houses and Residential Land which form part of this Contract (the Standard Conditions) are amended as follows:
- (a) This Contract is not subject to finance and Clause 3 is deleted in its entirety;
- (b) This Contract is not subject to a Building. Pest or Pool Inspection and clause 4 is deleted in its entirety;
- (c) Clause 5.1 (1) is amended by replacing the words 4 pm AEST with 5 pm AEST;
- (d) The following clauses are deleted: clauses 7.4 (2) and (3); clauses 7.5 (2), (3) and (4); clause 7.6 (4) and clauses 7.7 (1) and (2);
- (e) Clause 7.4 (4) is amended to delete the words or clause 7.4 (2);
- (f) A new clause 7.5 (2) is inserted as follows
- The Buyer cannot not object to, delay settlement, terminate this Contract, deduct or retain any amount from the Purchase Price or make any requisition or claim for compensation about any error in the boundaries or area of the Land, any encroachment onto or from the Land or a mistake or omission in describing the Property or the Seller's Title to it (whether material or otherwise).
- 1.2 The Standard Conditions do not apply to this Contract to the extent of any inconsistency with these special conditions and the latter will prevail.
- 2. BUYER HAS SATISFIED ITSELF
- 2.1 The Buyer acknowledges and agrees that it has inspected or has had the opportunity to inspect the Property and has satisfied itself about:
- (a) The quality, condition and state of repair of the Property;
- (b) The purposes for which the Property can be used (including without limitation any restrictions on its use or development);
- (c) The fitness of the Property for the purposes the Buyer wants it;
- (d)The existence or otherwise of necessary approvals and licences from authorities relating to the Property including any failure to comply with any approvals or licences;
- (e) The carrying on of any notifiable activity on the land or the existence or otherwise of any hazardous contaminant on the land within the meaning of the Environmental Protection Act 1994;
- (f) The existence or otherwise of any legal and physical defects;
- (g) Accepts the Property on an as is where is basis, in its present condition and subject to any latent or patent defects.
- 2.2 The Buyer acknowledges and agrees that it has not relied on any representations, warranties or information provided or statements about the Property made by or on behalf of the Seller, the Seller's Agent or their respective employees or agents, other than as set out in this Contract.
- 3. Inventory
- 1 x Microwave
- 1x Fridge
- 1x Electric Jug
- 1x Toaster
- Assorted Cutlery + Crockery+ Glasses + Pots/Pans
- 1x Desk
- 1x Chair
- 2x cane chairs
- 2x small bedside tables
- 1x lamp
- 1x blanket box/coffee table
- 1x foldout sofa bed lounge
- 1x TV
- 1x outdoor setting (table + 4 x chairs)
- 1x double (split single) bed

